Nguyen T. Vu

Phone: (202) 373-6254 Fax: (202) 373-6001 nguyen.vugbingham.com

November 18, 2009

#### By Overnight Mail

Julie P. Orchard, Secretary Public Service Commission of Utah Heber M. Wells Building, 4th Floor 160 East 300 South Salt Lake City, UT 84111

IntelePeer, Inc. - Application for a Certificate of Public Convenience and Re: Necessity to Provide Facilities-Based and Resold Local Exchange, Access, and Interexchange Telecommunications Services

Dear Ms. Orchard:

Enclosed for filing with the Commission are an original and five (5) copies of the above referenced Application.

Please note that the information provided in Exhibits C, E and F in support of the Application, are proprietary and confidential as they contains sensitive financial information. Unrestricted availability of these documents would provide competitors with key knowledge regarding Applicants' finances and business that would otherwise be unavailable and could place Applicants at a significant competitive disadvantage. Confidential treatment of the information contained in Exhibits C, E and F is therefore required to avoid commercial and a competitive injury. Accordingly, five (5) copies of Exhibits C, E and F are being filed under seal in sealed envelopes.

Please date-stamp the enclosed extra copy of this filing and return it in the attached selfaddressed, postage prepaid envelope provided. Also enclosed is a diskette with an electronic copy of this filing in PDF and MS Word format. Should you have any questions concerning this filing, please do not hesitate to contact Nguyen Vu at (202) 373-6254.

Respectfully submitted,

Ronald W. Del Sesto, Jr. Nguyen T. Vu

Counsel for IntelePeer, Inc. 2020 K Street NW

Bingham McCutchen LLP Washington, DC 20086-1806

> 202.373.6000 202.373.6001 bingham.com

Boston Hartford

Hong Kong

Los Angeles New York

San Francisco Santa Monica

Silicon Valley Tokvo Walnut Creek

Orange County

London

A/73218791.1

## BEFORE THE UTAH PUBLIC SERVICE COMMISSION

Application of		
IntelePeer, Inc.	) Docket No	_
for a Certificates of Public Convenience and Necessity to Provide Resold and Facilities-Based Local Exchange, Access, and Interexchange Telecommunications Services in the	) ) ) )	
State of Utah	)	

#### APPLICATION OF INTELEPEER, INC.

IntelePeer, Inc. ("IntelePeer" or "Applicant"), by its undersigned counsel and pursuant to Chapter 8b of Title 54 of the Utah Code, Utah Admin. Code §§ 746-100 et seq., and the rules and regulations of the Utah Public Service Commission ("Commission"), hereby applies for a Certificate of Public Convenience and Necessity to authorize it to provide facilities-based and resold local exchange, access, and interexchange telecommunications service throughout the State of Utah.

The State of Utah has adopted a policy of allowing competition in the local and long distance telecommunications markets, recognizing that it is in the public interest to develop effective competition to ensure that all consumers will have access to high quality, low cost, and innovative telecommunications services. The Federal Telecommunications Act of 1996 also seeks to promote competition and reduce regulation in order to secure lower prices and higher quality telecommunications services for American consumers. Both goals will be promoted by granting this Application.

The Applicant submits the following information pursuant to Utah Admin. Code R746

349 in support of its request.

1. Applicant's legal name is IntelePeer, Inc. Applicant may be reached at its

principal place of business:

IntelePeer, Inc.

2855 Campus Drive, Suite 200

San Mateo, CA 94403

(650) 525-9200 (Tel)

(650) 287-2628 (Fax)

www.intelepeer.com

2. R746-349-3(A)(16). Proof of authority to conduct business. IntelePeer, Inc. is a

corporation organized under the laws of the State of Delaware. IntelePeer was originally

founded on July 1, 2003 as VoEx, Inc. ("VoEx"). On September 18, 2007, VoEx changed its

name to IntelePeer, Inc. A copy of its Articles of Incorporation is attached hereto as Exhibit A.

IntelePeer's authority to transact business as a foreign corporation in Utah is attached hereto as

Exhibit B.

3. Correspondence or communications pertaining to this Application should be

directed to:

Ronald W. Del Sesto, Jr.

Nguyen T. Vu

Bingham McCutchen LLP

2020 K Street, NW

Washington, DC 20006-1806

Tel: (202) 373-6000

Fax: (202) 373-6001

Email: r.delsesto@bingham.com

nguyen.vu@bingham.com

2

4 Questions concerning the ongoing operations of IntelePeer following certification should be directed to:

Andre Simone, Chief Financial Officer 2855 Campus Drive, Suite 200 San Mateo, CA 94403 (650) 525-9200 (Tel) (650) 287-2628 (Fax) Email: asimone@intelepeer.com

5. IntelePeer's registered agent in the State of Utah is:

Corporation Service Company 2180 South 1300 East, Suite 6509 Salt Lake City, UT 84106

6. For purposes of consumer inquiries, the contact information will be:

IntelePeer, Inc.
Customer Relations
2855 Campus Drive, Suite 200
San Mateo, CA 94403
cs@intelepeer.com, or online at <a href="http://vvvvw.intelepeer.com/contactus/contactus.php">http://vvvvw.intelepeer.com/contactus/contactus.php</a>.

IntelePeer's 24-hour customer service number is:

(866) 780-8639

- 7. R746-349-3(A)(3). Facilities to be used. IntelePeer has not yet reached an interconnection or resale agreement with any incumbent local exchange carrier in Utah.
- 8. R746-349-3(A)(4). Services to be offered. IntelePeer seeks authority to provide facilities-based and resold local exchange and interexchange telecommunications services to and from all points in the State of Utah. Therefore, IntelePeer seeks statewide authority.

IntelePeer seeks authority to provide facilities-based and resold local exchange and interexchange telecommunications services in Utah. IntelePeer will offer basic local exchange services, custom calling features, and interexchange toll services, including toll free services, to business and enterprise customers.

While IntelePeer will not initially offer prepaid local exchange and interexchange services, IntelePeer may offer such services in the future. IntelePeer may also provide access services to payphone service providers. IntelePeer will offer basic local exchange services, custom calling features, and interexchange toll services, including toll free services, to business and enterprise customers. Facilities-based local exchange service will be provided via (1) commercial wholesale agreement with incumbents, (2) IntelePeer's own facilities, or (3) a combination thereof Initially, toll services will be provided via IntelePeer's underlying long distance carrier. IntelePeer, however, seeks the full range of resold and facilities-based local exchange and interexchange authority so that it can have the flexibility in provisioning services in the future.

IntelePeer will offer interexchange telecommunications statewide. Initially, IntelePeer intends to provide local exchange service in the areas served by Qwest and does not plan to provide service in areas of any small or rural local exchange carriers. However, IntelePeer does seek statewide authority so that it may expand into other services as market conditions warrant and as additional service areas become open to competition.

Grant of this application will promote the public interest by increasing telecommunications service competition in Utah. IntelePeer will provide customers with high quality, cost effective telecommunications services, with an emphasis on customer service. In addition to driving prices closer to costs, thereby ensuring just and reasonable rates, competition also promotes efficiency in the delivery of services and in development of new services. These benefits work to maximize the public interest by providing continuing incentives for carriers to reduce costs while, simultaneously, promoting the availability of potentially desirable services.

At this time, Applicant does not seek to terminate any exemption held by a rural telephone company pursuant to 47 U.S.C. § 251(f).

IntelePeer will continuously monitor and maintain a high level of control over its network on a 24-hours-a-day, 7-days-a-week basis. IntelePeer's proposed local exchange tariff is provided as Exhibit G and its proposed intrastate access tariff is provided as Exhibit H...

- 8. (a) R746-349-3(A)(4)(a). Classes of customers. IntelePeer will initially offer its services to business customers. IntelePeer may serve residential customers in the future as market conditions warrant.
- 8. (b) R746-349-3(A)(4)(b). Location of service. IntelePeer intends to provide resold and facilities-based services to subscribers from all points within the State of Utah and therefore seeks statewide authorization. Although IntelePeer requests authority to operate statewide in Utah, it will initially provide local exchange service in the existing service areas of incumbent local exchange carriers ("LECs") that are not exempted from the general duty to interconnect pursuant to section 251(f) of the Telecommunications Act of 1996. IntelePeer seeks statewide authority, so that it may expand its service areas as market conditions warrant and as additional service areas become open to competition.
- 9. R746-349-3(A)(5). Access to standard services. IntelePeer will provide access to ordinary intraLATA and interLATA message toll calling, operator services, directory assistance, directory listings, and emergency services through interconnection with its underlying carrier.
- 10. R746-349-3(A)(7). Professional experience and education of managerial personnel and personnel responsible for Utah operations. IntelePeer does not plan to operate an office within Utah. Responsibility for Utah operations will be handled by Applicant's current management team from its headquarters in San Mateo, California. Descriptions of the extensive telecommunications and managerial experience of Applicant's key personnel are attached hereto as Exhibit D.

- 11. <u>R746-349-3\_(A)(1), R746-349-3(A)(10)--(12).</u> Financial abilities. IntelePeer is financially qualified to provide local exchange telecommunications services in Utah. IntelePeer has access to the financing and capital necessary to conduct the telecommunications operations as specified in this application, and will rely upon its existing personnel and technological and financial resources to provide intrastate services.
- 11. (a) R746-349-3(A)(9). Chart of accounts. IntelePeer's chart of accounts including account numbers, names, and brief descriptions is attached hereto as Exhibit E. Please note that the information provided in Exhibit E is proprietary and has been attached in a sealed envelope marked "Confidential." As it contains sensitive information regarding IntelePeer's business operations, IntelePeer respectfully requests that this information be treated confidentially and not be released to any member of the public absent IntelePeer's prior written permission.
- Financial Statements. 11. (b) <u>R746-349-3(A)(10)(a)--(d).</u> IntelePeer's success in developing innovative products and services and in expanding its geographic reach has translated into impressive growth in recent years, and as such IntelePeer is extremely well-qualified financially to operate and expand its business. Financial information demonstrating IntelePeer's financial qualifications is provided confidentially as Exhibit C. As shown in the attached information, IntelePeer is financially qualified to operate within the State of Utah. Please note that the information provided in Exhibit C is proprietary and has been attached in a sealed envelope marked "Confidential." As it contains sensitive information regarding IntelePeer's business operations, IntelePeer respectfully requests that this information be treated confidentially and not be released to any member of the public absent IntelePeer's prior written permission. IntelePeer possesses the sound financial background necessary to effectively procure, install, and operate the facilities and services requested in this application. In

accordance with R746-349-3(A)(11)(a), these financial statements show that IntelePeer has a positive net worth and that IntelePeer has sufficient cash flow to provide its proposed services.

- 11. (c) R746-349-3(A)(10)(b). Sufficient projected and verifiable cash flow. Paragraph 12, below, provides evidence of a sufficient projected and verifiable cash flow to meet cash needs as shown in IntelePeer's three-year projection of expected operations.
- 11. (d) R746-349-3(A)(2), R746-349-3(A)(11)(c). Bond requirement. IntelePeer respectfully requests a waiver of the requirement in R746-34-2(A)(2) for proof of a bond in the amount of \$100,000. IntelePeer will not require Utah customers to make a service deposit. Further, as demonstrated by its financial statements, IntelePeer has the financial resources to cover other liabilities to telecommunications customers.

#### 12. R746-349-3(A)(12). Three-year projection of expected operations.

IntelePeer provides evidence of a sufficient projected and verifiable cash flow to meet cash needs as shown in IntelePeer's three-year projection of expected operations, as evidenced in confidential Exhibit F. IntelePeer respectfully requests a waiver of the requirement in R746-34-2(A)(12) that it provide a 5-year projection of expected operations. As evidenced by IntelePeer's financial statements provided with this application, IntelePeer has the financial stability to provide telecommunications in Utah.

12. (a) R746-349-3(12)(a). Income and cash flow statements. Attached hereto under seal as Exhibit F is IntelePeer's five-year projection of expected operations. Please note that the information provided in Exhibit F is proprietary and has been attached in a sealed envelope marked "Confidential." As it contains sensitive information regarding IntelePeer's business operations, IntelePeer respectfully requests that this information be treated confidentially and not be released to any member of the public absent IntelePeer's prior written permission.

- 12. (b) R746-349-3(A)(12)(b). Types of technology to be deployed. Initially, technology deployed will be the same as that currently deployed by the incumbent providers whose networks should be utilized on a wholesale basis. Other network technologies will be considered if an when IntelePeer deploys its own network facilities.
- 12. (c) R746-349(A)(12)(c). Maps of facilities locations. As stated in Paragraph 7 above, IntelePeer currently does not initially plan to construct facilities in Utah and has not yet completed its plans for leasing facilities in Utah. IntelePeer has not yet determined the exact location of future facilities and descriptions of the specific facilities to be deployed in the future.
- 13. R746-349-3(A)(6), R746-349-3(A)(13). Implementation schedule. IntelePeer proposes to begin providing service in Utah shortly after it has obtained the requisite Commission approval and as soon as it is able to establish interconnection agreements and obtain collocation space with the relevant incumbent LECs in Utah.
- 14. R746-349-3(A)(1), R746-349-3(A)(14). Technical and managerial abilities. IntelePeer is technically and managerially qualified to provide competitive local exchange and interexchange services in Utah. IntelePeer's Utah operations will be directed by its existing corporate management, technical and operations staffs who are responsible for the interexchange and local exchange operations in other states. A description of the background of Applicant's key personnel, which demonstrates the extensive telecommunications experience of IntelePeer's management team, is attached hereto as Exhibit D.

IntelePeer is well qualified to provide telecommunications services in Utah. Currently, IntelePeer has authority to provide facilities-based and resold, local exchange and interexchange telecommunications services in Colorado, the District of Columbia, Indiana, Maryland, Massachusetts, New York, Oregon, Texas and Washington State, and is in the process of

applying for authority in the remaining states. IntelePeer has not been denied requested certification in any jurisdiction, nor has any permit, license, or certificate been revoked by any authority.

- 15. <u>R746-349-3(A)(8)</u>. <u>Employees.</u> IntelePeer's Utah operations will be directed by its management team located at IntelePeer's headquarters in San Mateo, California. In lieu of an organizational chart, IntelePeer offers the biographies of its key personnel set forth in Exhibit D to satisfy R746-349-3(A)(8).
- 16. R746-349-3(A)(1), R746-349-3(A)(15). Public interest. Granting this Application will promote the public interest by increasing competition in the provision of telecommunications services in Utah. IntelePeer will provide customers high quality, cost effective telecommunications services, with an emphasis on customer service. In addition to driving prices closer to costs, thereby ensuring just and reasonable rates, competition also promotes efficiency in the delivery of services and in the development of new services. These benefits work to maximize the public interest by providing continuing incentives for carriers to reduce costs while, simultaneously, promoting the availability of potentially desirable services.
- 17. R746-349-3(A)(17)—(18). <u>Unauthorized switching, solicitation of new customers,</u> and prevention of unauthorized switching or other illegal activities. IntelePeer has not had any complaints filed against it in any jurisdiction.

\* \* \*

For the reasons stated above, Applicant respectfully submits that the public interest, convenience, and necessity would be furthered by a grant of this Application for the authority to provide all types of facilities-based and resold local exchange and interexchange telecommunications services.

	Respectfully submitted, N
By:	1

Ronald Del Sesto, Jr.
Nguyen T. Vu
BINGHAMMCCUTCHENLLP2020 K Street, NW
Washington, DC 20006 (202) 373-6000 (Tel)
(202) 373-6001 (Fax) r.delsesto@bingham.com nguyen.vu@bingham.com

COUNSEL FOR APPLICANT

Dated: November 18, 2009

#### **LIST OF EXHIBITS**

Exhibit A Articles of Incorporation

Exhibit B Authority to Transact Business as a Foreign Corporation in Utah

Exhibit C Financial Statements

(CONFIDENTIAL - FILED UNDER SEAL)

Exhibit **D** Management Biographies

Exhibit E Chart of Accounts

(CONFIDENTIAL - FILED UNDER SEAL)

Exhibit F 3 Year Projection of Expected Operations

(CONFIDENTIAL - FILED UNDER SEAL)

Exhibit G Proposed Local Exchange Tariff

Exhibit H Proposed Intrastate Access Tariff

Verification

### **EXHIBIT A**

Articles of Incorporation



PAGE 1

#### The First State

I, HARRIET SMITH WINDSOR, SECRETARY OF STATE OF THE STATE OF DELAWARE, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT COPY OF THE RESTATED CERTIFICATE OF "INTELEPEER, INC.", FILED IN THIS OFFICE ON THE THIRTY—FIRST DAY OF OCTOBER, A.D. 2008, AT 2 O'CLOCK P.M.

A FILED COPY OF THIS CERTIFICATE HAS BEEN FORWARDED TO THE NEW CASTLE COUNTY RECORDER OF DEEDS.

4231218 8100

081083618

DATE: 10-31-08

Harriet Smith Windsor, Secretary of State **AUTHENTICATION:** 6944242

State of Delaware Secretary of State Division of Corporations Delivered 02:01 FM 10/31/2008 FILED 02:00 PM 10/31/2008 SRV 081083618 - 4231218 FILE

# AMENDED & RESTATED CERTIFICATE OF INCORPORATION OF INTELEPEER, INC.

IntelePeer, Inc., a corporation organized and existing under and by virtue of the provisions of the General Corporation Law of the State of Delaware (the ''General Corporation Law''),

#### **DOES HEREBY CERTIFY:**

FIRST: That the name of this Corporation is IntelePeer, Inc. and that this Corporation was originally incorporated pursuant to the General Corporation Law on October 5, 2006 under the name Voex, Inc.

SECOND: That the Board of Directors duly adopted resolutions proposing to amend and restate the Certificate of Incorporation of this Corporation, declaring said amendment and restatement to be advisable and in the best interests of this Corporation and its stockholders, and authorizing the appropriate officers of this Corporation to solicit the consent of the stockholders therefor, which resolution setting forth the proposed amendment and restatement is as follows:

RESOLVED, that the Certificate of Incorporation of this Corporation be amended and restated in its entirety as follows (the ''Restated Certificate''):

#### ARTICLE

The name of this Corporation is IntelePeer, Inc. (the "Corporation").

#### ARTICLE II

The address of the registered office of this Corporation in the State of Delaware is 2711 Centerville Road, Suite 400, in the City of Wilmington, County of Newcastle 19808. The name of its registered agent at such address is Corporation Service Company.

#### **ARTICLE HI**

The nature of the business or purposes to be conducted or promoted is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law.

#### ARTICLE IV

A. <u>Authorization of Stock.</u> This Corporation is authorized to issue two classes of stock to be designated, respectively, common stock and preferred stock. The total number of shares that this Corporation is authorized to issue is 80,000,000 shares of Common Stock, par value \$0.0001 per share (the ''Common Stock'') and 43,760,592 shares of Preferred Stock, par value \$0.0001 per share (the ''Preferred Stock''). The Preferred Stock shall be divided into three series. The first series of Preferred Stock shall consist of 1,280,210 shares and shall be designated ''Series A Preferred Stock.'' The second series of Preferred Stock shall consist of 24,730,382 shares and shall be designated ''Series B Preferred Stock.'' The third series of Preferred Stock shall consist of 17,750,000 shares and shall be designated ''Series C Preferred Stock''.

WES1121559665.8 -1-

Irrespective of any contrary provisions contained in Section 242(bX2) of the General Corporation Law, the number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares then outstanding) by the holders of shares of Common Stock voting together with the holders of shares of Preferred Stock as a single class (on an as-converted to Common Stock basis), and the holders of shares of Common Stock shall not be entitled to a separate class vote with respect thereto.

<u>B.</u> <u>Rights, Preferences and Restrictions of Preferred Stock.</u> The rights, privileges and restrictions granted to and imposed on the Preferred Stock are as set forth below in this Article IV(B).

#### 1. Dividend Provisions.

- The holders of shares of Preferred Stock shall be entitled to receive dividends, out of any assets legally available therefor, prior and in preference to any declaration or payment of any dividend (payable other than in Common Stock or other securities and rights convertible into or entitling the holder thereof to receive, directly or indirectly, additional shares of Common Stock of this Corporation) on the Common Stock of this Corporation, at the applicable Dividend Rate (as defined below), payable when, as and if declared by the Board of Directors. Such dividends shall not be cumulative. Declared but unpaid dividends with respect to a share of Preferred Stock shall, upon conversion of such share to Common Stock, be paid to the extent assets are legally available therefor either in cash or in Common Stock (valued at the fair market value on the date of payment as determined by the Board of Directors of this Corporation). The holders of the outstanding Preferred Stock can waive any dividend preference that such holders shall be entitled to receive under this Section 1 upon the affirmative vote or written consent of the holders of at least 66% of the shares of Preferred Stock then outstanding (voting together as a single class and not as separate series, and on an as-converted basis). For purposes of this subsection 1(a), "Dividend Rate" shall mean (i) \$0.0368 per annum for each share of Series A Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to the Series A Preferred Stock) (ii) \$0.0590 per annum for each share of Series B Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to the Series B Preferred Stock) and (iii) \$0.0848 per annum for each share of Series C Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to the Series C Preferred Stock).
- (b) Upon the Conversion of any shares of Series A Preferred Stock, Series B Preferred Stock or Series C Preferred Stock into Common Stock as provided in Section 4, all then accrued dividends on such shares that are undeclared as of the date of Conversion shall be waived and all Series A Preferred Stock, Series B Preferred Stock and Series C Preferred Stock dividends shall cease to accrue on such converted shares.
- So long as any shares of Series B Preferred Stock or Series C Preferred Stock shall be outstanding, no dividend, whether in cash or property, shall be paid or declared, nor shall any other distribution be made, on any Series A Preferred Stock or Common Stock, nor shall any shares of any Series A Preferred Stock or Common Stock of the Corporation be purchased, redeemed, or otherwise acquired for value by the Corporation (except for acquisitions of Common Stock by the Corporation pursuant to agreements which permit the Corporation to repurchase such shares upon termination of services to the Corporation or in exercise of the Corporation's right of first refusal upon a proposed transfer) until all dividends (set forth in Section 1(a) above) on the Series B Preferred Stock and Series C Preferred Stock shall have been paid or declared and set apart. The provisions of this Section 1(c) shall not, however, apply to (i) a dividend payable in Common Stock, or (ii) any repurchase of any outstanding securities of the Corporation that is approved by the Corporation's Board of Directors, with such

WEST121559665.8 -2-

approval to include the approval of each of the Series B Director (as defined below) and the Series C Director (as defined below).

- (d) So long as any shares of Series A Preferred Stock shall be outstanding, no dividend, whether in cash or property, shall be paid or declared, nor shall any other distribution be made, on any Common Stock, nor shall any shares of any Common Stock of the Corporation be purchased, redeemed, or otherwise acquired for value by the Corporation (except for acquisitions of Common Stock by the Corporation pursuant to agreements which permit the Corporation to repurchase such shares upon termination of services to the Corporation or in exercise of the Corporation's right of first refusal upon a proposed transfer) until all dividends (set forth in Section 1(a) above) on the Series A *Preferred* Stock shall have been paid or declared and set apart. The provisions of this Section 1(d) shall not, however, apply to (i) a dividend payable in Common Stock, or (ii) any repurchase of any outstanding securities of the Corporation that is approved by the Corporation's Board of Directors.
- (e) Subject to the foregoing clauses (a), (b), (c), and (d), after payment of dividends described in Section 1(a), any additional dividends or distributions shall be distributed out of any assets legally available therefor, payable when, as and if declared by the Board of Directors, among all holders of Common Stock and Preferred Stock in proportion to the number of shares of Common Stock that one held and/or would be held by each such holder if all shares of Preferred Stock were converted to Common Stock at the then effective conversion rate.
- 2. <u>Liquidation Preference.</u> hi the event of any Liquidation Event (as defined below), either voluntary or involuntary, distribution of the proceeds of such Liquidation Event (the "*Proceeds*") of this Corporation to the stockholders of this Corporation shall be made in the following manner:
- (a) the holders of the Series C Preferred Stock shall be entitled to receive, prior and in preference to any distribution of the Proceeds to the holders of the Series B Preferred Stock, Series A Preferred Stock and/or the Common Stock, by reason of their ownership of such stock, an amount equal to the Original Issue Price per share (as defined below) of the Series C Preferred Stock, plus all accrued or declared but unpaid dividends on such share, for each share of Series C Preferred Stock then held by them. If, upon the occurrence of such event, the Proceeds thus distributed among the holders of the Series C Preferred Stock shall be insufficient to permit the payment to such holders of the full aforesaid preferential amounts, then the entire Proceeds legally available for distribution shall be distributed ratably among the holders of the Series C Preferred Stock in proportion to the full preferential amount that each such holder is otherwise entitled to receive.
- (b) upon the completion of the distribution to the holders of Series C Preferred required by subsection (a) of this Section 2 and before distribution of any remaining Proceeds to the holders of Series A Preferred Stock and/or the Common Stock or any further distribution of the remaining Proceeds to the holders of Series C Preferred Stock, the holders of the Series B Preferred Stock shall be entitled to receive by reason of their ownership of such stock, an amount equal to the Original Issue Price per share (as defined below) of the Series B Preferred Stock, plus all accrued or declared but unpaid dividends on such share, for each share of Series B Preferred Stock then held by them. If, upon the occurrence of such event, the remaining Proceeds available for distribution among the holders of the Series B Preferred Stock after completion of the distribution to the holders of Series C Preferred Stock required by subsection (a) of this Section 2 shall be insufficient to permit the payment to the holders of the Series B Preferred Stock of the full aforesaid preferential amounts, then the entire remaining Proceeds legally available for distribution after completion of the distribution to the holders of Series C Preferred Stock required by subsection (a) of this Section 2 shall be distributed ratably among the holders of the

WES1121559665.8 -3-

Series B Preferred Stock in proportion to the full preferential amount that each such holder is otherwise entitled to receive.

(c) upon the completion of the distribution to the holders of Series C Preferred Series B Preferred Stock required by subsections (a) and (b) of this Section 2 and before distribution of any remaining Proceeds to the holders of the Common Stock or any further distribution of the remaining Proceeds to the holders of Series C Preferred Stock and Series B Preferred Stock, the holders of the Series A Preferred Stock shall be entitled to receive, by reason of their ownership of such stock, an amount equal to the Original Issue Price per share (as defined below) of the Series A Preferred Stock, plus all accrued or declared but unpaid dividends on such share, for each share of Series A Preferred Stock then held by them. If, upon the occurrence of such event, the remaining Proceeds available for distribution to the Series A Preferred Stock after completion of the distribution to the holders of Series C Preferred Stock and Series B Preferred Stock required by subsections (a) and (b) of this Section 2, shall be insufficient to permit the payment to such holders of the full aforesaid preferential amounts, then such remaining Proceeds legally available for distribution after completion of the distribution to the holders of Series C Preferred Stock and Series B Preferred Stock required by subsections (a) and (b) of this Section 2 shall be distributed ratably among the holders of the Series A Preferred Stock in proportion to the full preferential amount that each such holder is otherwise entitled to receive.

For purposes of this Restated Certificate, "Original Issue Price" for the Series A Preferred Stock the Series B Preferred Stock and the Series C Preferred Stock shall mean \$0.46 per share, \$0.7374 per share, and \$1.0597 per share, respectively (each as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to such series of Preferred Stock).

- (d) Upon completion of the distributions required by subsections (a), (b) and (c) of this Section 2, all of the remaining Proceeds shall be distributed among the holders of Common Stock Preferred Stock pro rata and with equal priority based on the number of shares of Common Stock held by each such holder, with the shares of Preferred Stock being treated for this purpose as if they had been converted to shares of Common Stock at the then applicable Conversion Rate.
- (e) (i) For purposes of this Section 2, a "Liquidation Event" shall include:

  (A) any reorganization by way of share exchange, consolidation or merger, in one transaction or series of transaction transaction "), in which the Corporation is a constituent corporation or related transactions (each, a

is a party with another entity if; as a result of such combination transaction, the voting securities of the Corporation that are outstanding immediately prior to the consummation of such combination transaction (other than any such securities that are held by an "Acquiring Stockholder," as defined below) do not represent. or are not converted into, securities of the surviving entity of such combination transaction (or such surviving entity's parent entity if the surviving entity is owned by the parent entity) that, immediately after the consummation of such combination transaction, together possess at least a majority of the total voting power of all securities of such surviving entity (or its parent entity, if applicable) that are outstanding immediately after the consummation of such combination transaction, including securities of such surviving entity (or its parent entity, if applicable) that are held by the Acquiring Stockholder; (B) a combination transaction in which stockholders of the Corporation sell or otherwise transfer for consideration voting securities of the Corporation that represent at least fifty percent (50%) of the total voting power of all then outstanding securities of the Corporation; (C) a sale, transfer, lease or other disposition of all or substantially all of the assets of the Corporation (including an exclusive license of all or substantially all of the Corporation's intellectual property); or (D) a liquidation, dissolution or winding up of the Corporation. For purposes of this Section 2, an "Acquiring Stockholder" means a stockholder or stockholders of the Corporation that (i) merges or combines with the Corporation in such combination transaction or (ii) owns or controls a majority of the voting power of another entity that merges or combines with the Corporation in

VVESTU1559665.8 -4-

For purposes of Section 2(e)(i)(13), a "combination transaction" shall not include a financing effected by the Corporation for capital raising purposes. The treatment of any particular transaction or series of related transactions as a Liquidation Event may be waived by the vote or written consent of the holders of at least 66% of the outstanding Series B and Series C Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis).

- (ii) In any Liquidation Event, if any portion of the Proceeds received by this Corporation or its stockholders is other than cash, its value will be deemed its fair market value as determined in good faith by the Board of Directors of this Corporation (including the approval of the Series B Director (as defined below) and Series C Director (as defined below)), unless otherwise determined pursuant to the definitive agreement governing such transaction. Any securities shall be valued as follows:
- (A) Securities not subject to investment letter or other similar restrictions on free marketability covered by (B) below:
- (1) If traded on a securities exchange or through the Nasdaq National Market, the value shall be deemed to be the average of the closing prices of the securities on such exchange or system over the thirty (30) trading-day period *ending* three (3) trading days prior to the closing of the Liquidation Event;
- (2) If actively traded over-the-counter, the value shall be deemed to be the average of the closing bid or sale prices (whichever is applicable) over the thirty (30) trading-day period ending three (3) trading days prior to the closing of the Liquidation Event; and
- (3) If there is no active public market, the value shall be the fair market value thereof, as mutually determined in good faith by the Board of Directors of this Corporation (including the approval of the Series B Director and Series C Director).
- (B) The method of valuation of securities subject to investment letter or other restrictions on free marketability (other than restrictions arising solely by virtue of a stockholder's status as an affiliate or former affiliate) shall be to make an appropriate discount from the market value determined as above in (A) (1), (2) or (3) to reflect the approximate fair market value thereof, as determined in good faith by the Board of Directors of the Corporation (including the approval of the Series B Director and Series C Director).
- have received all amounts due to them under Sections 2(a), 2(b) and 2(c) above, the Proceeds distributed or distributable to such holders of Preferred Stock shall include only cash and other property which such holders of Preferred Stock receive upon the closing of the transaction constituting a Liquidation Event under Section 2(e)(i), and which cash and other property is not subject to an "earn out" or similar contingency, or subject to escrow or similar risk of forfeiture (the "Liquidation Event Closing Proceeds"). If subsequently the holders of Preferred Stock receive additional cash and other property through an "earn out" or a release upon the occurrence of a contingency, and which additional cash and other property is not subject to a risk of forfeiture (the "Liquidation Event Post-Closing Proceeds"), then at the time the holders of Preferred Stock receive the Liquidation Event Post-Closing Proceeds such Proceeds will be deemed "received" by the holders of Preferred Stock under Sections 2(a), 2(b) and 2(c) above.
- (iv) <u>Notice of Transaction.</u> The Corporation shall give each holder of record of Series A Preferred Stock, Series B Preferred Stock and Series C Preferred Stock prior written notice of

WEST121559665.8 -5-

any transaction described in Section 2(eXi) in connection with materials delivered to stockholders in connection with the approval of such transaction not later than fifteen (15) days prior to the stockholders' meeting called to approve such transaction, or fifteen (15) days prior to the closing of such transaction, whichever is earlier, and shall also notify such holders in writing of the final approval of such transaction. The first of such notices shall describe the material terms and conditions of the impending transaction and the provisions of this Section 2, and the Corporation shall thereafter give such holders prompt notice of any material changes. The transaction shall in no event take place sooner than fifteen (15) days after the Corporation has given the first notice provided for herein or sooner than ten (10) days after this corporation has given notice of any material changes provided for herein.

(b) <u>Waiver of Notice.</u> The holders of at least 66% of the outstanding shares of Preferred Stock may, at any time upon written notice to the Corporation, waive any notice provisions specified herein for the benefit of such holders, and any such waiver shall be binding upon the holders of all such securities.

#### 3 Redemption.

- (a) Redemption Request. Subject to the terms and conditions of this Section 3 and subject to any liquidation preference rights which may have been previously invoked under Section 2 hereof to the extent that any outstanding shares of Series C Preferred Stock and Series B Preferred Stock have not been redeemed or converted into Common Stock at least three (3) days prior to the in\_\_ st date set for redemption, the Corporation shall, upon receiving a written request at any time after September 30, 2013, signed by the holders of at least 66% of the then outstanding shares of Series C Preferred Stock and Series B Preferred Stock (voting together as a single class and not as separate series) to the extent it may lawfully do so, redeem, (a ''Redemption'') on the date three (3) months following its receipt of such written redemption request and on the last day of each calendar quarter thereafter (each referred to hereafter as a ''Redemption Date''), all the number of Series C Preferred Stock and Series B Preferred Stock that are outstanding on the date the Corporation receives such written redemption request. The Series C Redemption Price (as defined below) and the Series B Redemption Price (as defined below) shall be paid from any source of funds legally available therefor, until all outstanding shares of Series C Preferred Stock and Series B Preferred Stock to be redeemed have been redeemed or converted to Common Stock as provided in Section 4 or the request for redemption has been withdrawn or terminated as provided below.
- (b) Withdrawal or Termination of Request. A redemption request may be withdrawn or terminated upon the request of the holders of at least 66% of the issued and outstanding shares of Series C Preferred Stock and Series B Preferred Stock (voting together as a single class and not as separate series) on the date of the request for withdrawal or termination, but only with respect to the shares of Series C Preferred Stock and Series B Preferred Stock that had not been redeemed in full in cash as of such Redemption Date. After any such withdrawn or terminated redemption request, the shares of Series C Preferred Stock and Series B Preferred Stock shall again be subject to redemption pursuant to this Section 3 upon the request of the holders of Series C Preferred Stock and Series B Preferred Stock as provided above.
- (c) Redemption Price. Upon a Redemption of Series C Preferred Stock and the Series 13 Preferred Stock, the Corporation shall pay in cash to the holder of a redeemed share a sum equal to the Original Issue Price multiplied by two, plus declared but unpaid dividends (the ''Redemption Price''). The Redemption Price shall be paid in cash.
- (d) Holder Notice. At least fifteen (15) but no more than thirty (30) days prior to the Redemption Date, if the holders of Series C Preferred Stock and Series C Preferred Stock exercise their right of Redemption pursuant to Section 3(a) above, written notice shall be mailed, first class postage

WESTUI559665.8 **-6-**

prepaid, to each holder of rec	ord (at the close of bus	iness on the business da	ny next preceding the day on

which notice is given) of the Series C Preferred Stock and Series B Preferred Stock to be redeemed, at the address last shown on the records of the Corporation for such holder, notifying such holder of the Redemption to be effected, specifying the number of shares to be redeemed from such holder, the Redemption Date, the Redemption Price the manner in which payment shall be obtained, and calling upon such holder to surrender to this Corporation, in the manner and at the place designated, such holder's certificate or certificates representing the shares to be redeemed (the "Holder Notice"). Except as provided in Section 3(e), each holder of (i) Series C Preferred Stock and (ii) Series B Preferred Stock to be redeemed shall surrender to this Corporation on or after the Redemption Date the certificate or certificates representing such shares, in the manner and at the place designated in the Holder Notice, and thereupon the Redemption Price of such shares shall be payable to the order of the person or entity whose name appears on such certificate or certificates as the owner thereof in the manner specified in Section 3(c), and each surrendered certificate shall be cancelled. In the event less than all the shares represented by any such certificate are redeemed, a new certificate shall be issued representing the unredeemed shares.

- Rights. From and after the Redemption Date, unless there shall have been a default in payment of the Redemption Price, all rights of the holders of shares of Series C Preferred Stock and Series B Preferred Stock designated for Redemption in the Holder Notice as holders of Series C Preferred Stock and Series B Preferred Stock (except the right to receive the Redemption Price, upon surrender of their certificate or certificates) shall cease with respect to such shares, and such shares shall not thereafter be transferred on the books of this Corporation or be deemed to be outstanding for any purpose whatsoever. If the funds of this Corporation legally available for Redemption of shares of Series C Preferred Stock and Series B Preferred Stock on the Redemption Date are insufficient to redeem the total number of shares of Series C Preferred Stock and Series B Preferred Stock to be redeemed on such date, (i) those funds that are legally available shall be used to redeem the maximum possible number of such shares ratably among the holders of such shares to be redeemed in proportion to the amounts that the Series C Preferred Stock and Series B Preferred Stock would otherwise have been entitled to receive if all amounts payable on or with respect to such Series C Preferred Stock and Series B Preferred Stock in such Redemption bad been paid in full and (ii) the Corporation will make best efforts to pursue a capital raising transaction to allow it to complete such Redemption. If the Corporation is unable to raise enough capital to complete the Redemption, the Corporation may complete such Redemption in thirty-six equal monthly installments, including interest at 13% per annum. The shares of Series C Preferred Stock and Series B Preferred Stock not redeemed shall remain outstanding and be entitled to all the rights and preferences provided
- 4. <u>Conversion.</u> The holders of the Preferred Stock shall have conversion rights as follows (the ''Conversion Rights''):
- (a) Right to Convert. Each share of Preferred Stock shall be convertible, at the option of the holder thereof, at any time after the date of issuance of such share, at the office of this Corporation or any transfer agent for such stock, into such number of fully paid and nonassessable shares of Common Stock as is determined by dividing the applicable Original Issue Price for such series by the applicable Conversion Price for such series (the conversion rate for a series of Preferred Stock into Common Stock is referred to herein as the "Conversion Rate" for such series), determined as hereafter provided, in effect on the date the certificate is surrendered for conversion. The initial Conversion Price per share for each series of Preferred Stock shall be the Original Issue Price applicable to such series; provided, however, that the Conversion Price for the Preferred Stock shall be subject to adjustment as set forth in subsection 4(d).
- (b) <u>Automatic Conversion.</u> Each share of Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such series *Preferred* Stock immediately upon the this Corporation's sale of its Common Stock in a firm commitment

WEST121559665.8 -7-

Securities Act of 1933, as amended, that results in aggregate gross proceeds to the Corporation, net of underwriting expenses, in excess of \$50,000,000 (a "Qualified Public Offering") at an offering price per share equal to three times (3x) the Original Issue Price for the Series C Preferred. Additionally, each share of Preferred Stock shall automatically be converted into shares of Common Stock at the respective Conversion Rates at the time in effect for such series of Preferred Stock immediately upon the date specified by written consent or agreement of the holders of at least 66% of the then outstanding shares of Series B Preferred Stock and Series C Preferred Stock (voting together as a single class and not as a separate series, and on an as-converted basis).

#### Mechanics of Conversion.

- Before any holder of Preferred Stock shall be entitled to voluntarily convert the same into shares of Common Stock, such holder shall surrender the certificate or certificates therefor, duly endorsed, (or shall execute and deliver such reasonable and appropriate documentation, including an affidavit of loss, if such certificate or certificates, are lost, stolen or destroyed) at the office of the Corporation or of any transfer agent for the Preferred Stock, and shall give written notice to the Corporation at such office that such holder elects to convert the same and shall state therein the name or names in which such holder wishes the certificate for shares of Common Stock to be issued. The Corporation shall, as soon as practicable thereafter, issue and deliver at such office to such holder of Preferred Stock, or to such holder's nominee or nominees, a certificate or certificates for the number of shares of Common Stock to which such holder shall be entitled as aforesaid. Such conversion shall be deemed to have been made immediately prior to the close of business on the date of such surrender of the shares of Preferred Stock to be converted, and the person or persons entitled to receive the shares of Common Stock issuable upon such conversion shall be treated for all purposes as the record holder or holders of such shares of Common Stock on such date. If the conversion is in connection with an underwritten offering of securities registered pursuant to the Securities Act of 1933, as amended, the conversion may, at the option of any holder tendering Preferred Stock for conversion, be conditioned upon the closing with the underwriters of the sale of securities pursuant to such offering, in which event the persons entitled to receive the Common Stock upon conversion of the Preferred Stock shall not be deemed to have converted such Preferred Stock until immediately prior to the closing of such sale of securities.
- (ii) Notwithstanding the foregoing, if any shares of Preferred Stock are converting into shares of Common Stock pursuant to the provisions of Article IV, Section 4(b), the conversion shall occur automatically without any further action by the holders of Preferred Stock affected thereby and whether or not the certificates representing such shares of Preferred Stock are surrendered to the Corporation or any transfer agent for the Preferred Stock. The Corporation shall not be obligated to issue a certificate or certificates evidencing the shares of Common Stock resulting from the automatic conversion unless the certificate or certificates evidencing such shares of Preferred Stock are either delivered to the Corporation or any transfer agent for the Preferred Stock, or the holder of Preferred Stock notifies the Corporation or any transfer agent for the Preferred Stock that such certificate or certificates have been lost, stolen or destroyed and executes an agreement reasonably satisfactory to the Corporation to indemnify the Corporation from any loss incurred by it in connection with such certificate or certificates. The Corporation shall, as soon as practicable after such delivery, or such agreement and indemnification in the case of a lost, stolen or destroyed certificate, issue and deliver to such holder of Preferred Stock a certificate or certificates for the number of shares of Common Stock to which such holder shall be entitled as aforesaid.
- (d) <u>Conversion Price Adjustments of Preferred Stock for Certain Dilutive Issuances.</u>
  <u>Splits and Combinations.</u> The Conversion Price of the Preferred Stock shall be subject to adjustment from time to time as follows:

WES7121559665.8 -8-

(i)

- **(A)** If this Corporation shall issue, on or after the date upon which a share of Preferred Stock was first issued (the "Effective Date"), any Additional Stock (as defined below) without consideration or for a consideration per share less than the Conversion Price applicable to a series of Preferred Stock in effect immediately prior to the issuance of such Additional Stock, the Conversion Price for such series in effect immediately prior to each such issuance shall forthwith (except as otherwise provided in this clause (i)) be adjusted to a price determined by multiplying such Conversion Price by a fraction: (x) the numerator of which is equal to the number of shares of Common Stock Outstanding (as defined below) immediately prior to such issuance plus the *number* of shares of Common Stock which the aggregate consideration received by this Corporation for the total number of Additional Stock so issued would purchase at the Conversion Price in effect immediately prior to such issuance; and (y) the denominator of which is equal to the number of shares of Common Stock Outstanding (as defined below) prior to such issuance plus the number of Additional Stock so issued. For purposes of this Section 4(dXIXA), the term ''Common Stock Outstanding'' shall mean and include the following: (1) outstanding Common Stock, (2) Common Stock issuable upon conversion of outstanding Preferred Stock, (3) Common Stock issuable upon exercise of outstanding stock options and (4) Common Stock issuable upon exercise (and, in the case of warrants to purchase Preferred Stock, conversion) of outstanding warrants. Shares described in (1) through (4) above shall be included whether vested or unvested, whether contingent or non-contingent, and whether exercisable or not yet exercisable.
- (B) No adjustment of the Conversion Price for the Preferred Stock shall be made in an amount less than one cent per share, provided that any adjustments that are not required to be made by reason of this sentence shall be carried forward and shall be either taken into account in any subsequent adjustment made prior to three (3) years from the date of the event giving rise to the adjustment being carried forward, or shall be made at the end of three (3) years from the date of the event giving rise to the adjustment being carried forward. Except to the limited extent provided for in subsections (EX3) and (E)(4), no adjustment of such Conversion Price pursuant to this subsection 4(d)(i) shall have the effect of increasing the Conversion Price above the Conversion Price in effect immediately prior to such adjustment
- (C) In the case of the issuance of Common Stock for cash, the consideration shall be deemed to be the amount of cash paid therefor before deducting any reasonable discounts, commissions or other expenses allowed, paid or incurred by this Corporation for any underwriting or otherwise in connection with the issuance and sale thereof.
- (D) In the case of the issuance of the Common Stock for a consideration in whole or in part other than cash, the consideration other than cash shall be deemed to be the fair market value thereof as determined by the Board of Directors (including the approval of the Series B Director and the Series C Director) irrespective of any accounting treatment.
- (E) In the case of the issuance of options to purchase or rights to subscribe for Common Stock, securities by their terms convertible into or exchangeable for Common Stock or options to purchase or rights to subscribe for such convertible or exchangeable securities, the following provisions shall apply for purposes of determining the number of shares of Additional Stock issued and the consideration paid therefore:
- (1) The aggregate maximum number of shares of Common Stock deliverable upon exercise (assuming the satisfaction of any conditions to exercisability, including without limitation, the passage of time, but without taking into account potential antidilution adjustments) of such options to purchase or rights to subscribe for Common Stock shall be deemed to have been issued

WEST121559665.8 -9-

at the time such options or rights were issued and for a consideration equal to the consideration (determined in the manner provided in subsections 4(d)(i)(C) and (dXiXD)), if any, received by this Corporation upon the issuance of such options or rights plus the minimum exercise price provided in such options or rights (without taking into account potential antidilution adjustments) for the Common Stock covered thereby.

Stock deliverable upon conversion of, or in exchange (assuming the satisfaction of any conditions to convertibility or exchangeability, including, without limitation, the passage of time, but without taking into account potential antidilution adjustments) for, any such convertible or exchangeable securities or upon the exercise of options to purchase or rights to subscribe for such convertible or exchangeable securities and subsequent conversion or exchange thereof shall be deemed to have been issued at the time such securities were issued or such options or rights were issued and for a consideration equal to the consideration, if any, received by this Corporation for any such securities and related options or rights (excluding any cash received on account of accrued interest or accrued dividends), plus the minimum additional consideration, if any, to be received by this Corporation (without taking into account potential antidilution adjustments) upon the conversion or exchange of such securities or the exercise of any related options or rights (the consideration in each case to be determined in the manner provided in subsections 4(d)(i)(C) and (dXi)(D)).

Common Stock deliverable or in the consideration payable to this Corporation upon exercise of such options or rights or upon conversion of or in exchange for such convertible or exchangeable securities, the Conversion Price of the Preferred Stock, to the extent in any way affected by or computed using such options, rights or securities, shall be recomputed to reflect such change, but no further adjustment shall be made for the actual issuance of Common Stock or any payment of such consideration upon the exercise of any such options or rights or the conversion or exchange of such securities.

(4) Upon the expiration of any such options or rights, the termination of any such rights to convert or exchange or the expiration of any options or rights related to such convertible or exchangeable securities, the Conversion Price of the Preferred Stock, to the extent in any way affected by or computed using such options, rights or securities or options or rights related to such securities, shall be recomputed to reflect the issuance of only the number of shares of Common Stock (and convertible or exchangeable securities that remain in effect) actually issued upon the exercise of such options or rights, upon the conversion or exchange of such securities or upon the exercise of the options or rights related to such securities.

(5) The number of shares of Additional Stock deemed issued and the consideration deemed paid therefor pursuant to subsections 4(dXi)(EX1) and (2) shall be appropriately adjusted to reflect any change, termination or expiration of the type described in either subsection 4(dXiXE)(3) or (4).

(ii) "Additional Stock" shall mean any shares of Common Stock issued (or deemed to have been issued pursuant to subsection 4(d)(i)(E)) by this Corporation on or after the Effective Date other than:

(A) shares of Preferred Stock or Common Stock issued pursuant to a stock split, reverse stock split, reincorporation, reclassification, or combination of the Corporation's capital stock, including shares issued prior to the date hereof in connection with the Corporation's reincorporation in the State of Delaware and the previous reclassification of the Corporation's Series B Preferred Stock

- (B) shares of Common Stock issuable or issued to directors, officers, employees, and consultants of the Corporation directly or pursuant to stock or stock option plans, agreements or other arrangements approved by the Board of Directors of the Corporation (including the Series B Director and the Series C Director);
- (C) shares of capital stock, or options or warrants to purchase capital stock, issued to financial institutions or lessors in connection with commercial credit arrangements, equipment financings, or similar transactions, which issuances are primarily for other than equity financing purposes, and provided that such issuance of options or warrants is approved by the Board of Directors, including the Series B Director and Series C Director;
- (D) shares of Common Stock or Preferred Stock issuable upon exercise of options and warrants outstanding as of the date of filing of this Restated Certificate;
- (E) shares of capital stock or warrants or options to purchase capital stock issued in connection with bona fide acquisitions, mergers, or similar transactions, the terms of which are approved by the Board of Directors, including the Series B Director and Series C Director;
- (F) shares of Series C Preferred Stock issued or issuable pursuant to the that certain Series C Preferred Stock Purchase Agreement dated on or around October 30, 2008, as the same may be amended from time to time pursuant to its terms;
- (0) shares of Common Stock issued or issuable upon conversion of the Corporation's Preferred Stock; and
- $\qquad \qquad \text{(H)} \qquad \text{shares of Common Stock issued or issuable in a Qualified Public Offering.}$

The "Effective .Price" of Additional Stock shall mean the quotient determined by dividing the total number of shares of Additional Stock issued or sold, or deemed to have been issued or sold by the Corporation, into the aggregate consideration received, or deemed to have been received by the Corporation for such issue for such Additional Stock.

- (iii) Adjustment For Stock Splits And Combinations. If the Corporation shall at any time or from time to time after the Effective Date effect a subdivision of the outstanding Common Stock without a corresponding subdivision of the Preferred Stock, the Conversion Price for the applicable series of Preferred Stock in effect immediately before that subdivision shall be proportionately decreased. Conversely, if the Corporation shall at any time or from time to time after the Effective Date combine the outstanding shares of Common Stock into a smaller number of shares without a corresponding combination of the Preferred Stock, the Conversion Price for the applicable series of Preferred Stock in *effect immediately* before the combination shall be proportionately increased. Any adjustment under this Section 4(ii) shall become effective at the close of *business on the date the* subdivision or combination becomes effective.
- (iv) Adjustment For Common Stock Dividends And Distributions. If the Corporation at any time or from time to time after the Effective Date makes, or fixes a record date for the determination of holders of Common Stock entitled to receive, a dividend or other distribution payable in additional shams of Common Stock, in each such event the Conversion Price for the applicable series of Preferred Stock that is then in effect shall be decreased as of the time of such issuance or, in the event such record date is fixed, as of the close of business on such record date, by multiplying the Conversion Price then in effect by a fraction (i) the numerator of which is the total number of shares of

Common Stock issued and outstanding immediately prior to the time of such issuance or the close of business on such record date, and (ii) the denominator of which is the total number of shares of Common Stock issued and outstanding immediately prior to the time of such issuance or me close of business on such record date plus the number of shares of Common Stock issuable in payment of such dividend or distribution: provided, however, that if such record date is fixed and such dividend is not fully paid or if such distribution is not fully made **on** the date fixed therefor, the Conversion Price shall be recomputed accordingly as of the close of business on such record date and thereafter the Conversion Price shall be adjusted pursuant to this Section 4(iii) to reflect the actual payment, if any, of such dividend or distribution.

- (v) Adjustment For Reclassification, Exchange And Substitution. If at any time or from time to time after the Effective Date, the Common Stock issuable upon the conversion of the Preferred Stock is changed into the same or a different number of shares of any class or classes of stock, whether by recapitalization, reclassification or otherwise (other than pursuant to a transaction deemed to be a Liquidation Event pursuant to Section 2(e)(i) or a subdivision or combination of shares or stock dividend or a reorganization, merger, consolidation or sale of assets provided for elsewhere in this Section 4), in any such event each holder of Preferred Stock shall have the right thereafter to convert such stock into the kind and amount of stock and other securities and property receivable upon such recapitalization, classification or other change by holders of the maximum number of shares of Common Stock into which such shares of Preferred Stock could have been converted immediately prior to such recapitalization, reclassification or change, all subject to further adjustment as provided herein or with respect to such other securities or property by the terms thereof.
- (e) Reorganizations, Mergers Or Consolidations. If at any time or from time to time after the Effective Date, there is a capital reorganization of the Common Stock or the merger or consolidation of the Corporation with or into another corporation or another entity or person (other than pursuant to a transaction deemed to be a Liquidation Event pursuant to Section 2(e)(i) or a recapitalization, subdivision, combination, classification, exchange or substitution of shares provided for elsewhere in this Section 4), as a part of such capital reorganization, provision shall be made so that the holders of the Preferred Stock shall thereafter be entitled to receive upon Conversion of such Preferred Stock the number of shares of stock or other securities or property of the Corporation to which a holder of the number of shares of Common Stock deliverable upon Conversion would have been entitled on such capital reorganization, subject to adjustment in respect of such stock or securities by the terms thereof. In any such case, appropriate adjustment shall be made in the application of the provisions of this Section 4 with respect to the rights of the holders of such **Prete**, **led** Stock after the capital reorganization to the end that the provisions of this Section 4 (including adjustment of the Conversion Price then in effect and the number of shares issuable upon Conversion of such Preferred Stock) shall be applicable after that event and be as nearly equivalent as practicable.
- Incorporation or through any reorganization, recapitalization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities, or any other voltmtary action, avoid or seek to avoid the observance or performance of any of the terms to be observed or performed hereunder by the Corporation, but shall at all times in good faith assist in the carrying out of all the provisions of this Section 4 and in the taking of all such action as may be necessary or appropriate in order to protect the Conversion Rights of the holders of Preferred Stock against impairment.
- (g) No Fractional Shares. No fractional shares shall be issued upon the Conversion of any share or shares of Preferred Stock, and the number of shares of Common Stock to be issued shall be rounded to the nearest whole share. Whether or not fractional shams are issuable upon *such* Conversion shall

WEST121559665.8 -12-

be determined on the basis of the total number of shares of Preferred Stock the holder is at the time converting into Common Stock and the number of shares of Common Stock issuable upon such aggregate Conversion.

(h) <u>Certificate As To Adjustment.</u> Upon the occurrence of each adjustment or readjustment of the Conversion Price of Preferred Stock pursuant to this Section 4, the Corporation, at its expense, shall promptly compute such adjustment or readjustment in accordance with the terms hereof and prepare and furnish to each holder of such Preferred Stock a certificate setting forth such adjustment

or

Corporation shall, upon the written request at any time of any holder of Preferred Stock, furnish or cause to be furnished to such holder a like certificate setting forth: (1) *such* adjustment and readjustment, (ii) the Conversion Price for such Preferred Stock at the time in effect, and (iii) the number of shares of Common Stock and the amount, if any, of other property that at the time would *be* received upon the Conversion of a share of such Preferred Stock.

- (i) Notices Of Record Date. In the event of any taking by the Corporation of a reward of the holders of any class of securities for the purpose of determining the holders thereof who are entitled to receive any dividend or other distribution, any right to subscribe for, purchase, or otherwise acquire any shares of stock of any class or any other securities or property, or to receive any other right, the Corporation shall mail to each holder of Preferred Stock at least 10 days' prior written notice specifying the date on which any such record is to be taken for the purpose of such dividend, distribution, or right, and the amount and character of such dividend, distribution, or right. No event described herein shall take place sooner than ten (10) days after this Corporation has given the first notice provided for herein; provided, however, that subject to compliance with the General Corporation Law such periods may be shortened or waived upon the written consent of the holders of Preferred Stock that represent at least 66% of the voting power of all then outstanding shares of such Preferred Stock (voting together as a single class and not as separate series, and on an as-converted to Common Stock basis).
- (j) Reservation Of Stock Issuable Upon Conversion. The Corporation shall at times reserve and keep available out of its authorized but unissued shares of Common Stock, solely for the purpose of effecting the Conversion of shares of Preferred Stock, such number of its shares of Common Stock as shall from time to time be sufficient to effect the Conversion of all outstanding shares of such Preferred Stock; and if at any time the number of authorized but unissued shares of Common Stock shall not be sufficient to effect the Conversion of all the outstanding shares of Preferred Stock, in addition to such other remedies as shall be available to the holder of such Preferred Stock, the Corporation shall take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of Common Stock to such number of shares as shall be sufficient for such purposes, including, without limitation, engaging in best efforts to obtain the requisite stockholder approval of any necessary amendment to this Certificate of Incorporation.
- (k) <u>Notices.</u> Any notice required by the provisions of this Section 4 to be given to the holders of shares of Preferred Stock shall be deemed given if deposited in the United States mail, postage prepaid, and addressed to each holder of record at the holder's address appearing on the books of the **Corporation.**

#### 5. Voting Rights.

(1) <u>General Voting Rights.</u> The holder of each share of Preferred Stock shall the right to one vote for each share of Common Stock into which such Preferred Stock could then be converted, and with respect to such vote, such holder shall have full voting rights and powers equal to the voting rights and powers of the holders of Common Stock, and shall be entitled, notwithstanding any provision hereof, to notice of any stockholders' meeting in accordance with the Bylaws of this

WESR21559665.8 -13-

Corporation, and except as provided in subsection 5(b) below with respect to the election of directors by the separate class vote of the holders of Common Stock or with respect to other matters required by law to be submitted to a class vote, shall be entitled to vote, together with holders of Common Stock, with respect to any question upon which holders of Common Stock have the right to vote. Fractional votes shall not, however, be permitted and any fractional voting rights available on an as-converted basis (after aggregating all shares into which shares of Preferred Stock held by each holder could be converted) shall be rounded to the nearest whole number (with one-half being rounded upward).

#### (m) Voting for the Election of Directors.

- (i) As long as any shares of Series B Preferred Stock remain outstanding, the holders of such shares of Series B Preferred Stock, voting as a separate series, shall be entitled to elect one (1) director of this Corporation at any election of directors (the "Series B Director"). The Series B Director may be removed from the Board only by the affirmative vote of the holders of a majority of the Series B Preferred Stock voting as a separate series.
- (ii) As long as any shares of Series C Preferred Stock remain outstanding, the holders of such shares of Series C Preferred Stock, voting as a separate series, shall be entitled to elect one (1) director of this Corporation at any election of directors (the "Series C Director"). The Series C Director may be removed from the Board only by the affirmative vote of the holders of a majority of the Series C Preferred Stock voting as a separate series.
- (iii) The holders of outstanding Common Stock, voting as a separate class, shall be entitled to elect two (2) directors of this Corporation at any election of directors (the *"Common Directors"*). Each Common Director may be removed from the Board only by the affirmative vote of the holders of a majority of the Common Stock voting as a separate class.
- (iv) The holders of Preferred Stock and Common Stock (voting together as a single class and not as separate series, and on an as-converted basis) shall be entitled to elect three (3) directors of this Corporation (the `*Mutual Directors''*). Each Mutual Director may be removed from the Board only by the affirmative vote of the holders of a majority of the Common Stock and Preferred Stock, voting as a single class and not as separate series, and on an as-converted basis.

Notwithstanding the provisions of Section 223(a)(1) and 223(a)(2) of the General Corporation Law, subject to any agreement among the stockholders of this Corporation, any vacancy, including newly created directorships resulting from any increase in the authorized number of directors or amendment of this Restated Certificate, and vacancies created by removal or resignation of a director, may be filled by a majority of the directors then in office, though less than a quorum, or by a sole remaining director, and the directors so chosen shall hold office until the next annual election and until their successors are duly elected and shall qualify, unless sooner displaced; provided, however, that where such vacancy occurs among the directors elected by the holders of a class or series of stock, the consent of a majority-in-interest of the holders of shares of such class or series shall be required prior to any such action by the Board.

#### 6. Protective Provisions.

(n) The Corporation shall not (by amendment, merger, consolidation or otherwise) without first obtaining the approval (by vote or written consent, as provided by law) of the holders of at least 66% of the then outstanding shares of Series B Preferred Stock and Series C Preferred Stock (voting together as a single class, not as a separate series and on an as converted basis) take any of the following actions:

- (i) amend this Corporation's Restated Certificate or its Bylaws;
- (ii) authorize or issue or obligate itself to issue, any equity security (including any other security convertible into or exercisable for any such equity security) except for compensatory issuances and in lease, borrowing and comparable transactions, in each such case, without first obtaining the approval of (i) the Series C Director and (ii) the Series B Director;
- (iii) redeem, purchase or otherwise acquire (or pay into or set aside for a sinking fund for such purpose) any share or shares of Preferred Stock or Common Stock; provided, however, that this restriction shall not apply to the repurchase of shares at cost of Common Stock from employees, officers, directors, consultants or other persons performing services for this Corporation or any subsidiary pursuant to agreements under which this Corporation has the option to repurchase such shares upon the occurrence of certain events, such as the termination of employment or service, or pursuant to a right of first refusal, with the approval of the Series C Director and the Series B Director;
- (iv) declare or pay a dividend to or distribute any property to holders of Common Stock or holders of Series A Preferred Stock, except as required in connection with a Liquidation Event;
- (v) authorize the liquidation, dissolution, recapitalization, or filing for bankruptcy of this Corporation;
- (vi) prior to October 31, 2011, consummate a Liquidation Event, unless the per share proceeds distributable to the holders of Series C Preferred Stock, free of any restrictions or contingencies (including, but not limited to, any proceeds subject to an "earn out" or similar contingency, or subject to escrow or similar risk of forfeiture), is equal to or exceeds three times (3x) the Original Issuance Price per share of Series C Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to the Series C Preferred Stock):
  - (vii) materially change the nature of the Corporation's business;
- (viii) change the authorized number of directors of this Corporation or delegate substantial board authority to committees;
- (ix) authorize the issuance of any securities by a subsidiary of the Corporation to any other person other than the Corporation;
- (x) authorize the increase of the number of shares issuable under any stock option or purchase plan without first obtaining the approval of (i) the Series C Director and (ii) the Series B Director;
- $\hbox{ (xi)} \qquad \text{transact with an affiliate or stockholder other than in the ordinary course of business; or } \\$
- (xii) authorize or issue debt instruments that obligate the Corporation to repay more than \$5,000,000 of principal.
- (0) The Corporation shall not (by amendment, merger, consolidation or otherwise) without first obtaining the approval (by vote or written consent, as provided by law) of the holders of 65% of the then outstanding shares of Series B Preferred Stock (i) alter or change the rights, preferences

WEST121559665.8 -15-

privileges of the shares of the Series B Preferred Stock in a manner that would materially and adversely affect such shares in a manner different than the Series C Preferred Stock or (ii) authorize additional shares of Series B Preferred Stock.

- (p) The Corporation shall not (by amendment, merger, consolidation or otherwise) without first obtaining the approval (by vote or written consent, as provided by law) of the holders of a majority of the then outstanding shares of Series C Preferred Stock (i) alter or change the rights, preferences or privileges of the shares of the Series C Preferred Stock in a manner that would materially and adversely affect such shares in a manner different than the Series B Preferred Stock or (ii) authorize additional shares of Series C Preferred Stock.
- 7. <u>Status of Converted Stock.</u> In the event any shares of Preferred Stock shall be converted pursuant to Section 4 hereof, the shares so converted shall be cancelled and shall not be issuable by this Corporation. The Restated Certificate of this Corporation shall be appropriately amended to effect the corresponding reduction in this Corporation's authorized capital stock.
- C. <u>Common Stock.</u> The rights, preferences, privileges and restrictions granted to and imposed on the Common Stock are as set forth below in this Article 1V(C).
- 1. <u>Dividend Rights.</u> Subject to the prior rights of holders of all classes of stock at the time outstanding having prior rights as to dividends, the holders of the Common Stock shall be entitled to receive, when, as and if declared by the Board of Directors, out of any assets of this Corporation legally available therefor, any dividends as may be declared from time to time by the Board of Directors.
- <u>2. Liquidation Rights.</u> Upon the liquidation, dissolution or winding up of this Corporation, the assets of this Corporation shall be distributed as provided in Section 2 of Article IV(B) hereof.
  - 3. Redemption. The Common Stock is not redeemable at the option of the holder.
- 4. Voting Rights. The holder of each share of Common Stock shall have the right to one vote for each such share, and shall be entitled to notice of any stockholders' meeting in accordance with the Bylaws of this Corporation, and shall be entitled to vote upon such matters and in such manner as may be provided by Iaw.

#### ARTICLE V

In furtherance and not in limitation of the powers conferred by the laws of the State of Delaware:

- A. The Board of Directors of this Corporation is expressly authorized to adopt, amend or repeal the Bylaws of the Corporation.
- B. Elections of directors need not be by written ballot unless the Bylaws of the Corporation shall so provide.
- C. The books of the Corporation may be kept at such place within or without the State of Delaware as the Bylaws of the Corporation may provide or as may be designated from time to time by the Board of Directors ofihe Corporation.

#### **ARTICLE VI**

Whenever a compromise or arrangement is proposed between Corporation and its creditors or any class of them and/or between the Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of the Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for the Corporation under the provisions of §291 of Title 8 of the General Corporation Law or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under §279 of Title 8 of the General Corporation Law order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of the Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three-fourths in value of creditors or class of creditors, and/or of the stockholders or class of stockholders of the Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of the Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of the Corporation, as the case may be, and also on the Corporation.

#### **ARTICLE** VU

A director of the Corporation shall not be liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to me extent that exculpation from liability is not permitted under the General Corporation Law as now in effect or as it may hereafter be amended. No amendment or repeal of this Article VII shall apply to or have any effect on the liability or alleged liability of any director of the Corporation for or with respect to any acts or omissions of such director occurring prior to such amendment or repeal.

#### ARTICLE 'MI

The Corporation shall, to the maximum extent permitted from time to time under the law of the State of Delaware, indemnify and upon request shall advance expenses to any person who is or was a party or is threatened to be made a party to any threatened, pending or completed action, suit, proceeding or claim, whether civil, criminal, administrative or investigative, by reason of the feet that such person is or was or has agreed to be a director or officer of the Corporation or while a director or officer is or was serving at the request of the Corporation as a director, officer, partner, member, trustee, employee or agent of any corporation, partnership, joint venture, trust or other enterprise, including, without limitation, service with respect to employee benefit plans, against expenses (including, without limitation, attorney's fees and expenses), judgments, fines, penalties and amounts paid in settlement incurred in connection with the investigation, preparation to defend or defense of such action, suit, proceeding or claim; provided, however, that the foregoing shall not require the Corporation to indemnify or advance expenses to any person in connection, with, any action, suit; proceeding, claim or counterclaim initiated by or on behalf of such person. Such indemnification and advancement of expenses shall not be exclusive of other indemnification rights arising as a matter of law, under any Bylaw, agreement, vote of directors or stockholders or otherwise, both as to action in such person's official capacity and as to action in another capacity while holding such office, and shall inure to the benefit of the heirs and legal representatives of such person. Any person seeking indemnification under this Article VIII shall be deemed to have met the standard of conduct required for such indemnification unless the contrary shall be established. Any repeal or modification of the foregoing provisions of this Article VIII shall not adversely affect any right or protection of a director or officer of the Corporation with respect to any acts or omissions of such director or officer occurring prior to such repeal or modification.

The Corporation shall have the power to purchase and maintain, at its expense, insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another partnership, joint venture, trust or other enterprise, against any expense, liability or loss asserted against person and incurred by such person in any such capacity, or arising out of such persons status as such, whether or not the Corporation would have the power to indenmify such, person against such expense, liability or loss under the General Corporation Law or the terms of the Restated Certificate.

#### **ARTICLE TX**

To the maximum extent permitted from time to time under the law of the State of Delaware, the Corporation renounces any interest or expectancy of the Corporation in, or in being offered an opportunity to participate in, business opportunities that are from time to time presented to its officers, directors or stockholders or the affiliates of the foregoing, other than those officers, directors, stockholders or affiliates who are employees of the Corporation. No amendment or repeal of this Article IX shall apply to or have any effect on the liability or alleged liability of any such officer, director, stockholder or affiliate for or with respect to any business opportunities of which such officer, director, stockholder or affiliate becomes aware prior to such amendment or repeal.

#### **ARTICLE X**

In accordance with Section 203(b)(1) of the General Corporation Law relating to the application of Section 203 thereof, the Corporation shall not be governed by Section 203 of the General Corporation Law.

#### **ARTICLE XI**

To the maximum extent permitted from time to time under the law of the State of Delaware, the Corporation renounces any interest or expectancy of the Corporation in, or in being offered an opportunity to participate in, business opportunities that are from time to time presented to its officers, directors or stockholders or the affiliates of the foregoing, other than those officers, directors, stockholders or affiliates who are employees of the Corporation. No amendment or repeal of this Article XI shall apply to or have any effect on the liability or alleged liability of any such officer, director, stockholder or affiliate for or with respect to any business opportunities of which such officer, director, stockholder or affiliate becomes aware prior to such amendment or repeal.

#### **ARTICLE XII**

To the extent that this Corporation is subject to Section 2115 of the California General Corporation Law, in connection with repurchases by this Corporation of its Common Stock from employees, officers, directors, advisors, consultants or other persons performing services for this Corporation or any subsidiary pursuant to agreements under which this Corporation has the option to repurchase such shares at cost upon the occurrence of certain events, such as the termination of employment, Sections 502 and 503 of the California Corporations Code shall not apply in all or in part with respect to such repurchases.

#### ARTICLE XIII

The Corporation reserves the right to amend or repeal any provision contained in this Restated Certificate, in the manner now or hereafter prescribed by statute, and all rights conferred upon a stockholder are granted subject to this reservation.

WEST\21559665.8 -18-

**THIRD:** The foregoing amendment and restatement was approved by the holders of the requisite number of shares of said Corporation in accordance with Section 228 of the General Corporation Law.

**FOURTH:** That said Restated Certificate, which restates and integrates and further amends the provisions of this Corporation's Certificate of Incorporation, has been duly adopted in accordance with Sections 242 and 245 of the General Corporation Law.

WES1121559665.8 -19-

IN WITNESS WHEREOF, the undersigned has caused this Amended	and Restated Certificate of
Incorporation to be executed by Andre Simone, its Chief Financial Officer,	this 31st day of October,
2008.	

IntelePeer, Inc.	
/s/ Andre Simone	
Andre Simone, Chief Financial Officer	_

## Delaware PAGE

## ire first State

HARRIET SMITH WINDSOR, SECRETARY OF STATE OF THE STATE OF DE:LANARK, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT COPY OF THE C101171'7CATS OF CORRECTION OF "1112711.11P1M, INC", = RD IN THIS OFFICE ON THE FIFIN DAY OF NOVEMER, A. D. 2008, AT 4:03 O'CLOCK P.

A FILED COPY OF MS CERTIFICATE MS MEN FORWARDED TO THE

NEW CASTLE COUNTY RECORDER or

DEEDS.



Harriet Smith Windsor. Secretary of Stale AUTHENTICATTON: 6953539 DATE: 11-06-08

Mato of iklalvoro or Mato Division | hirra a 7 04/03 £11 11/05/2008/008

# CERTIFICATE OF CORRECTION TO CORRECT ANERROR IN THE AMENDED AND RESTATED CERTDICATE OF INCORPORATION OF INC.

## FILED IN THE OFFICE OF =SECRETARY OF STATE OF THE STATE OF DELAWARE ON OCTOBER 31, 2608

Welcher, Inc., a corporation organized and existing under the laws of the State of Delaware, hereby certifies as follows:

- I. The name of the corporation is IntelePeer,
- 2. An Amended and Restated Cedilla:de of Incorporation (the 'Restated C.edificate) was filed with the Secretary of State of the State of Delaware on October 342008, and said Restated Certificate requires correction as permitted by Section 103 of the General Corporation Law of the State of Delaware.
- 3., The inaccuracy or defect of said Restated Certificate to be corrected is as subsections (1), (iii) and (lv) of Article IV, Section B(5)(m) should have eccpressly provided far ramoval of directors by "vote or written consent" of the applicable stockholders.
  - 4. Article IV, Section B(5)(sit) is corrected to read in its entirety as follows:
- (I) As long as any shams of Series B Preferred Stock =min outstanding, the holders of such shares of Series B Preferred Stock, voting as a separate series, shall be entitled to elect one (1) director of this Corporation at any election of directors (the "Series B Director,. The Series B Director may be removed from the Board only by the affirmative vote or written consent of the holders of a majority of the Series B Preferred Stock voting as a separate series.
- (ii) As long as any shares of Series C Preferred Stock remain outstanding, the holders of such shares of Series C Prekned Stock, voting as a \*sparate\* wade% shall be endtled to elect one (1) director of this Corporation at any election of directors (the \*Werke C Director\*\*). The Series C Director may be removed from the Board only by the affinnative vote or written consent dike holders of a majority of the Series C Preferred Stock voting as a separate series.
- (iii) The holders of outstanding Common Stock, voting *as* a separate class, shall be entitled to elect two (2) directors &Ibis Corporation at any election of (the "Contemn Directors"). Each Common Director may be removed from the Board only by the affirmative vote or written consent of the holders of a majority of the Common Stock voting as a separate class.

(iv) The holders of Preferred Stock and Common Stock (voting together as I single class and not as separate sedge, and on an as-converted basis) shall be ceded to elect three (3) directors of this Corporation (the "Alain" Directors". Each Mutual Ditector may be removed from the Board only by the nahmative vote or written consent of the holders of a majority of the Common Stock and Preferred Stock voting as a single class and net as separate series, and on an as converted basis.

Notwithstanding the provisions of Section 223(aXI) and 223(a)(2) of the General corporation Law, subject to any agreement among the stockholders of this Corporation, any mummy, including newly createddirectorahip's resulting from *any* increase in the authorized Abe!' of dhectors or amendment of this Restated Certificate, and vacancies created by removal er resignation of a director, may be filled by a majority of the directors then in face, though less thso a quorum, or by a sole remaining dhector, and the directors so chosen shall hold office until the next annual election and until their successors are duly elected *and* shall qualify, unless sooner displaced; provided, however, that where such vacancy occurs among the directors elected by the holders of a class or series of stock, the consent of a mtdoft-in-interest dire holders of edit/ of such class or series shall be required prior to any such action by the Board.

[Signahnv Page Follows]

 $IN\ WITNRSS\ NVERREOle$ , hateleher, Inc. has caused this Cerdfioate of Correction to die Amended and Restated Certificate of Incorporation to be signed by its duly authorized officer thi\_s 5th day of November, 2008.

/s/Andy Simime Andre Simone Chlef Financial Officer WITNESS WHEREOF, IntelePeer, Inc. has caused this Certificate of Correction to the Amended and Restated Certificate of Incorporation to be signed by its duly authorized officer this 3rd day of November, 2008.

Andre run6ne

**Chief Financial Officer** 

## **Delaware**

PAGE 1

## The First State

I, JEFFREY W. BULLOCK, SECRETARY OF STATE OF THE STATE OF
DELAWARE, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT
COPY OF THE CERTIFICATE OF AMENDMENT OF "INTELEPEER, INC.",
FILED IN THIS OFFICE ON THE FOURTEENTH DAY OF APRIL, A.D. 2009,
AT 5:15 O'CLOCK P.M.

A FILED COPY OF THIS CERTIFICATE HAS BEEN FORWARDED TO THE NEW CASTLE COUNTY RECORDER OF DEEDS.

4231218 8100

090362614

You **may** verify this certificate online at corp, delaware, groviauthrer, shtel

AUTHEN

Jeffrey W, Bullock, Secretary of State **TION:** 7245392

DATE: 04-14-09

State of Delaware Secretary of State Division of Corporations Delivered 05:19 PM 04/14/2009 FLIED 05:15 PM 04/14/2009 SRV 090362614 - 4231218 FIZZ

#### CERTIFICATE OF AMENDMENT

#### TO THE

#### AMENDED AND RESTATED CKEITNICATE OF INCORPORATION

OF

#### DITELEPF, ER, INC.

IntelePeer, Inc., a corporation organized and existing under the laws of the State of Delaware (the ''Corporation''), hereby certifies as follows:

1. The first paragraph of Section A of Article IV of the Corporation's Amended and Restated Certificate of Incorporation (the "Certificate") presently reads as follows:

"AutbOrization ot Stock, This Corporation is authorized to issue two classes of stock to be designated, respectively, common stock and preferred stock. The total number of shares that this Corporation is authorized to issue is 80,000,000 shares of Common Stock, par value \$0.0001 per share (the "Common Stock") and 43,760,592 shares of Preferred Stock, par value \$0.0001 per share (the "Preferred Stock"). The Preferred Stock shall be divided into three series. The first series of Preferred Stock shall consist of 1,280,210 shares and shall be designated "Series A Preferred Stock," The second series of Preferred Stock shall consist of 24,730,382 shares and shall be designated "Series B Preferred Stock." The third series of Preferred Stock shall consist of 17,750,000 shares and shall be designated "Series C Preferred Stock",

and such first paragraph of Section A of Article IV is hereby amended and restated in its entirety to read as follows:

"Authorization of Stock." This Corporation is authorized to issue two classes of stock to be designated, respectively, common stock and preferred stock. The total number of shares that this Corporation is authorized to issue is 80,000,000 shares of Common Stock, par value \$0.0001 per share (the "Common Stock") and 44,760,592 shares of Preferred Stock, par value \$0,0001 per share (the "Preferred Stock"). The Preferred Stock shall be divided into three series. The first series of Preferred Stock shall consist of 1,280,210 shares and shall be designated "Series A Preferred Stock," The second series of Preferred Stock shall consist of 24,730,382 shares and shall be designated "Series B Preferred Stock." The third series of Preferred Stock shall consist of 18,750,000 shares and shall be designated "Series C Preferred Stock".

- 2. The foregoing Certificate of Amendment to the Certificate has been duly approved by the Board of the Directors of the Corporation in accordance with the provisions of Sections 141 and 242 of the General Corporation Law of Delaware.
- 3. The foregoing Certificate of Amendment to the Certificate has been duly approved by the Corporation's stockholders in accordance with Sections 228 and 242 of the General Corporation Law of Delaware and the Certificate.

State of Delaware	
	Remainder Of Page Intentionally Left Blank.
WBSn21677170.1 335453-000001	

 $4 \qquad \text{The foregoing Certificate of Amendment to the Corporation's Certificate shall be effective on and as of the date of filing of this Certificate of Amendment with the Secretary of State of the state of the secretary of State of S$ 

IN WITNESS WHEREOF, this Certificate of Amendment to the Amended and Restated Certificate of Incorporation has been executed by the Chief Financial Officer of the Corporation this 14th day of April, 2009.

IN	TEL	EP	EER.

By: Andre Simone
Andre Simone
Chief Financial Officer

## EXHIBIT B

Authority to Transact Business as a Foreign Corporation



#### **Utah Department of Commerce**

#### **Division of Corporations & Commercial Code**

160 East 300 South, 2nd Floor, PO Box 146705 Salt Lake City, UT 84114-6705 Service Center: (801) 530-4849 Toll Free: (877) 526-3994 Utah Residents Fax: (801) 530-6438

Web Site: http://www.commerce.utah.gov

09/14/2009 7450308-014309142009-615385

## **CERTIFICATE OF EXISTENCE**

**Registration Number:** 7450308-0143

Business Name: INTELEPEER, INC. Registered Date: August 31, 2009

**Entity Type:** Corporation - Foreign - Profit

Current Status: Good Standing

The Division of Corporations and Commercial Code of the State of Utah, custodian of the records of business registrations, certifies that the business entity on this certificate is authorized to transact business and was duly registered under the laws of the State of Utah. The Division also certifies that this entity has paid all fees and penalties owed to this state; its most recent annual report has been filed by the Division (unless Delinquent); and, that Articles of Dissolution have not been filed.



o Bury
Kathy Berg

Director
Division of Corporations and Commercial Code

## **EXHIBIT C**

## **Financial Statements**

## CONFIDENTIAL - FILED UNDER SEAL

## EXHIBIT D

Management Biographies

#### **MANAGEMENT BIOGRAPHIES**

## Haydar Haba, Founder & Chief Visionary Officer

An established expert in the telecommunications industry, Mr. Haba founded the company after a thorough study of the issues limiting global communication and realizing the future of SIP-based VoIP communications.

Mr. Haba has over 18 years experience in the communications industry. Prior to IntelePeerTM, Mr. Haba was Founder and Chief Technology Officer of Telco 214, an emerging leader in the international voice market that was successful in winning large market share from AT&T, MCI and Sprint. Mr. Haba began his career at Symetrics Industries, a publicly traded defense communications and computer telephony integrator.

Mr. Haba is recognized internationally as an innovative and revolutionizing force within the communications industry. His knowledge and opinions on global communications and the future of VoIP have been featured commentary in print and broadcast media. Mr. Haba has completed PhD course work in Electrical Engineering and holds Bachelor of Science and Masters degrees from Florida Institute of Technology.

#### Frank Fawzi, Chief Executive Officer & Chairman of the Board

In 2007, Frank M. Fawzi joined IntelePeer as CEO. Mr. Fawzi is the guiding force in supporting the company's growth efforts in its drive to be the leading innovator of communications services and applications for top-tier carriers, voice service providers, enterprise software and web companies.

Between 1991 and 2001, Mr. Fawzi founded, built, and subsequently sold CommTech Corporation, a leader in the communications software sector, to ADC, a leading provider of broadband solutions for the telecommunication industry for \$178M. He also served on the Communications Board of the New Jersey Technology Council, which named CommTech #1 mid-stage company of the year in 1997.

Prior to founding CommTech, Mr. Fawzi worked with AT&T Bell Laboratories and other divisions of AT&T. While at AT&T, he was the lead data communications architect on a team responsible for AT&T being awarded a \$1.4 billion contract from the U.S. Internal Revenue Service (IRS).

Mr. Fawzi received a Bachelor of Science in Engineering and Computer Science and a Masters of Science in Management Information Systems from Stevens Institute of Technology, and has participated in the Wharton Executive Management Program at the University of Pennsylvania.

## **Andre Simone, Chief Financial Officer**

Mr. Simone was CFO at Dimatix which was acquired by Fuji Film. Previously he had been CFO at SAP Markets Inc., which was sold to SAP; ACTA Technology, where he completed private financing and prepared the company for an IPO and subsequent acquisition by Business Objects; and ADAC Laboratories, a publicly traded company later acquired by Philips, where he was responsible for acquisitions, investor relations and finance.

Earlier in his career, Simone held a variety of positions including foreign currency manager at Hewlett Packard and consultant at Bain & Company.

Simone holds an MBA from the University of Pennsylvania's Wharton School and a B.S. degree from Stanford University with Distinction.

## EXHIBIT E

Chart of Accounts

## CONFIDENTIAL - FILED UNDER SEAL

## **EXHIBIT F**

3 Year Projection of Expected Operations

CONFIDENTIAL - FILED UNDER SEAL

## EXHIBIT G

Proposed Local Exchange Tariff

## Tariff Schedule Applicable to

Resold and Facilities-Based

Competitive Local Exchange and Interexchange Services

Furnished by

IntelePeer, Inc.

Between Points Within the State of Utah

This tariff contains the descriptions, regulations, and rates applicable to the furnishing of service for telecommunications services provided by IntelePeer, Inc. with principal offices at 2855 Campus Drive, Suite 200, San Mateo, CA 94403. This tariff applies for services furnished within the State of Utah. This tariff is on file with the Public Service Commission of Utah, and copies may be inspected, during normal business hours, at the Company's principal place of business.

Issued:	, 2009	Effective:	, 2009

Issued By:

## **CHECK SHEET**

The sheets of this tariff inclusive herein are effective as of the date shown at the bottom of the respective sheet(s). Original and revised sheets as named below comprise all changes from the original tariff and are currently in effect as of the date on the bottom of this sheet.

<u>Page</u>	Number of Revision	<u>Page</u>	Number of Revision
1	Original	28	Original
2	Original	29	Original
3	Original	30	Original
4	Original	31	Original
5	Original	32	Original
6	Original	33	Original
7	Original	34	Original
8	Original	35	Original
9	Original	36	Original
10	Original	37	Original
11	Original	38	Original
12	Original	39	Original
13	Original	40	Original
14	Original	41	Original
15	Original	42	Original
16	Original	43	Original
17	Original	44	Original
18	Original	45	Original
19	Original	46	Original
20	Original	47	Original
21	Original	48	Original
22	Original	49	Original
23	Original	50	Original
24	Original	51	Original
25	Original	52	Original
26	Original	53	Original
27	Original		-

Issued By:

## **Table of Contents**

Title Sheet	
Check Sheet	
Table Of Contents	2
Explanation of Symbols Sheet	
Tariff Format	4
Section 1. —Technical Terms and Abbreviations	5
Section 2. — Rules and Regulations	7
Section 3. — Description of Service	38
Section 4. — Rates	43
Section 5 IntraLATA Toll Presubscription	48

Issued By:

	EXPLANATION OF SYMBOLS
(C) —	To signify a changed listing, rule or condition which may affect rates or charges
(D) —	To signify discontinued material, including listing, rate, rule or condition
(1) –	To signify an increase in a rate
(L) –	To signify material relocated from or to another part of the tariff schedules with no change in text, rate, rule or condition.
(N) —	To signify new material including listing, rate rule or condition
(R) —	To signify a reduction in a rate
(T)—	To signify a change in wording of text but no change in rate, rule or condition

Issued:	, 2009	Effective:	, 2009

Issued By:

## **TARIFF FORMAT**

- <u>A.</u> Page Numbering Page numbers appear in the upper right corner of the page. Pages are numbered sequentially. However, new pages are occasionally added to the tariff. When a new page is added between pages already in effect, a decimal is added. For example, a new page added between Page 14 and 15 would be 14.1.
- <u>B.</u> Page Revision Numbers Revision numbers also appear in the upper right corner of each page. These numbers are used to determine the most current page version on file with the Commission. For example, the 4<sup>th</sup> revised Page 14 cancels the 3<sup>rd</sup> revised Page 14.
- C. **Paragraph Numbering Sequence -** There are various levels of paragraph coding. Each level of coding is subservient to its next higher level:

2.1

2.1.1

2.1.1

<u>D.</u> Check Sheets - When a tariff filing is made with the Commission/, an updated Check Sheet accompanies the tariff filing. The Check Sheet lists the pages contained in the tariff, with a cross-reference to the current revision number. When new pages are added, the Check Sheet is changed to reflect the revision. An asterisk designates all revisions made in a given filing (\*). There will be no other symbols used on this page if these are the only changes made to it (i.e., the format, etc. remain the same, just revised revision levels on some pages.) The tariff user should refer to the latest Check Sheet to find out if a particular page is the most current on file with the Commission.

Issued:, 2009	Effective:	, 2009

Issued By:

#### Section 1. TECHNICAL TERMS AND ABBREVIATIONS

- "Applicant" refers to an individual, partnership, corporation, association, or government agency who applies to the Company for any new or additional telephone service.
- "Business Hours" refers to the time after 8:00 A.M. and before 5:00 P.M., Monday through Friday excluding holidays
- "Company" refers to IntelePeer, Inc.
- "Commission" or to the Public Service Commission of Utah
- "Completed call" is a call which the Company's network has determined has been answered by a person, answering machine, fax machine, computer modem device, or other answering device.
- "Customer" refers to the firm, company, corporation, or other entity that contracts for service under this tariff and that is responsible for the payment of charges as well as compliance with the Company's regulations pursuant to this tariff.
- "Delinquent or Delinquency" refers to an account for which payment has not been paid in full on or before the last day for timely payment.
- "Division" refers to the Public Utilities Division.

Issued:	, 2009		Effective:	, 2	2009	

Issued By:

## Section 1. TECHNICAL TERMS AND ABBREVIATIONS(Cont'd)

- "Local Access Transport Area ("LATA")" refers to a geographical area established by the U.S. District Court for the District of Columbia in Civil Action No. 820192.
- "Local Exchange Carrier/LEC" refers to a telecommunications corporation that provides the local access line services within the geographic territory authorized by the Commission.
- "Nonrecurring Charges" refer to a one-time charge associated with given service or item of equipment which applies on a per-service and/or per item basis each time the service or item of equipment is provided.
- "Service" refers to any telecommunications service(s) provided by the Company under this tariff.
- "Station" refers to a telephone instrument consisting of a connected transmitter, receiver, and associated apparatus to permit sending or receiving telephone messages.
- "Tariff' refers to the entire body of rates, tolls, rentals, charges classifications and rules collectively enforced by the utility, although the book or volumes incorporating the same may consist of one or more sheets applicable to distinct service classifications.
- "Telephone Numbers" refer to The North American Numbering Plan [NPA-NXX-XXXX] numbers assigned to customers and used in conjunction with the Services provided pursuant to this tariff.

Issued: , 2009	Effective:	
Issued By:		

#### **SECTION 2. RULES AND REGULATIONS**

- 2.1 Undertaking of the Company
  - 2.1.1 This tariff contains the regulations, rates and charges applicable to facilities-based and resold interexchange and/or local exchange services provided by the Company that originate and terminate within the State of Utah.
  - 2.1.2 The Company installs (or may utilize vendors service to install), operates and maintains the communications services provided herein in accordance with the terms and conditions set forth under this tariff. It may act as the customer's agent for ordering access connection facilities provided by other carriers or entities when authorized by the customer, to allow connection of a customer's location to the Company's network. The customer shall be responsible for all charges due for such service arrangement.
  - 2.1.3 The Company's services are provided on a monthly basis unless ordered on a longer term basis, and are available twenty-four hours per day, seven days per week.
  - 2.1.4 The Company's services are available to business/non-residential customers.

Issued:	. 2009	Effective:	, 2009

Issued By:

## **SECTION 2. RULES AND REGULATIONS**

2.2	Limitations	
<b>∠</b> .∠		

- 2.2.1 Service is offered subject to the provisions of this tariff.
- 2.2.2 The Company's reserves the right to discontinue furnishing service, or limit the use of service necessitated by conditions beyond its control: or when the customer is using service in violation of the law or the provisions of this tariff.
- 2.2.3 All regulations and conditions contained in this tariff shall apply to all such permitted assignees or transferees, as well as all conditions for service.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 2.3 Obligations of the Customer
  - 2.3.1 The customer shall be responsible for:
    - 2.3.1.1The payment of all applicable charges pursuant to this tariff.
    - 2.3.1.2 Reimbursing the Company for damage to, or loss of the Company's equipment caused by the acts or omissions of the customer; or the noncompliance by the customer, with these regulations, or by fire or theft or other casualty on the customer's premises unless caused by the negligence or willful misconduct of the employees or agents of the Company.
      - 2.3.1.3Providing at no charge, as specified from time to time by the Company, any needed space and power to operate the Company's equipment installed on the customer's premises.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 2.3 Obligations of the Customer (Cont'd)
  - 2.3.1.4 Complying with all laws and regulations regarding the working conditions on the premises at which the Company's employees and agents shall be installing or maintaining the Company's equipment. The customer may be required to install and maintain the Company's equipment within a hazardous area if, in the Company's opinion, injury or damage to the Company's employees or property might result from installation or maintenance by the Company. The customer shall be responsible for identifying, monitoring, removing and disposing of any hazardous material prior to any installation work.
  - 2.3.1.5 Complying with all laws and regulations applicable to, and obtaining all consents, approvals, licenses and permits as may be required with respect to, the location of the Company's equipment in any customer premises for the purpose of installing, inspecting, maintaining, repairing, or upon termination of service as stated herein, removing the equipment of the Company.
  - 2.3.1.6 Making Company equipment available periodically for maintenance purposes at a time agreeable to both the Company and the customer. No allowance for interruptions in service will be made for the period during which service is interrupted for such purposes.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 2.3 Obligations of the Customer (Cont'd)
  - 2.3.2 With respect to any service provided by the Company, the customer shall indemnify, defend and hold harmless the Company from all claims, actions, damages, liabilities, costs and expenses for:
    - 2.3.2.1 Any loss, destruction or damage to property of the Company or any third party, or injury to persons, including, but not limited to, employees or invitees of either the Company or the customer, to the extent caused by or resulting from the negligent or intentional act or omission of the customer, its employees, agents, representatives or invitees; or
    - 2.3.2.2 Any claim, loss, damage, expense or liability for infringement of any copyright, patent, trade secret, or any proprietary infringement of any copyright, patent, trade secret, or any proprietary or intellectual property right of any third party, arising from any act or omission by the customer.
  - 2.3.3 The customer is responsible for ensuring that customer-provided equipment connected to Company equipment is compatible with such equipment. The connection, operation, testing, or maintenance of such equipment shall be such as not to cause damage to the Company—provided equipment or injury to the Company's employees or other persons. Any additional protective equipment required to prevent such damage or injury shall be provided by the Company at the customer's expense.
  - 2.3.4 The Company's services may be connected to the services or other communications carriers only when authorized by, and in accordance with, the terms and conditions of the tariffs or contracts which are applicable to such connections.

Issued: , 2009	Effective:	<u>,</u> 2009
Issued By:		

- 2.3 Obligations of the Customer (Cont'd)
  - 2.3.5 Upon reasonable notification to the customer, and at a reasonable time, the Company may make such tests and inspections as may be necessary to determine that the customer is complying with the requirements set forth in this tariff for the installation, operation, and maintenance of customer-provided facilities and equipment that is connected to Company-owned equipment.

Januarda 0000	Effectives	2000
Issued: , 2009	Effective:	, 2009

Issued By:

## 2.4 Liability of the Company

## 2.4.1 Limitations on Liability

- (A) Except as otherwise stated in this section, the liability of the Company for damages arising out of either: (1) the furnishing of its services, including but not limited to mistakes, omissions, interruptions, delays, or errors, or other defects, representations, or use of these services, or (2) the failure to furnish its service, whether caused by acts or omission, shall be limited to the extension of allowances to the Customer for interruptions in service as set forth in Section 2.9.
- (B) Except for the extension of allowances to the Customer for interruptions in service as set forth in Section 2.9, the Company shall not be liable to a Customer or third party for any direct, indirect, special, incidental, reliance, consequential, exemplary or punitive damages, including, but not limited to, loss of revenue or profits, for any reason whatsoever, including, but not limited to, any act or omission, failure to perform, delay, interruption, failure to provide any service or any failure in or breakdown of facilities associated with the service.
- (C) The liability of the Company for errors in billing that result in overpayment by the Customer shall be limited to a credit equal to the dollar amount erroneously billed or, in the event that payment has been made and service has been discontinued, to a refund of the amount erroneously billed.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 2.4 Liability of the Company (Cont'd)
  - 2.4.1 Limitations on Liability (Cont'd)
    - (D) The Company shall not be liable for any claims for loss or damages involving:
      - 1. Any act or omission of: (a) the Customer, (b) any other entity furnishing service, equipment or facilities for use in conjunction with services or facilities provided by the Company; or (c) common carriers or warehousemen;
      - 2. Any delay or failure of performance or equipment due to causes beyond the Company's control, including but not limited to, acts of God, fires, floods, earthquakes, hurricanes, or other catastrophes; national emergencies, insurrections, riots, wars or other civil commotions; strikes, lockouts, work stoppages or other labor difficulties; criminal actions taken against the Company; unavailability, failure or malfunction of equipment or facilities provided by the Customer or third parties; and any law, order, regulation or other action of any governing authority or agency thereof;
      - 3. Any unlawful or unauthorized use of the Company's facilities and services;
      - Libel, slander, invasion of privacy or infringement of patents, trade secrets, or copyrights arising from or in connection with the transmission of communications by means of Company-provided facilities or services; or by means of the combination of Company-provided facilities or services with Customer-provided facilities or services;

Issued:, 2009	Effective:, 2009
Issued By:	Iulia Rarghouthi

- 2.4 Liability of the Company (Cont'd)
  - 2.4.1 Limitations on Liability (Cont'd)
    - 5. Breach in the privacy or security of communications transmitted over the Company's facilities;
    - 6. Changes in any of the facilities, operations or procedures of the Company that render any equipment, facilities or services provided by the Customer obsolete, or require modification or alteration of such equipment, facilities or services, or otherwise affect their use or performance, except where reasonable notice is required by the Company and is not provided to the Customer, in which event the Company's liability is limited as set forth in Paragraph A.
    - 7 Defacement of or damage to Customer premises resulting from the furnishing of services or equipment on such premises or the installation or removal thereof;
    - 8. Injury to property or injury or death to persons, including claims for payments made under Workers' Compensation law or under any plan for employee disability or death benefits, arising out of, or caused by, any act or omission of the Customer, or the construction, installation, maintenance, presence, use or removal of the Customer's facilities or equipment connected, or to be connected to the Company's facilities;

Issued:	. 2009	Fffective:	. 2009

Issued By:

- 2.4 Liability of the Company (Cont'd)
  - 2.4.1 Limitations on Liability (Cont'd)
    - Any intentional, wrongful act of a Company employee when such act is not within the scope of the employee's responsibilities for the Company and/or is not authorized by the Company;
    - 10. Any representations made by Company employees that do not comport, or that are inconsistent, with the provisions of this Tariff;
    - 11. Any act or omission in connection with the provision of 911, E911, or similar services;
    - 12. Any noncompletion of calls due to network busy conditions;
    - 13. Any calls not actually attempted to be completed during any period that service is unavailable.

Issued:	2009	Effective:	, 2009

Issued By:

- 2.4 Liability of the Company (Cont'd)
  - 2.4.1 Limitations on Liability (Cont'd)
    - (E) The Company shall be indemnified, defended and held harmless by the Customer or end user from and against any and all claims, loss, demands, suits, expense, or other action or any liability whatsoever, including attorney fees, whether suffered, made, instituted, or asserted by the Customer or by any other party, for any personal injury to or death of any person or persons, and for any loss, damage or destruction of any property, including environmental contamination, whether owned by the Customer or by any other party, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, presence, condition, location, use or removal of any Company or Customer equipment or facilities or service provided by the Company.
    - (F) The Company does not guarantee nor make any warranty with respect to installations provided by it for use in an explosive atmosphere. The Company shall be indemnified, defended and held harmless by the Customer from and against any and all claims, loss, demands, suits, or other action, or any liability whatsoever, including attorney fees, whether suffered, made, instituted or asserted by the Customer or by any other party, for any personal injury to or death of any person or persons, and for any loss, damage or destruction of any property, including environmental contamination, whether owned by the Customer or by any other party, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, presence, condition, location, use or removal of any equipment or facilities or the service.

Issued: ,2009	Effective:	, 2009

Issued By:

- 2.4 Liability of the Company (Cont'd)
  - 2.4.1 Limitations on Liability (Cont'd)
    - (G) The Company assumes no responsibility for the availability or performance of any cable or satellite systems or related facilities under the control of other entities, or for other facilities provided by other entities used for service to the Customer, even if the Company has acted as the Customer's agent in arranging for such facilities or services. Such facilities are provided subject to such degree of protection or nonpreemptibility as may be provided by the other entities.

Issued:	, 2009	Effective:	2009

Issued By:

## 2.5 Application for Service

#### 2.5.1 Minimum Contract Period

- 2.5.1.1 Except as otherwise provided, the minimum contract period is one month for all services furnished. However, if a new customer notifies the Company within twenty days after receipt of the first bill that certain services or equipment are not desired, the Company will delete such services or equipment from the customer's account without a record keeping or service ordering charge. The customer nonetheless shall be responsible for all monthly usage and installation charges incurred for the use of such service and equipment.
- 2.5.1.2 Except as provided in 2.5.2.1, the length of minimum contract period for directory listings, and for joint user service where the listing actually appears in the directory, is the directory period. The directory period is from the day on which the directory is first distributed to customers to the day the succeeding directory is first distributed to customers.
- 2.5.1.3 The Company may require a minimum contract period longer than one month in connection with special, non-standard types or arrangements of equipment necessary to meet special demands for service.

#### 2.5.2 Cancellation of Service

2.5.2.1 Where the applicant cancels an order for service prior to the start of the installation no charge shall apply, except to the extent the Company incurs a service order or similar charge from a supplying carrier.

Issued:	, 2009	Effective:	2009

Issued By:

- 2.5 Application for Service (Cont'd)
  - 2.5.2 Cancellation of Service
    - 2.5.2.1 Where the applicant cancels an order for service prior to the start of the installation no charge shall apply, except to the extent the Company incurs a service order or similar charge from a supplying carrier.
    - 2.5.2.2 The Company reserves the right to redefine its regions, add new regions, or remove regions from its current offering, as it deems appropriate in its sole discretion and will provide the Customer with at least thirty (30) days' notice of any change in the definition of the Company's regions.
    - 2.5.2.3 In the event that the Company plans to exit a current region, the Company will make application and request approval from the Commission and provide notice of not less than fifty (50) days notice prior to the Intended Date of Discontinuance to the Commission and the Division, as required by R746-350-3(A) and by R746-350-4(B) of the Commission's rules. The Customer shall be allowed to immediately terminate services in the affected region without penalty.

Issued:, 2009	Effective:	, 2009

Issued By:

# 2.6 Payment for Service

- 2.6.1 The Company will bill Customer monthly, with recurring charges being billed in advance and any usage charges billed in arrears. Payment is due upon receipt by Customer and payable within thirty (30) days of the Bill Date (the "Due Date"). The customer is responsible for payment of all charges for service furnished to the customer. Charges based on actual usage during a month will be billed monthly in the month following the month in which the service was used. All fixed monthly and nonrecurring charges for services ordered will be billed monthly in advance.
- 2.6.2 Payments are past due if not received by the Company by the Due Date. A late payment charge of 1.5% will be assessed on all unpaid balances more than thirty (30) days old. Late payment charges shall not apply if payment is made before the next bill is rendered by the Company. Bills not paid within thirty (30) days of the Bill Date and which have not been disputed in accordance with the procedures set forth in Section 2.8.1 of this tariff, may result in suspension of service until the overdue payments and any additional charges that may be imposed to restore service have been paid. Customer agrees to pay all costs incurred by the Company in collecting any unpaid amounts, including attorneys' fees.
- 2.6.3 The Company reserves the right to require from an applicant for service advance payments of fixed charges and nonrecurring charges. The advance payment will not exceed an amount equal to the nonrecurring charge(s) and one month's charges for the service. The advance payment will be applied to any indebtedness for the service for which the advance payment is made on the customer's initial bill.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 2.7 Customer Deposits

- 2.7.1 The Company may require a deposit or guarantee of payment from any customer or applicant who has not established good credit with the Company. Deposit or guarantee of payment requirements as prescribed by the Company will be based upon standards which bear a reasonable relationship to the assurance of payment. A deposit shall not exceed an estimated two months' gross bill or existing two months' bill where applicable. All deposits shall be in addition to payment of an outstanding bill or a part of such bill as has been resolved to the satisfaction of the Company, except where such bill has been discharged in bankruptcy. The Company may determine whether a customer has established good credit with the Company, except as herein restricted:
  - 2.7.1.1 A customer, who within the last 12 months has not had service disconnected for nonpayment of a bill and has not been liable for disconnection of service for nonpayment of a bill, and the bill is not in dispute, shall be deemed to have established good credit.
  - 2.7.1.2 The Company shall not require a deposit or a guarantee of payment based upon income, home ownership, residential location, employment tenure, nature of occupation, race, color, creed, sex, marital status, age, national origin, or any other criteria which does not bear a reasonable relationship to the assurance of payment or which is not authorized by this section.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 2.7 Customer Deposits (Cont'd)

#### 2.7.2 Return of Deposit

When a deposit is to be returned, the Customer may request that the full amount of the deposit be issued by check. If the Customer requests that the full amount be credited to amounts owed the Company, the Company will apply the deposit to any amount currently owed to the Company, and return any remaining amount of the deposit to the Customer by check.

#### 2.7.3 Interest on Deposits

Interest shall be paid on deposits at a rate prescribed and specified by the Commission and shall be payable from the date of deposit to the date of refund or disconnection.

Upon termination of service, the deposit with accrued interest shall be credited to the final bill and the balance shall be returned to the Customer as soon as practicable.

Issued:	, 2009	Effective:	<u>,</u> 2009

Issued By:

- 2.8 Customer Complaints and Billing Disputes
  - 2.8.1 In the event that Customer disputes any charges, Customer must submit an oral or written claim describing the disputed amount. Customer shall submit all documentation as may reasonably be required to support the claim. Payment may be withheld for the amounts subject to a dispute submitted prior to the Due Date. All disputes and claims for refunds must be submitted to the Company within ninety (90) days of the Bill Date. If Customer does not submit a claim as stated above, Customer waives all rights to file a claim thereafter. The Company shall investigate and resolve all disputes within fifteen (15) days of receipt of the dispute and the Company's resolution of such a dispute is final. Any portion of a disputed amount deemed payable by the Company must be paid in full within ten (10) days of resolution or Customer's service may be subject to disconnection and late payment charges imposed on the overdue amount.
  - 2.8.2 Any unresolved dispute may be directed in writing to the Division of Public Utilities at PO Box 146751, Salt Lake City, Utah 84114-6751 or by phone at (801) 530-7622 or (800) 874-0904.

Upon receipt of a notice from the Division of the filing of an information complaint, the Company shall investigate and attempt to resolve the complaint within five (5) working days.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 2.9. Allowances for Interruptions in Service

In the event that a Customer's service is interrupted, other than by the negligence or the willful act of the Customer, and it remains out of service for a specified number of hours, after being reported or found by the Company to be out of order, credit adjustments shall be made to the Customers account

#### 2.9.1 General

- A service is interrupted when it becomes unusable to the Customer, e.g., the Customer is unable to transmit or receive, because of a failure of a component furnished by the Company under this tariff.
- <u>B.</u> An interruption period begins when the Customer reports a service, facility or circuit to be inoperative and releases it for testing and repair. An interruption period ends when the service, facility or circuit is operative.
- C. If the Customer reports a service, facility or circuit to be inoperative but declines to release it for testing and repair, the service, facility or circuit is considered to be impaired but not interrupted. No credit allowances will be made for a service, facility or circuit considered by the Company to be impaired.

Issued: , 2009	Effective:	, 2009
Issued By:		

#### 2.9. Allowances for Interruptions in Service (Cont'd)

#### 2.9.2 Limitations on Allowances

No credit allowance will be made for any interruption of service:

- A. due to the negligence of, or noncompliance with the provisions of this Tariff by, any person or entity other than the Company, including but not limited to the Customer or other common carriers connected to the service of the Company;
- B. due to the failure of power, equipment, systems, or services not provided by the Company;
- C. due to circumstances or causes beyond the control of the Company;
- during any period in which the Company is not given full and free access to its facilities and equipment for the purposes of investigating and correcting interruptions;
- E. during any period in which the Customer continues to use the service on an impaired basis;
- F. during any period when the Customer has released service to the Company for maintenance purposes or for implementation of a Customer order for a change in service arrangements;
- G. that occurs or continues due to the Customer's failure to authorize replacement of any element of special construction; and
- H. that was not reported to the Company within thirty (30) days of the date that service was affected.

Issued: , 2009	Effective:	
Issued By:		

- 2.9. Allowances for Interruptions in Service (Cont'd)
  - 2.9.3 Application of Credits for Interruptions of Service
    - A. Credits for interruptions of service, for which charges are specified on the basis of per minute of use, or on the usage of a fraction of a minute, shall in no event exceed an amount equal to the initial period charge provided for under this tariff.

The Company shall allow for interruptions in exchange telephone service of 24 hours or more not due to conduct of Customer, an amount equal to the fixed monthly charges for exchange service multiplied by the ratio of the days of interruption to thirty days When interruptions continue beyond 24 hours, credit allowance will be given in successive 24-hour multiples.

Issued:, 2009	Effective:	, 2009

Issued By:

#### 2.10 Taxes and Fees

- 2.10.1 All state and local taxes and fees shall be listed as separate line items on the customer's bill.
- 2.10.2 If a municipality, other political subdivision or local agency of government, or the Commission imposes and collects from the Company a gross receipts tax, occupation tax, license tax, permit fee, franchise fee, or regulatory fee, such taxes and fees shall, as allowed by law, be billed pro rata to the customer receiving service from the Company within the territorial limits of such municipality, other political subdivision or local agency of government.
- 2.10.3 Service shall not be subject to taxes for a given taxing jurisdiction if the customer provides the Company with written verification, acceptable to the Company and to the relevant taxing jurisdiction, that the customer has been granted a tax exemption.

# 2.11 Returned Check Charge

A return check charge of \$25.00 will be assessed for checks returned for insufficient funds.

#### 2.12 Restoration of Service

A reconnection fee of \$25.00 per occurrence is charged when service is reestablished for customers who had been disconnected for non-payment.

lssued:	. 2009	Effective:	2009

Issued By:

2.13	Special	Customer	Arrangement	ts

In cases where a customer requests special or unique arrangements which may include but are not limited to engineering, conditioning, installation, assembly, purchase or other special services not offered under this tariff, the Company, may provide the requested services. Appropriate recurring charges and/or nonrecurring charges and other terms and conditions will be developed for the customer for the provisioning of such arrangements.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 2.14 Disconnection and Termination of Service

The Company shall not disconnect basic local service for nonpayment of toll or information service charges or any service other than basic local service.

#### 2.14.1 Disconnection of Service Without Notice

The Company shall have the right to refuse or discontinue telephone service or service arrangements without advance notice, if the acts of the Customer or the conditions upon their premises are such as to indicate a clear emergency or serious health or safety hazard exists, or when there is unauthorized use of or diversion of the Company's service or tampering with lines, or other property owned by the Company

The Company will attempt to contact the Customer by telephone prior to discontinuing the Service or portions thereof. If the Company is unable to contact the Customer by telephone, a letter will be mailed to the Customer on the same date that their service or service arrangement is discontinued, explaining the reasons for such action and the Customer's right to dispute such action.

Customer is responsible for all charges attributable to Customer, even if incurred as a result of fraudulent or unauthorized use of the Service by third parties. The Company may, but is not obligated to, detect or report unauthorized or fraudulent use of Service.

Issued:, 2009	Effective:	, 2009

Issued By:

- 2.14 Disconnection and Termination of Service (Cont'd)
  - 2.14.2 Disconnection of Service Requiring Notice
    - 2.14.2.1 The Company may disconnect service for any of the following reasons provided it has notified the customer of its intent, in writing, to disconnect service and has allowed the customer a reasonable time of not less than seven (7) days in which to remove the cause for disconnection:
      - 2.14.2.1.A Non-compliance with Regulations. For violation of or non-compliance with Commission's rules and regulations or for violation of or non-compliance with the Company's tariffs on file with the Commission, using the service for unlawful purposes.
      - 2.14.2.1.B Failure on Contractual Obligations. For failure of the customer to fulfill his contractual obligations for service subject to regulation by the Commission.
      - 2.14.2.1.0 Refusal of Access. For failure of the customer to permit the Company to have reasonable access to its equipment and property.
      - 2.14.2.1.D Failure to meet the Company's deposit and credit requirements.
      - 2.14.2.1.E For non-payment of a bill for regulated service, with an account balance of more then \$15.00 provided that the Company has made a reasonable attempt to effect collection and has given the customer written notice of its intent to deny service if settlement of account is not made and provided the customer has at least seven (7) days notice.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 2.14 Disconnection and Termination of Service (Cont'd)
  - 2.14.2. Disconnection of Service Requiring Notice (Cont'd)
    - 2.14.2.1.F Failure to Comply with Service Conditions. For failure of the customer to furnish the service equipment, permits, certificates, or rights-of-way, specified by the Company as a condition to obtaining service, or if the equipment or permissions are withdrawn or terminated; tampering with or destroying Company lines, equipment and other properties.
    - 2.14.2.1.G Failure to Comply with Municipal Ordinances. For failure to comply with municipal ordinances or other laws pertaining to telephone service.
    - 2.14.2.1.H. Abusive use of the telephone services in a manner that interferes with the service of another person;
    - 2.14.2.1.1 Intentionally using the service in a manner that causes wrongful billing charges to another person;
    - 2.14.2.1.J Intentionally using the service to transmit messages or to locate a person to give or obtain information, without payment of appropriate message charges;
    - 2.14.2.1K. Using the service with fraudulent intent by impersonating someone else.
    - 2.14.2.1.L Deliberately furnishing false information when applying for and obtaining telephone services.
    - 2.14.2.1.M Abandonment of service.

Issued:	, 2009	Effective:	. 2009

Issued By:

- 2.15 Unlawful Use of Service
  - 2.15.1 Service shall not be used for any purpose in violation of law or for any use as to which the customer has not obtained all required governmental approvals, authorizations, licenses, consents, and permits. The Company shall refuse to furnish service to an applicant or shall disconnect the service without notice of a customer when:
    - 2.15.1.1 An order shall be issued, signed by a judge finding that probable cause exists to believe that the use made or to be made of the service is prohibited by law, or
    - 2.15.1.2 The Company is notified in writing by a law enforcement agency acting within its jurisdiction is being used or will be used for the purpose of transmitting or receiving gambling information in interstate or foreign commerce in violation of the law.
  - 2.15.2 If service has been physically disconnected by law enforcement officials at the customer's premises and if there is not presented to the Company the written finding of a judge, then upon written or verbal request of the subscriber, and agreement to pay restoral of service charges and other applicable service charges, the Company shall promptly restore such service.

Issued:	, 2009	Effective: _	, 2009

Issued By:

Service shall not be used in any manner that interferes with other persons in the use of their service, prevents other persons from using their service, or otherwise impairs the quality of service to other customers. The Company may require a customer to immediately shut down its transmission of signals if said transmission is causing interference to others or impairing the service of others.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 2.17 Incomplete Calls

There shall be no charge for incomplete calls. No charge will be levied for unanswered calls. Customers will receive credit for calls placed to a wrong number if the customer notifies the Company of the error.

#### 2.18 Overcharge

When a customer has been overcharged by the Company, the amount shall be refunded or credited to the customer. When a customer has been underbilled and the amount owed by the Customer exceeds \$25.00 and the period over which the underbilling accumulated exceeds one month, the Company shall offer reasonable payment arrangements.

Issued:	_, 2009	Effective:	, 2009
	<b>-</b> '		

Issued By:

#### 2.19 Notices

Any notice required or permitted to be given under this tariff shall be in writing and delivered by hand, mail, national overnight courier service or by fax if confirmed by telephone to the customer, at the address or phone numbers shown herein or at such other address or phone numbers as shall be designated from time to time.

#### 2.20 Minimum Call Completion Rate

A customer can expect a call completion rate (number of calls completed / number of calls attempted) of not less than 98% during peak use periods for all Feature Group D services ("1+" dialing).

Issued:	, 2009	Effective:	, 2009

Issued By:

# 2.21 Emergency Calling

Access to 911 and E911 service is offered solely as an aid in handling assistance calls in connection with fire, police, medical, and other emergencies. The Company is not responsible, in the absence of gross negligence or willful misconduct, for any losses, claims, demands, suits, or any liability, whether suffered, made, instituted, or asserted by the Customer or by any other party or person, for any personal injury to or death of any person or persons, and for any loss, damage, or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused by the installation, operation, failure to operate, maintenance, removal, presence, condition, location, or use of such facilities. By dialing 911, the Customer agrees to release, indemnify, defend, and hold harmless the Company from any and all loss or claims, whatsoever, whether suffered, made, instituted, or asserted by the destruction of any property, whether owned by the Customer or others. Not withstanding any provision to the contrary, in no event shall the Company be liable for any special, incidental, consequential, exemplary, or punitive damages of any nature whatsoever.

The Company is not responsible for any infringement or invasion of the right of privacy of any person or persons, caused, or claimed to have been caused, directly or indirectly, by the installation, operation, failure to operate, maintenance, removal, presence, condition, occasion or use of the 911 service features and the equipment associated therewith, including, but not limited to, the identification of the telephone number, address or name associated with the telephone used by the party or parties accessing the 911 service.

Issued:	, 2009	Effective:	. 2009

Issued By:

#### **SECTION 3. DESCRIPTION OF SERVICES**

#### 3.1 Promotional Offerings

The Company may, from time to time, engage in special Promotional Offerings limited to certain dates, times, or locations designed to attract new customers or to increase customers awareness of a particular tariff offering. These promotions will be approved by the Commission with specific starting and ending dates, and be made part of this tariff.

#### 3.2 Individual Case Basis ("ICB") Offerings

Arrangements may be developed on a case-by-case basis for tariffed service or services to a specific customer at rates, terms or conditions provided through an agreement instead of pursuant to tariff. The Company may or may not have an equivalent service in its the tariff on file with the Commission, and the quoted ICB rates may be different than the tariffed rates.

The Company will maintain records of its ICB contracts for Commission review as conditions or circumstances may require.

Issued:, 2009	Effective:	, 2009

Julie Barghouthi
Sr. Vice President Product Development & Access Management
IntelePeer, Inc.
2855 Campus Drive, Suite 200
San Mateo, CA 94403

Issued By:

#### 3.3 State Universal Service Charges

The Company will contribute to the Utah universal service fund, pursuant to Commission orders, regulations, and Utah Code Annotated, § 54-8b-15-501 The Commission will establish the method by which the contributions shall be calculated, collected and distributed, and additional monthly charges to customers.

#### 3.4 Telephone Relay Service ("711")

The Company will provide Telephone Relay Service ("TRS") to speech and/or hearing-impaired customers. The service permits telephone communications between hearing and/or speech impaired individuals who must use a Telecommunications Device for the Deaf (TDD) or a Teletypewriter (TTY) and individuals with normal hearing and speech. The service enables callers to type a message into a telephone set equipped with a keypad and message screen and to have a live operator read the message to a recipient and to type the message recipient's response to the speech or hearing-impaired caller. The service will be available at no additional charge, for use by customers who are speech or hearing-impaired.

Monthly Charge: \$0.10 per line

Issued: , 2009 Effective: \_\_\_\_\_\_, 2009

Issued By:

3.5	Long	Distance	Service

Long Distance Service is a voice service offered to business customers. The service permits direct dialed outbound calling at a single per minute rate. Service is provided from presubscribed, dedicated or shared use access lines. No monthly recurring charges or minimum monthly billing requirements apply.

ssued:	, 2009	Effective:	, 2009

Issued By:

3.6 Local Exchange Service

Local Exchange Service is telephone service that entitles the customer to originate local calls, without toll charges, to all local exchange access lines connected to a Central Office (CO) of the exchange, or to all exchange access lines served by COs of the extended local service area where comprised of more than one exchange. Service will be provided where facilities are available from the Local Exchange Company (LEC). Service is classified as business service and business rates apply when any of the following conditions exist:

When the service is furnished at a location where a business, trade or practice is performed and where the use of the location is not confined primarily to domestic activities.

Service for social clubs will be considered business service.

When the directory listing is to be a business listing, except when a residence telephone number is advertised as an alternate call number in connection with a business telephone number.

Issued:	, 2009	Effective:	, 2009

Issued By:

#### 3.7 Directory Assistance

Directory Assistance (DA) is defined as furnishing aid in obtaining telephone numbers. The Directory Assistance operator will not transfer, forward or redial a customer's call to any other location for any purpose other than the provision of DA service. The service is furnished subject to the condition that there will be no abuse or fraudulent use of the service. Abuse or fraudulent use of the service includes the obtaining, or attempting to obtain, or assisting another to obtain or to attempt to obtain Directory Assistance service, by rearranging, tampering with, or making connection with any facilities of the Company, or by any trick, scheme, false representation, or false credit device, or by or through any other fraudulent means or device whatsoever, with attempt to avoid payment, in whole or in part, of the regular charge for such service. In addition to any other action authorized by this Tariff, the Company may, in such cases of abuse or fraudulent use, assess appropriate Directory Assistance charges on the Customer's regular telephone account.

## 3.8 Custom Calling Services

Custom Calling Service is an optional service arrangement of central office services furnished to business customers. Custom Calling Features are offered subject to availability and suitably equipped central office facilities.

_			
	2000	Effective	2000
issuea:	7009	LIIECUVE.	7009

Issued By:

#### **SECTION 4. RATES AND CHARGES**

- 4.1 Calculation of Rates
  - 4.1.1 The customers usage charge is based on the actual usage of the Company's network. The chargeable time for a call is determined by the duration of the call. Chargeable time begins when connection is established between the calling station and the called station. Chargeable time ends when the calling station hangs up. If the called station hangs up, but the calling station does not, chargeable time ends when the connection is released by either automatic timing equipment in the telecommunications network or by an operator.
  - 4.1.2 Calls are billed in (six) 6 second increments with an eighteen (18) second minimum for interLATA calls and a twenty-four (24) second minimum on intraLATA calls. Billing will be rounded up to the nearest penny for each call.

4.1.3 Different rates based on the time of day or day of week are described in the following rate table.

escribed in the following rate table.					
		To, but not	Days		
Rate Periods	From	Including			
Weekdays	8:00 a.m.	5:00 p.m.	Monday-Friday		
Evenings	5:00 p.m.	11:00 p.m.	Monday-Friday		
	5:00 p.m.	11:00 p.m.	Sunday		
Night/Weekends	11:00 p.m.	8:00 a.m.	Monday-Sunday		
	8:00 a.m.	5:00 p.m.	Saturday-Sunday		
	5:00 p.m.	11:00 p.m.	Saturday		

The Company charges weekend rates on the following Federal holidays: New Year's Day, President's Day, Independence Day, Labor Day, Thanksgiving Day, and Christmas Day.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 4.2 Local Exchange Service
  - 4.2.1 Business Service
    - (A) Service Charges

Non-Recurring

<u>Charge</u>

Service Order Charge

Per Line or Trunk \$50.00

Changes

To change class, type or grade

Of Service (Per Line or Trunk) \$32.00

Moves \$42.00

Issued: , 2009 Effective: , 2009

Issued By:

- 4.2 Local Exchange Service (Cont'd)
  - 4.2.1 Business Service (Cont'd)
    - (B) Monthly Recurring Charges

Rate

(1) Flat Rate Service

Per Main and Each
Additional Line

\$26.00

(2) Message Rate Service

Per Main and Each Additional Line

\$20.00

Each Additional Unit

\$0.08

Issued: , 2009 Effective: \_\_\_\_\_\_, 2009

Issued By:

4.3 Directory Assistance Service

Customer Dialed \$1.99 Per Call

4.4 Custom Calling Services

Monthly Recurring Charge

Per Line

Call Waiting \$8.00
Call Forwarding \$6.00
Three Way Calling \$6.00

Issued By:

	4.5	Telephone	Relay	Service	("711")
--	-----	-----------	-------	---------	---------

Monthly Charge: \$0.10 per line

4.6 State Universal Service Charges

Monthly Charge: \$.0067 per line

Issued: , 2009 Effective: \_\_\_\_\_, 2009

Issued By:

#### 5.1 General

IntraLATA toll presubscription is a procedure whereby an end user may select and designate an IntraLATA Toll Provider ("ITP") to access IntraLATA toll calls without dialing an access code. The end user may designate an ITP for IntraLATA toll, a different carrier for InterLATA toll, or the same carrier for both. This ITP is referred to as the end user preferred IntraLATA toll provider.

Each carrier will have one or more access codes assigned to it for various types of service. When an end user selects a carrier as its preferred IntraLATA toll provider, only one access code of that carrier may be incorporated into the switching system of the Company permitting access to that carrier by the end user without dialing an access code. Should the same end user wish to use other services of the same carrier, it will be necessary for the end user to dial the necessary access code(s) to reach that carrier's other service(s).

An ITP must use Feature Group D ("FGD") Switched Access Service to qualify as an IntraLATA toll provider. All ITPs must submit a Letter of Intent ("LOD to the Company at least twenty days prior to the IntraLATA toll-presubscription-conversion date or, if later, forty-five days prior to the date on which the carrier proposed to begin participating in IntraLATA toll presubscription.

Selection of an ITP by an end user is subject to the terms and conditions in Section 5.2.

Issued:	. 2009	Effective:	. 2009	

Issued By:

#### 5.2 Presubscription Charge Application

5.2.1 Initial Free Presubscription Choice for New Users

New end users (including an existing customer who orders an additional line) who subscribe to service will be asked to select a primary ITP when they place an order for Company Exchange Service. If a customer cannot decide upon an IntraLATA toll carrier at the time, the customer will have thirty days following completion of the service request to make an IntraLATA PIC choice without charge. In the interim, the customer will be assigned a "No-PIC" and will have to dial an access code to make IntraLATA toll calls. The free selection period available to new end users is the period within thirty days of installation of the new service.

Initial free selections available to new end user are:

- 1. Designating an ITP as their primary carrier, thereby requiring no access code to access that ITP's service. Other carriers are accessed by dialing 101-XXXX or other required codes.
- 2. Choosing no carrier as a primary carrier, thus requiring 101-XXXX code dialing to access all ITPs. This choice can be made by directly contacting the Company.

Issued: , 2009	Effective:	, 2009
Issued By:		

## 5.2 Presubscription Charge Application (Cont'd)

5.2.2 Cancellation of IntraLATA Toll Presubscription by an ITP

If an ITP elects to discontinue Feature Group D service after implementation of the IntraLATA toll presubscription option, the ITP is obligated to contact, in writing, all end users who have selected the canceling ITP as their preferred IntraLATA toll provider. The ITP must inform the end users that it is canceling its Feature Group D service, request that the end user select a new ITP and state that the canceling ITP will pay the PIC change charge. The ITP must provide written notification to the Company that this activity has taken place.

Issued:	, 2009	Effective:	, 2009

Issued By:

- 5.3 End User/Pay Telephone Service Provider Charge Discrepancy ("Anti-Slamming Measure")
  - 5.3.1 When a discrepancy is determined regarding an end user's designation of a preferred IntraLATA toll carrier, the following applies depending upon the situation described:

A signed letter of authorization takes precedence over any order other than subsequent, direct customer contact with the Company.

When two or more orders are received for an end user line generated by telemarketing, the date field on the mechanized record used to transmit PIC change information will be used as the PIC authorization date. The order with the latest application date determines customer choice.

5.3.2 Verification of Orders for Telemarketing

No ITP shall submit to the Company a PIC change order generated by outbound telemarketing unless, and until, the order has first been conformed in accordance with the following procedures:

- 5.3.2.1 The ITP has obtained the customer's written authorization to submit the order that explains what occurs when a PIC is changed and confirms:
  - 5.3.2.1.A The customer's billing name and address and each telephone number to be covered by the PIC change order;
  - 5.3.2.1.B The decision to change the PIC to the ITP; and
  - 5.3.2.1.0 The customer's understanding of the PIC change fee; or

Issued:	2000	Effective:	, 2009
155UEU.	, 2009	Lifective.	, 2003
	,		

Issued By:

# 5.3 End User/Pay Telephone Service Provider Charge Discrepancy ("Anti-Slamming Measure") (Cont'd)

- 5.3.2 Verification of Orders for Telemarketing (Cont'd)
  - 5.3.2.2 The ITP has obtained the customer's electronic authorization, placed from the telephone number(s) on which the PIC is to be changed, to submit the order that confirms the information described in 5.3.2.1 preceding to confirm the authorization; or
  - 5.3.2.3 An appropriately qualified and independent third party operating in a location physically separate from the outbound telemarketing representative has obtained the customer's oral authorization to submit the PIC change order that confirms and includes appropriate verification date (e.g., the customer's date of birth or social security number).
- 5.3.3 The Company will follow the Federal Communications Commission's and the Commission's regulations regarding slamming. The Company will not impose a penalty or charge for unauthorized IntraLATA toll provider changes.
  - 5.3.4 The customer owns the exclusive right to select the PIC of their choice, and may choose to migrate from one carrier to another at any time. There is no reason a carrier may refuse to release a customer who has stated their intent to select a different carrier.

Issued:	, 2009	Effective:	, 2009

Issued By:

5.4	Rates	and	Cha	rges
-----	-------	-----	-----	------

Non-recurring Charge

Authorized PIC/IPIC Change Charge Per Line, Per Occurrence

\$5.00

Issued: , 2009 Effective: \_\_\_\_\_, 2009

Issued By:

## **EXHIBIT H**

Proposed Intrastate Access Tariff

#### **Tariff Schedule Applicable to**

#### **Intrastate Switched Access Services**

#### Furnished by

#### IntelePeer, Inc.

#### **Between Points Within the State of Utah**

This tariff contains the descriptions, regulations, and rates applicable to the furnishing of service for telecommunications services provided by IntelePeer, Inc. with principal offices at 2855 Campus Drive, Suite 200, San Mateo, CA 94403. This tariff applies for services furnished within the State of Utah. This tariff is on file with the Utah Public Service Commission, and copies may be inspected, during normal business hours, at the Company's principal place of business.

Issued:	, 2009	Effective:	, 2009

#### **CHECK SHEET**

The sheets of this tariff inclusive herein are effective as of the date shown at the bottom of the respective sheet(s). Original and revised sheets as named below comprise all changes from the original tariff and are currently in effect as of the date on the bottom of this sheet.

Page	Number or Revision	<u>Page</u>	Number of Revision
1	Original	30	Original
2	Original	31	Original
3	Original	32	Original
4	Original	33	Original
5	Original	34	Original
6	Original	35	Original
7	Original	36	Original
8	Original	37	Original
9	Original	38	Original
10	Original	39	Original
11	Original	40	Original
12	Original	41	Original
13	Original	42	Original
14	Original	43	Original
15	Original	44	Original
16	Original	45	Original
17	Original	46	Original
18	Original	47	Original
19	Original	48	Original
20	Original	49	Original
21	Original	50	Original
22	Original	51	Original
23	Original	52	Original
24	Original	53	Original
25	Original	54	Original
26	Original	55	Original
27	Original	56	Original
28	Original	57	Original
29	Original	58	Original

Issued: , 2009 Effective Date: , 2009

# CHECK SHEET (Cont'd.)

<u>Page</u>	Number or Revision	<u>Page</u>	Number of Revision
<u>59</u>	Original	88	Original
<u>60</u>	Original	<u>89</u>	Original
61	Original	<u>90</u>	Original
62	Original	91	Original
<u>63</u>	Original	92	Original
<u>64</u>	Original	<u>93</u>	Original
<u>65</u>	Original	<u>94</u>	Original
<u>66</u>	Original	<u>95</u>	Original
<u>67</u>	Original	<u>96</u>	Original
<u>68</u>	Original	97	Original
<u>69</u>	Original	<u>98</u>	Original
70	Original	<u>99</u>	Original
71	Original	100	Original
72	Original	101	Original
73	Original	102	Original
74	Original	103	Original
75	Original	104	Original
76	Original	105	Original
77	Original	106	Original
78	Original	107	Original
<u>79</u>	Original	108	Original
<u>80</u>	Original	109	Original
81	Original	110	Original
82	Original	111	Original
<u>83</u>	Original	112	Original
<u>84</u>	Original	113	Original
<u>85</u>	Original	114	Original
<u>86</u>	Original	115	Original
87	Original	116	Original
		117	Original
		118	Original

Issued:	, 2009	Effective Date:	, 2009
ibbucu.	. 2007	Effective Bate.	, 2007

## TABLE OF CONTENTS

## TABLE OF CONTENTS

CHECK SHI	EET	1
TABLE OF C	CONTENTS	3
TARIFF FOI	RMAT	5
EXDLANIAT	CION OF CVMDOLC AND DEFEDENCE	
	TION OF SYMBOLS AND REFERENCE ED IN THIS TARIFF	6
DEFINITIO	NS	7
1.0 APPLI	CATION OF TARIFF	16
2.0 REGUI	LATIONS	17
21	Undertaking of the Company	
2.2	Prohibited Uses	
23	Obligations of the Customer	24
2.4	Customer Equipment and Channels	29
25	Customer Deposits and Advance Payments	30
2.6	Payment Arrangements	
2.7	Allowances for Interruptions in Service	
28	Application of Rates	48
3.0 SERVIC	CE AND RATE DESCRIPTIONS	53
31	Access Services	53
32	Miscellaneous Services	85

Issued: \_\_\_\_\_\_, 2009 Effective Date: \_\_\_\_\_, 2009

## TABLE OF CONTENTS (Cont'd.)

			<u>Page</u>
4.0	DILI	INC. AND COLLECTION SERVICES	89
		LING AND COLLECTION SERVICES	89
	4.1	General	89
	4.2	Recording Service	96
	43	Automatic Number Identification	96
	4.4	Billing Name and Address Service	
5.0	RAT	ES	101
	5.1	Access Service	101
	5.2	Reserved for Future Use	111
	53	Miscellaneous Services	112
	5.4	Billing and Collection Services	113
	5.5	Transit Traffic Service	113
	5.6	LNP Query Service	113
	5.7	Pay Phone Dial Around Recovery	113

Issued: , 2009 Effective Date: , 2009

#### TARIFF FORMAT

- A. Page Numbering Page numbers appear in the upper right corner of the page. Pages are numbered sequentially. However, new pages are occasionally added to the tariff. When a new page is added between pages already in effect, a decimal is added. For example, a new page added between Page 14 and 15 would be 14.1.
- <u>B.</u> **Page Revision Numbers -** Revision numbers also appear in the upper right corner of each page. These numbers are used to determine the most current page version on file with the Commission. For example, the 4<sup>th</sup> revised Page 14 cancels the 3"<sup>1</sup> revised Page 14.
- C. **Paragraph Numbering Sequence -** There are various levels of paragraph coding. Each level of coding is subservient to its next higher level:

2

2.1

2.1.1

2.1.1.1

**D.** Check Sheets - When a tariff filing is made with the Commission, an updated Check Sheet accompanies the tariff filing. The Check Sheet lists the pages contained in the tariff, with a cross-reference to the current revision number. When new pages are added, the Check Sheet is changed to reflect the revision. An asterisk designates all revisions made in a given filing (\*). There will be no other symbols used on this page if these are the only changes made to it (i.e., the format, etc. remain the same, just revised revision levels on some pages.) The tariff user should refer to the latest Check Sheet to find out if a particular page is the most current on file with the Commission.

Issued:	, 2009	Effective Date:	, 2009

(S)

(T)

# EXPLANATION OF SYMBOLS AND REFERENCE MARKS USED IN THIS TARIFF

The following symbols shall be used in this tariff for the purpose indicated below:

(C) - To signify a changed regulation or rate structure.
 (D) - To signify discontinued material.
 (I) - To signify an increased rate.
 (M) - To signify a move in the location of text.
 (N) - To signify a new rate, service, regulation, condition or sheet
 (R) - To signify a reduction in rate

To signify a change in text for clarification

To signify reissued material.

Issued: , 2009 Effective Date:\_\_\_\_\_\_, 2009

#### **DEFINITIONS**

Certain terms used generally throughout this tariff are described below.

#### Act

The Communications Act of 1934 (47 USC 153(R)), as amended by the Telecommunications Act of 1996, Public Law 104-104, 110 State. 56 (1996), codified throughout 47 USC and as interpreted by applicable law.

#### **Advance Payment**

A payment required before the start of service.

#### **Access Services**

The Company's intrastate telephone services offered pursuant to this tariff

## **Automatic Number Identification (ANI)**

A multi-frequency, signaling common switching feature that provides the automatic transmission of a seven or ten digit number and information digits to the customers premise for calls originating in the LATA to identify the calling party or station.

## **Calling Party Number**

(CPN) means a Signaling System 7 parameter whereby the ten (10) digit number is used to identify the calling or originating party.

#### **Central Office Prefix**

The first three digits (NXX) of the seven digit telephone number assigned to an end user's local exchange service.

#### **Central Office Switch**

A switch used to provide Telecommunications Services, including, but not limited to an End Office Switch or End Office which is a switching entity that is used to terminate Customer station lines for the purpose of interconnection to each other and to trunks; and, a Tandem Switch or Tandem Office or Tandem that is a switching entity that has billing and recording capabilities and is used to aggregate traffic and deliver traffic to carriers' aggregation points, points of termination, or points of presence, and to provide Switched Exchange Access Services. A Central Office Switch may also be employed as a combination End Office/Tandem Office Switch.

Issued:	, 2009	Effective Date:	, 2009
	,		,

#### **Collocation**

An arrangement where the equipment of a local exchange carrier is installed and maintained at the premise of another LEC.

#### **Common Carrier**

Denotes any individual, partnership, association, joint- stock company, trust, government entity or corporation engaged for hire in intrastate communication by wire or radio, between two or more exchanges.

#### Commission

Utah Public Service Commission, the State agency responsible for the regulation of telecommunications service within a particular state's border.

## **Common Channel Signaling Network**

Digital data network carrying signaling, routing and control information which interfaces with the telecommunications network.

## **Company**

IntelePeer, Inc., the issuer of this tariff

#### Customer

The person, firm or corporation that orders service and is responsible for the payment of charges and compliance with the Company's regulations.

#### **Customer Proprietary Network Information (CPNI)**

Information that relates to the quantity, technical configuration, type, destination, location, and amount of use of a telecommunications service subscribed to by any customer of a the Company and that is made available to the Company by the customer solely by virtue of the Company-customer relationship; and information contained in the bills pertaining to telephone exchange service or telephone toll service received by a customer of a carrier; except that such term (CPNI) does not include subscriber list information.

#### Digital Signal Level 1 (DS1)

1.544 Mbps first level signal in the time division multiplex hierarchy. In the time division multiplexing hierarchy of the telephone network, DS 1 is the initial level of multiplexing.

Issued:	, 2009	Effective Date:	, 2009

## Digital Signal Level 3 (DS3)

The 44.736 Mbps third level in the time division multiplex hierarchy. In the time division multiplexing hierarchy of the telephone network, DS3 is defined as the third level of multiplexing.

#### **End Office**

With respect to each NPA-NXX code prefix assigned to the Company, the location of the Company's "end office" for purposes of this tariff shall be the point of interconnection associated with an NPA-NXX code as listed in 2.8.2(A) following.

#### **End User**

A customer of telecommunications service who is not a telecommunications carrier.

#### **Exchange**

A unit generally smaller than a Market Service Area (MSA), established by a local service provider for the administration of communications services in the specific area. One or more exchanges comprise a MSA.

## **Exchange Telephone Company**

Denotes any individual, partnership, association, joint- stock company, trust, or corporation engaged in providing switched communication within an exchange.

#### **Extended Area Service (EAS)**

A switching and trunking arrangement which provides for non-optional, unlimited, two-way, flat rate calling service between two or more telephone exchanges, provided at either the applicable local exchange rate or the applicable local exchange rate plus an EAS increment rather than at the toll message rate.

#### **Facility or Facilities**

Lines, conduits, ducts, poles, wires, cables, cross-arms, receivers, transmitters, instruments, machines, appliances, instrumentalities and all devices, real estate, easements, apparatus, property and routes used, operated, owned or controlled by a telecommunications company to facilitate the provision of telecommunications service.

Issued:	. 2009	Effective Date:	. 2009

#### **FCC**

The Federal Communications Commission

#### Grandfathered

Services ordered under the provisions of this tariff but are no longer available to new customers.

#### **Individual Case Basis**

A condition in which the regulations (if applicable), rates and charges for an offering under the provision of this tariff are developed based on the circumstance in each case

#### **Intrastate Access Service**

Provides for a two-point communications path between a Customer's premises or a collocated interconnection location and an end user's premises for originating and terminating calls within the state.

#### **Internet Traffic**

Any traffic exchanged between Company and the customer that is originated by or delivered to an Enhanced Service Provider, including an Internet Service Provider.

#### **Local Access and Transport Area (LATA)**

A Local Access and Transport Area established pursuant to the Modification of Final Judgment entered by the United States District Court for the District of Columbia in Civil Action No. 82-0192; or any other geographic area designated as a LATA in the NATIONAL EXCHANGE CARRIER ASSOCIATION, INC. TARIFF F.C.C. NO. 4.

#### **Local Exchange Routing Guide (LERG)**

The telecommunications industry database tool used to provision NPA/NNXs and provide routing information to facilitate call completion.

Issued:	, 2009	Effective Date:	, 2009

#### **Local Traffic**

Traffic, including but not limited to Internet Traffic, that is originated by a Customer, user, subsidiary or affiliate of the Customer on the customer's network or service and terminates to the Company or another customer of the Company on the Company's network, within a given local calling area, or expanded area service ("EAS") area as defined in the originating carrier's effective Customer Tariffs, or, if the Commission has defined local calling areas applicable to all LECs, then as so defined by the Commission.

Loss

The value placed on injury or damages due to an accident caused by another's negligence, a breach of contract or other wrongdoing.

## **Meet Point Billing**

A situation when two or more exchange telephone companies are involved in the provisioning and billing of telecommunications service.

**MOUs** 

Minutes of Use

### **North American Numbering Plan (NANP)**

A three-digit area code (also known as a Number Plan Area (NPA), and a seven digit telephone number, made up of a three digit central office code and a four-digit station number. The NANP is administered by the North American Numbering Plan Administrator (NANPA).

#### **Originating Carrier**

The carrier whose end user originates a call on the carrier's network or switching equipment.

Issued:	, 2009	Effective Date:	, 2009

## **Percentage of Interstate Usage (PIU)**

The interstate jurisdictional use of a telecommunications service, as reported by the customer. This percentage is stated as a whole number percentage (a number from 0 through 100 percent) which is the best estimate of the percentage of the total use of the service that will be interstate in nature by the customer.

## **Percent of Local Usage (PLU)**

The local jurisdictional use of a telecommunications service as reported by the customer. This percentage is stated as a whole number percentage (a number from 0 through 100 percent) which is the best estimate of the percentage of the total use of the service that will be local in nature by the customer.

## **Point of Interconnection (POI)**

The physical location, building or equipment where two separate networks connect to each other in order to pass telecommunications traffic and signaling.

## **Recurring Charges**

The monthly charges to the Customer for services, facilities and equipment which continue to apply for duration of the service.

Issued:	, 2009	Effective Date:	, 2009

#### **Service Commencement Date**

The first date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this tariff in which case the Service Commencement Date is the date of the Customer's acceptance. The Company and the Customer may mutually agree on a substitute Service Commencement Date. If the Company does not have an executed Service Order from a Customer, the Service Commencement Date will be the first date on which the service or facility was used by a Customer.

#### **Service Order**

The request for access services, either written or electronic, executed by the Customer and the Company in the format devised by the Company. Such a request for service by the customer and the acceptance of the request by the Company initiates the respective obligations of the parties as set forth therein and pursuant to this tariff, but the duration of the service is calculated from the Service Commencement Date. Should a Customer use the Company's access service without an executed Service Order, the Company will then request the Customer to submit a Service Order.

#### **Serving Wire Center**

The wire center from which the customer designated premises would normally obtain dial tone from the Company.

#### Shared

A facility or equipment system or subsystem that can be used simultaneously by several Customers.

#### **Signaling Transfer Point (STP)**

A specialized switch that provides SS7 network access and performs SS7 message routing and screening.

Issued:	, 2009	Effective Date:	, 2009

## **Tandem Transit or Third Party Tandem Transit**

An arrangement that provides indirect Interconnection, via a third party's tandem, that is used by the Company and the Customer for the reciprocal exchange of local and IntraLATA Toll Traffic between their respective networks and/or service.

#### **Telecommunications Service**

The offering of Telecommunications for a fee directly to the public, or to such classes of users as to be effectively available directly to the public, provided through a system of switches, transmission equipment, or other facilities (or combination thereof) by which a subscriber can originate and terminate a Telecommunications Service, regardless of the facilities used. Telecommunications Service generally provides the Customer with a telephonic connection to, and a unique telephone number address on, the public switched telecommunications network, and enables such Customer to place or receive calls to all other stations served by the public switched telecommunications network. It also provides service within a telephone exchange or within a connected system of telephone exchanges within the same exchange area operated to furnish subscribers intercommunicating service of the character ordinarily furnished by a single exchange

## **Terminating Carrier**

The carrier who terminates a call to the carrier's end user a call on the carrier's network or switching equipment.

#### **Termination Point**

The point of demarcation within a customer designated premises or point of interconnection at which the Company's responsibility for the provision of service ends.

#### **Toll Traffic**

Between stations in different exchange areas for which there is a separate charge not included in contracts for local exchange service.

#### User

A Customer or any other person authorized by the Customer to use service provided under this tariff.

Issued:	, 2009	Effective Date:	, 2009

#### Julie Barghouthi

#### **Wire Center**

A building in which one or more end offices, used for the provision of Exchange Services, are located.

#### V and H Coordinates Method

A method of computing the distance, in airline miles, between two point by utilizing an established formula which is based on the vertical and horizontal coordinates of the two points.

Issued:	, 2009	Effective Date:	, 2009

#### **SECTION 1.** AP PLICATION OF TARIFF

- 1.1.1 This tariff applies to intrastate access service supplied to Customers for origination and termination of traffic to and from Central Office codes directly assigned to IntelePeer, Inc. and traffic from those central office codes assigned to other carriers that transits IntelePeer's facilities.
- 1.1.2 Except as provided in 3.1.3(A), this tariff applies only to the extent that services provided hereunder are used by a Customer for the purpose of originating or terminating intrastate communications. A communication is "intrastate" only if all points of origination and termination are located within the State of Utah.

Issued:	2009	Effective Date:	2009

#### **SECTION 2. REGULATIONS**

#### 2.1 Undertaking of the Company

### 2.1.1 Scope

The Company undertakes to furnish access services in accordance with the terms and conditions set forth in this tariff

#### 2.1.2 Shortage of Facilities

All service is subject to the availability of suitable facilities. The furnishing of service under this tariff is subject to the availability on a continuing basis of all the necessary facilities and equipment and is limited to the capacity of the Company's facilities to furnish service as required at the sole discretion of the Company.

#### 2.1.3 Terms and Conditions

- (A) Service is provided on the basis of a minimum period of at least one-month, 24-hours per day. For the purpose of computing charges in this tariff, a month is considered to have thirty (30) days.
- (B) The Company reserves the right to discontinue service, limit service, or to impose requirements when Customer is using the service in violation of the provisions of this tariff, or as required to meet changing regulatory or statutory rules and standards, or when such rules and standards have an adverse material affect of on the business or economic feasibility of providing service, as determined by the Company in its reasonable

Issued:	, 2009	Effective Date:	, 2009

## 21 Undertaking of the Company

#### 2.1.3 Terms and Conditions

- (C) In any action between the parties to enforce any provision of this tariff, the prevailing party shall be entitled to recover its legal fees and court costs from the non-prevailing party in addition to other relief a court may award.
- (D) Customers may be required to enter into written service orders which shall contain or reference a specific description of the service ordered, the rates to be charged, the duration of the services, and the terms and conditions in this tariff. Customers will also be required to execute any other documents as may be reasonably requested by the Company.
- (E) This tariff shall be interpreted and governed by the laws of the State of Utah without regard for its choice of laws provision.

Issued:	, 2009	Effective Date:	, 2009

21 Undertaking of the Company (Cont'd.)

#### 2.1.4 Liability of the Company

- (A) Except as otherwise stated in this Tariff, the liability of the Company for damages arising out of the furnishing of its Services, including but not limited to mistakes, omissions, interruptions, delays, or errors, or other defects, representations, or use of these services or arising out of the failure to furnish the service, whether caused by acts or omission, shall be limited to the extension of allowances for interruption as set forth in Section 2.7. The extension of such allowances for interruption shall be the sole remedy of the Customer and the sole liability of the Company. The Company will not be liable for any direct, indirect, incidental, special, consequential, exemplary or punitive damages to the Customer as a result of any Company service, equipment or facilities, or any acts or omissions or negligence of the Company's employees or agents.
- (B) The Company shall not be liable for any delay or failure of performance or equipment due to causes beyond its control, including but not limited to: acts of God, fire, flood, explosion or other catastrophes; any law, order, regulation, direction, action, or request of the United States government, or of any other government, including state and local governments having or claiming jurisdiction over the Company, or of any department, agency, commission, bureau, corporation, or other instrumentality of any one or more of these federal, state, or local governments, or of any civil or military authority; national emergencies; insurrections; riots; wars; unavailability of rights-of-way materials; or strikes, lock-outs, work stoppages, or other labor difficulties.
- (C) The Company shall not be liable for (a) any act or omission of any entity furnishing to the Company or to the Company's Customers facilities or equipment used for interconnection with Network Services; or (b) for the acts or omissions of common carriers or warehousemen.

Issued:	. 2009	Effective Date:	. 2009

- 2.1 Undertaking of the Company (Cont'd.)
  - 2.1.4 Liability of the Company (Cont'd.)
    - (D) The Company shall not be liable for any damages or losses due to the fault or negligence of the Customer or due to the failure or malfunction of equipment or facilities provided by the Customer or third parties.
    - (E) The Company does not guarantee nor make any warranty with respect to installations it provides for use in an explosive atmosphere. The Customer indemnifies and holds the Company harmless from any and all loss, claims, demands, suits, or other action, or any liability whatsoever, whether suffered, made, instituted, or asserted by any other party or person(s), and for any loss, damage, or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, removal, presence, condition, location, or use of any installation so provided. The Company reserves the right to require each Customer to sign an agreement acknowledging acceptance of the provisions of this Section 2.1.4(E) as a condition precedent to such installations.
    - (F) The Company is not liable for any defacement of or damage to Customer premises resulting from the furnishing of services or equipment on such Premises or the installation or removal thereof, unless such defacement or damage is caused by gross negligence or willful misconduct of the Company's agents or employees.
    - (G) The Company shall be indemnified, defended and held harmless by the Customer against any claim, loss or damage arising from Customer's use of services, involving claims for libel, slander, invasion of privacy, or infringement of copyright arising from the Customer's own communications.

Issued:	, 2009	Effective Date:	, 2009

21 Undertaking of the Company (Cont'd.)

#### 2.1.4 L iability of the Company (Cont'd.)

- (H) The entire liability for any claim, loss, damage or expense from any cause whatsoever shall in no event exceed sums actually paid the Company by the Customer for the specific services in the month in which the event giving rise to the liability occurred. No action or proceeding against the Company shall be commenced more than one year after the event giving rise to the liability occurred.
- (I) THE COMPANY MAKES NO WARRANTIES OR REPRESENTATIONS, EXPRESS OR IMPLIED EITHER IN FACT OR BY OPERATION OF LAW, STATUTORY OR OTHERWISE, INCLUDING WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR USE, EXCEPT THOSE EXPRESSLY SET FORTH HEREIN.

#### 2.1.5 Claims

The Company shall be indemnified and saved harmless by the Customer from and against all loss, liability, damage and expense, including reasonable counsel fees, due to claims of libel, slander, or infringement of copyright in connection with the material transmitted over the Company's facilities; and any other claim resulting from any act or omission of the Customer or end users of the Customer relating to the use of the Company's services or facilities.

Issued:	, 2009	Effective Date:	, 2009

## 21 Undertaking of the Company (Cont'd.)

#### 2.1.6 Provision of Equipment and Facilities

- (A) Except as otherwise indicated, customer-provided station equipment at the Customer's premises for use in conjunction with this service shall be so constructed, maintained and operated as to work satisfactorily with the facilities of the Company.
- (B) The Company shall not be responsible for the installation, operation or maintenance of any Customer-provided communications equipment. Where such equipment is connected to service furnished pursuant to this tariff, the responsibility of the Company shall be limited to the furnishing of services under this tariff and to the maintenance and operation of such services in the proper manner. Subject to this responsibility, the Company shall not be responsible for:
  - (1) the through transmission of signals generated by Customer-provided equipment or for the quality of, or defects in, such transmission; or
  - (2) the reception of signals by Customer-provided equipment; or
  - (3) network control signaling where such signaling is performed by Customer-provided network control signaling equipment.

#### 2.1.7 Ownership of Facilities

Title to all facilities provided in accordance with this tariff remains in the Company, its agents, contractors or suppliers.

Issued:	, 2009	Effective Date:	, 2009

#### 2.2 Prohibited Uses

- (A) The services the Company offers shall not be used for any unlawful purpose or for any use for which the Customer has not obtained all required governmental approvals, authorization, licenses, consents and permits.
- (B) The Company may require applicants for service who intend to use the Company's offering for resale and/or for shared use to file a letter with the Company confirming that their use of the Company's offerings complies with relevant laws and regulations, policies, orders, and decisions.
- (C) The Company may require a Customer to immediately shut down its transmission if such transmission is causing interference to others.
- (D) A Customer, joint user, or authorized user may not assign, or transfer in any manner, the service or any rights associated with the service without the written consent of the Company. The Company will permit a Customer to transfer its existing service to another entity if the existing Customer has paid all charges owed to the Company for regulated access services. Such a transfer will be treated as a disconnection of existing service and installation of new service, and non-recurring installation charges as stated in this tariff will apply.

Issued:	. 2009	Effective Date:	, 2009

#### 23 Obligations of the Customer

#### 2.3.1 Customer Premises Provisions

- (A) The Customer shall provide the personnel, power and space required to operate all facilities and associated equipment installed on the premises of the Customer.
- (B) The Customer shall be responsible for providing Company personnel access to premises of the Customer at any reasonable hour for the purpose of testing the facilities or equipment of the Company.

## 2.3.2 Liability of the Customer

The Customer will be liable for damages to the facilities of the Company caused by negligence or willful acts of its offices, employees, agents or contractors of the Customer where such negligence is not the direct result of the Company's negligence.

Iccued.	2000	Effective Date:	2009

- 2.3 Obligations of the Customer (Cont'd.)
  - 2.3.3 Jurisdictional Report Requirements
    - (A) For Feature Group D Switched Access Service(s), the Company, where jurisdiction can be determined from the call detail, will determine the projected interstate percentage as follows. For originating access minutes, the projected interstate percentage will be developed on a monthly basis by end office trunk group when the Feature Group D Switched Access Service access minutes are measured by dividing the measured interstate originating access minutes (the access minutes where the calling number is in one state and the called number is in another state) by the total originating access minutes when the call detail is adequate to determine the appropriate jurisdiction. For terminating access minutes, the Customer has the option to provide the Company with a projected PIU factor. Customers who provide a PIU factor shall supply the Company with an interstate percentage of the Feature Group D terminating access minutes for each account to which the Customer may terminate traffic.

Issued:	. 2009	Effective Date:	. 2009

- Obligations of the Customer (Cont'd.)
  - 2.3.3 Jurisdictional Report Requirements (Cont'd.)
    - (A) (Cont'd.)

Should the Customer not supply a terminating PIU Factor, the data used by the Company to develop the projected interstate percentage for originating access minutes will be used to develop projected interstate percentage for such terminating access minutes. When a Customer orders Feature Group D Switched Access Service, the Customer shall supply a projected interstate percentage of use for each end office trunk group involved to be used in the event that originating call details are insufficient to determine the jurisdiction for the call. This percentage shall be used by the Company as the projected interstate percentage for such call detail. For purposes of developing the projected interstate percentage, the Customer shall utilize the same considerations as those set forth in Section 2.3.3(B) following.

The Company will designate the number obtained by subtracting the projected interstate percentage for originating and terminating access minutes from 100 (100 -projected interstate percentage = intrastate percentage) as the projected intrastate percentage of use.

- (B) For purposes of developing the projected interstate percentage, the Customer shall consider every call that enters the Customer's network at a point within the same state as the state where the called station is located to be intrastate and every call that enters the Customer's network at a point in a state different from the state in which the called station is located to be interstate.
- (C) These whole number percentages will be used by the Company to apportion the use, rates, and/or nonrecurring charges between interstate and intrastate until a revised report is received.

Issued:	. 2009	Effective Date:	. 2009

- 2.3 Obligations of the Customer (Cont'd.)
  - 2.3.3 Jurisdictional Report Requirements (Cont'd.)
    - (D) The projected interstate percentage of use will be used to determine the charges as follows:

The number of access minutes for a trunk group will be multiplied by the projected interstate percentage of use to determine the interstate access minutes. (i.e., number of access minutes x projected interstate percentage of use = interstate access minutes). The number of interstate access minutes so determined will be subtracted from the total number of access minutes (i.e., number of access minutes - interstate access minutes = intrastate access minutes). The intrastate access minutes for the group will be billed as set forth in Section 5 following.

Effective on the first of January, April, July and October of each year, the (E) Customer may update the jurisdictional reports that require a projected interstate percentage. The Customer shall forward to the Company, to be received no later than 20 calendar days after the first of each such month, a revised report showing the interstate percentage of use for the past three months ending the last day of December, March, June and September, respectively, for each service arranged for interstate and intrastate use. Except as set forth in Section 2.3.3(A) preceding where jurisdiction can be determined from the recorded message detail, the revised report will serve as the basis for the next three months billing and will be effective on the bill date in the following month (i.e., February, May, August, and November) for that service. No prorating or back billing will be done based on the report. If the Customer does not supply the report, the Company will assume the percentage to be the same as that provided in the last quarterly report. For those cases in which a quarterly report has never been received from the Customer, the Company will assume the percentage to be the same as that provided in the order for service as set forth in Section 2.3.3(A) preceding.

Issued:	. 2009	Effective Date:	. 2009

Obligations of the Customer (Cont'd.)

## 2.3.3 Jurisdictional Report Requirements (Cont'd.)

- (F) The Customer reported projected interstate percentage of use as set forth in Section 2.3.3(A) preceding will be used for the apportionment of any monthly rates or nonrecurring charges associated with Feature Groups B or D Switched Access Service until the end of the quarter during which the service was activated. Thereafter, a projected interstate percentage for such apportionment will be developed quarterly by the Company based on the data used to develop the projected interstate percentage of use as set forth in Section 2.3.3(A) preceding. Where call detail is insufficient to make such a determination, the Customer will be requested to project an interstate percentage of use to be used by the Company for such apportionment.
- (G) The Customer shall keep sufficient detail from which the percentage of interstate use can be ascertained and upon request of the Company make the records available for inspection. Such a request will be initiated by the Company no more than once per year. The Customer shall supply the data within thirty (30) calendar days of the Company request.
- (H) The Customer may provide an additional percentage of interstate use for Entrance Facility and Direct Trunked Transport subject to the reporting requirements previously listed in this section. The percentage of interstate use may be provided per individual facility or at the billing account level. Should the Customer not provide a percentage of interstate use, the Company will use the reported Feature Group B or Feature Group D aggregated percentage of interstate use.
- (I) In the absence of both a customer provided PIU, and the necessary jurisdictional information in the call detail to determine jurisdiction, the default PIU will be zero.

Issued:	, 2009	Effective Date:	, 2009

## 2.4 Customer Equipment and Channels

#### 2.4.1 Interconnection of Facilities

(A) In order to protect the Company's facilities and personnel and the service furnished to other Customers by the Company from potentially harmful effects, the signals applied to the Company's service shall be such as not to cause damage to the facilities of the Company. Any special interface equipment necessary to achieve the compatibility between facilities of the Company and the channels or facilities of others shall be provided at the Customer's expense.

## 2.4.2 Inspections

- (A) The Company may, upon notification to the Customer, at a reasonable time, make such tests and inspections as may be necessary to determine that the requirements regarding the equipment and interconnections are being complied with in respect to the installation, operation and maintenance of Customer-provided equipment and in the wiring of the connection of Customer channels to Company-owned facilities.
- (B) If the protective requirements in connections with Customer-provided equipment are not being complied with, the Company may take such action as necessary to protect its facilities and personnel and will promptly notify the Customer by registered mail in writing of the need for protective action. In the event that the Customer fails to advise the Company within 10 days after such notice is received or within the time specified in the notice that corrective action has been taken, the Company may take whatever additional action is deemed necessary, including canceling service, to protect its facilities and personnel from harm. The Company will upon request 24 hours in advance provide Customer with a statement of technical parameters that the Customer's equipment must meet.

Issued:	, 2009	Effective Date:	, 2009
issucu.	, 2009	Effective Date	_, 2009

#### 2.5 Customer Deposits and Advance Payments

#### 2.5.1 Advance Payments

To safeguard its interests, the Company may require a Customer to make an advance payment before services and facilities are furnished. The advance payment will not exceed an amount up to two months of estimated monthly usage charges. In addition, where special construction is involved, the advance payment may also include an amount equal to the estimated non-recurring charges for the special construction and recurring charges (if any) for a period to be set between the Company and the Customer. The advance payment will be credited to the Customer's initial bill. An advance payment may be required in addition to a deposit.

## 2.5.2 Deposits

- (A) To safeguard its interests, the Company may require the Customer to make a deposit to be held as a guarantee for the payment of charges. A deposit does not relieve the Customer of the responsibility for the prompt payment of bills on presentation. The deposit will not exceed an amount equal to:
  - an estimated two months' customary use or estimated two months' bill where applicable. Customary use is calculated using charges for the previous three months of service.
  - (2) the charges that would apply for the minimum payment period for a service or facility which has a minimum payment period of more than one month; except that the deposit may include an additional amount in the event that a termination charge is applicable.
- (B) A deposit may be required in addition to an advance payment.

Issued:	, 2009	Effective Date:	, 2009

25 Customer Deposits and Advance Payments (Cont'd.)

#### 2.5.2 Deposits (Cont'd.)

When a service or facility is discontinued, the amount of a deposit, if any, will be applied to the Customer's account and any credit balance remaining will be refunded. Before the service or facility is discontinued, the Company may, at its option, return the deposit or credit it to the Customer's account. If the amount of the deposit is insufficient to cover the balance due to the Customer's account, the Company retains the right to collect any amounts owing after the deposit has been applied plus any costs related to the collection of any remaining balance.

Deposits held will accrue interest at a rate specified by the Commission without deductions for any taxes on such deposits. Interest will not accrue on any deposit after the date on which reasonable effort has been made to return it to the Customer.

Issued:	2009	Effective Date:	2009

## 2.6 Payment Arrangements

## 2.6.1 Payment for Service

The Customer is responsible for the payment of all charges for facilities and services furnished by the Company to the Customer.

#### (A) Taxes

The Customer is responsible for payment of any sales, use, gross receipts, excise, access or other local, state and federal taxes, charges or surcharges (however, designated) (excluding taxes on the Company's net income) imposed on or based upon the provision, sale or use of Network Services.

Icened:	2009	Effective Date:	2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.1 Payment for Service (Cont'd.)
    - (B) Surcharges, Fees and Assessments

The Customer is responsible for payment of any surcharge, assessment or fee, including but not limited to universal service fees, 911 charges, right of way fees or other types of infrastructure fees, and regulatory assessments, where allowed by applicable law.

## 2.6.2 Billing and Collection of Charges

The Customer is responsible for payment of all charges incurred by the Customer or other users for services and facilities furnished to the Customer by the Company.

- (A) Non-recurring charges are due and payable within thirty (30) days after the date of the invoice.
- (B) The Company shall present invoices for Recurring Charges monthly to the Customer, in advance of the month in which service is provided, and Recurring Charges shall be due and payable within thirty (30) days after the date of the invoice. When billing is based upon customer usage, usage charges will be billed monthly for the preceding billing period.
- (C) When service does not begin on the first day of the month, or end on the last day of the month, the charge for the fraction of the month in which service was furnished will be calculated on a pro rate basis. For this purpose, every month is considered to have thirty (30) days.

Issued:	. 2009	Effective Date:	. 2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.2 Billing and Collection of Charges (Cont'd.)
    - (D) Billing of the Customer by the Company will begin on the Service Commencement Date, which is the day on which the Company notifies the Customer that the service or facility is available for use, except that the Service Commencement Date may be postponed by mutual agreement of the parties, or if the service or facility does not conform to standards set forth in this tariff or the Service Order. Billing accrues through and includes the day that the service, circuit, arrangement or component is discontinued.
    - (E) If any portion of the payment is received by the Company after the date due, or if any portion of the payment is received by the Company in funds which are not immediately available upon presentment, then a late payment penalty shall be due to the Company. The late payment penalty shall be the portion of the payment not received by the date due, multiplied by a late factor. The late factor shall be the lesser of:
      - (1) a rate of 1.5 percent per month; or
      - (2) the highest interest rate which may be applied under state law for commercial transactions.
    - (F) The Customer will be assessed a charge of twenty-five dollars (\$25.00) for each check submitted by the Customer to the Company which a financial institution refuses to honor.

Issued:	, 2009	Effective Date:	, 2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.2 Billing and Collection of Charges (Cont'd.)
    - (G) Customers have up to 90 days (commencing 5 days after remittance of the bill) to initiate a dispute over charges or to receive credits. The Company shall investigate and resolve all disputes within fifteen (15) days of receipt of the dispute and the Company's resolution of such a dispute is final. Any portion of a disputed amount deemed payable by the Company must be paid in full within ten (10) days of resolution or Customer's service may be subject to disconnection and late payment charges imposed on the overdue amount.

Any unresolved dispute may be directed in writing to the Commission's Consumer Services Division).

(H) If service is disconnected by the Company in accordance with Section 2.6.3 following, and later restored, restoration of service will be subject to all applicable installation charges.

Issued:	2009	Effective Date:	2009

## 2.6 Payment Arrangements (Cont'd.)

#### 2.6.3 Discontinuance of Se rvice for Cause

The Company may disconnect service, without incurring any liability, for any of the following reasons provided it has notified the customer of its intent, in writing, to disconnect service and has allowed the customer a reasonable time of not less than eight (8) working days, after the date on which the notice is mailed, in which to remove the cause for disconnection.

- (A) Upon nonpayment of any amounts owing to the Company, the Company may, by giving 24 hours prior written notice to the Customer, discontinue or suspend service without incurring any liability.
- (B) Upon violation of any of the other material terms or conditions for furnishing service the Company may, by giving 24 hours prior notice in writing to the Customer, discontinue or suspend service without incurring any liability if such violation continues during that period.
- (C) Upon condemnation of any material portion of the facilities used by the Company to provide service to a Customer or if a casualty renders all or any material portion of such facilities inoperable beyond feasible repair, the Company, by notice to the Customer, may discontinue or suspend service without incurring any liability.
- (D) Upon the Customer's insolvency, assignment for the benefit of creditors, filing for bankruptcy or reorganization, or failing to discharge an involuntary petition within the time permitted by law, the Company may immediately discontinue or suspend service without incurring any liability.
- (E) Upon any governmental prohibition or required alteration of the services to be provided or any violation of an applicable law or regulation, the Company may immediately discontinue service without incurring any liability.

Issued:	, 2009	Effective Date:	, 2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.3 Discontinuance of Service for Cause (Cont'd.)
    - (F) In the event of fraudulent use of the Company's network, the Company may without notice suspend or discontinue service. The Customer will be liable for all related costs. The Customer will also be responsible for payment of any reconnection charges.
    - (G) Upon the Company's discontinuance of service to the Customer under Section 2.6.3(A) or 2.6.3(B), the Company, in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff, may declare all future monthly and other charges which would have been payable by the Customer during the remainder of the term for which such services would have otherwise been provided to the Customer to be immediately due and payable (discounted to present value at six percent).
  - 2.6.4 Notice to Company for Cancellation of Service

Customers desiring to terminate service shall provide Company thirty (30) days written notice of desire to terminate service. Notice should be sent to the following address:

IntelePeer, Inc. Attention: Customer Care 2855 Campus Drive, Suite 200 San Mateo, CA 94403

Issued: 2000 Effective Date: 2				
Issued	, 2009	Effective Date:	, 2009	Issued:

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.5 Ordering, Rating and Billing of Access Services Where More Than One Exchange Telephone Company is Involved

Meet point billing applies when more than one Exchange Telephone Company is involved in the provision of Access Service. All recurring and nonrecurring charges for services provided by each Exchange Telephone Company are billed under each company's applicable rates as set forth in Section 2.6.5 (A) following.

The Company accepts and adheres to the Ordering and Billing Forum guidelines, Multiple Exchange Carrier Access Billing (MECAB) and Multiple Exchange Carrier Ordering and Design (MECOD).

The Company will handle ordering, rating and billing of Access Services under this tariff where more than one Exchange Telephone Company is involved in the provision of Access Service as follows.

(A) For Feature Group D Switched Access Service, when the first point of switching is not in the same Exchange Telephone Company's territory as the Customer premises, the Customer must supply a copy of the order to the Exchange Telephone Company in whose territory the Customer premises is located and any other Exchange Telephone Company(s) involved in providing the service.

Each Exchange Telephone Company will provide the portion of Local Transport to an interconnection point (IP) with another Exchange Telephone Company, and will bill the charges in accordance with its Access Service tariff The rate for the Transport elements will be determined as set forth in (B) following. All other appropriate charges in each Exchange Telephone Company tariff are applicable.

Issued:	, 2009	Effective Date:	, 2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.5 Ordering, Rating and Billing of Access Services Where More Than One Exchange Telephone Company is Involved (Cont'd.)
    - (B) The charge for the Local Transport Facility and Termination rate elements for services provided as set forth in Section 2.6.5 (A) preceding are determined as follows:
      - (1) Determine the appropriate Switched Access Local Transport mileage by computing the airline mileage between the two ends of the Local Transport Facility, as defined in 3.1.2(B) following. Determine the airline mileage for the Local Transport Facility charge using the V&H method as set forth in Section 2.8.2 following.
      - (2) For Feature Groups D Switched Access Service, the Local Transport Facility and Termination charges are determined by using the steps set forth in (a) through (c) following for the total Local Transport-Common Switched Transport charges.
        - (a) Multiply:

The number of access minutes

bs

the number of airline miles as determined in (1) preceding

by

the Company's appropriate Local Transport Facility per mile per access minute rate

by

the Company's billing percentage factor.

Issued:	, 2009	Effective Date:	, 2009
	, = 0 0 /		,

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.5 Ordering, Rating and Billing of Access Services Where More Than One Exchange Telephone Company is Involved (Cont'd.)
    - (B) (Cont'd.)
      - (2) (Cont'd.)
        - (b) Multiply:

The number of access minutes

by

the Company's appropriate Local Transport Termination per minute rate. The resulting amount is the Company's total Local Transport Termination charge.

(c) Add:

The products of (a) and (b) for the Company's total Local Transport-Common Switched Transport charges.

Issued:	2009	Effective Date:	2009

- 2.6 Payment Arrangements (Cont'd.)
  - 2.6.5 Ordering, Rating and Billing of Access Services Where More Than One Exchange Telephone Company is Involved (Cont'd.)
    - (C) The interconnection points will be determined by the Exchange Telephone Companies involved. The billing percentage (BP) factor for the Company for the service between the involved offices is listed in the NATIONAL EXCHANGE CARRIER ASSOCIATION, INC. TARIFF F.C.C. NO. 4.
    - (D) Should any changes be made to the meet point billing arrangements as set forth in Section 2.6.5 (A) preceding, the Company will give affected Customers thirty (30) days' notice.
    - (E) Should the Company act as an intermediate, non-terminating local exchange carrier,

Local Transport Termination rates, as determined in Section 2.6.5 (B) preceding, will not be applied to the meet point billing arrangement.

Issued:	, 2009	Effective Date:	, 200
ibbucu.	. 2007	Directive Dute.	

## 2.6 Payment Arrangements (Cont'd.)

## 2.6.6 Changes in Se rvice Requested

If the Customer makes or requests material changes in circuit engineering, equipment specifications, service parameters, premises locations, or otherwise materially modifies any provision of the application for service, the Customer's installation fees shall be adjusted according to the term and conditions set forth in 3.1.1.2 following, Access Order Modifications.

## 2.6.7 Customer Overpayment

The Company will pay interest on a Customer overpayment. Customer overpayment shall mean a payment to the Company in excess of the correct charges for service when caused by erroneous billing by the Company. The rate of interest shall be the unadjusted interest rate paid on Customer deposits or the late payment penalty rate, whichever is greater. Interest shall be paid from the date when the Customer overpayment was made, adjusted for any changes in the deposit interest rate or late payment penalty rate, and compounded monthly, until the date when the overpayment is refunded. No interest shall be paid on Customer overpayments that are refunded within thirty (30) days after such overpayment is received by the Company.

Issued.	2009	Effective Date:	2009

## 27 Allowances for Interruptions in Service

Except as set forth in 2.1.4(B) preceding and 2.7.2 following, interruptions in service, which are not due to the negligence of, or noncompliance with the provisions of this tariff by, the Customer or the operation or malfunction of the facilities, power or equipment provided by the Customer, will be credited to the Customer as set forth in 2.7.1 for the part of the service that the interruption affects.

## 2.7.1 Credit for Interruptions

- (A) A credit allowance will be made when an interruption occurs because of a failure of any component furnished by the Company under this tariff An interruption period begins when the Customer reports a service, facility or circuit to be interrupted and releases it for testing and repair. An interruption period ends when the service, facility or circuit is operative. If the Customer reports a service, facility or circuit to be inoperative but declines to release it for testing and repair, it is considered to be impaired, but not interrupted.
- (B) For calculating credit allowances, every month is considered to have thirty (30) days. A credit allowance is applied on a pro rata basis against the rates specified hereunder and is dependent upon the length of the interruption. Only those facilities on the interrupted portion of the circuit will receive a credit.

Issued:	2009	Effective Date:	2009

- 2.7 Allowances for Interruptions in Service (Cont'd.)
  - 2.7.1 Credit for Interruptions (Cont'd.)
    - (B) (Cont'd.)

A credit allowance will be given for interruptions of 30 minutes or more. Credit allowances shall be calculated as follows:

Interruptions of 24 Hours or Less

Length of Interruption	Interruption Period To Be Credited
Less than 30 minutes	None
30 minutes up to but not including 3 hours	1/10 Day
3 hours up to but not including 6 hours	1/5 Day
6 hours up to but not including 9 hours	2/5 Day
9 hours up to but not including 12 hours	3/5 Day
12 hours up to but not including 15 hours	4/5 Day
15 hours up to but not including 24 hours	One Day

Two or more interruptions of 15 minutes or more during any one 24-hour period shall be considered as one interruption.

Issued: , 2009 Effective Date: , 2009

- 27 Allowances for Interruptions in Service (Cont'd.)
  - 2.7.1 Credit for Interruptions (Cont'd.)
    - (B) (Cont'd.)

Interruptions Over 24 Hours and Less Than 72 Hours. Interruptions over 24 hours and less than 72 hours will be credited 1/5 day for each 3-hour period or fraction thereof. No more than one full day's credit will be allowed for any period of 24 hours.

Interruptions Over 72 Hours. Interruptions over 72 hours will be credited 2 days for each full 24-hour period. No more than thirty (30) days credit will be allowed for any one-month period.

Issued.	2009	Effective Date:	2009

- 2.7 Allowances for Interruptions in Services (Cont'd.)
  - 2.7.2 Limitations on Allowances

No credit allowance will be made for:

- (A) interruptions due to the negligence of, or noncompliance with the provisions of this tariff by the Customer, authorized user, joint user, or other common carrier providing service connected to the service of the Company;
- (B) interruptions due to the negligence of any person other than the Company, including but not limited to the Customer or other common carriers connected to the Company's facilities;
- (C) interruptions due to the failure or malfunction of non-Company equipment;
- (D) interruptions of service during any period in which the Company is not given full and free access to its facilities and equipment for the purpose of investigating and correcting interruptions;
- (E) interruptions of service during a period in which the Customer continues to use the service on an impaired basis;
- (F) interruptions of service during any period when the Customer has released service to the Company for maintenance purposes or for implementation of a Customer order for a change in service arrangements; or
- (G) interruption of service due to circumstances or causes beyond the control of Company.

Issued:	2009	Effective Date:	, 2009

2.7 Allowances for Interruptions in Service (Cont'd.)

## 2.7.3 Cancellation For Service Interruption

Cancellation or termination for service interruption is permitted only if any circuit experiences a single continuous outage of 8 hours or more or cumulative service credits equaling 16 hours in a continuous 12-month period. The right to cancel service under this provision applies only to the single circuit which has been subject to the outage or cumulative service credits.

Issued: , 2009 Effective Date:\_\_\_\_\_\_, 2009

## 28 Application of Rates

The regulations set forth in this section govern the application of rates for services contained in other sections of this tariff.

#### 2.8.1 Charges Based on Duration of Use

Customer traffic to end offices will be measured by the Company at end office switches. Originating and terminating calls will be measured by the Company to determine the basis for computing chargeable access minutes.

For originating calls over Feature Group D, usage measurement begins when the originating Feature Group D switch receives the first wink supervisory signal forwarded from the Customer's point of termination.

The measurement of originating call usage ends when the originating Feature Group B or D switch receives disconnect supervision from either the originating end user's end office, indicating the originating end user has disconnected, or the Customer's point of termination, whichever is recognized first by the switch.

For terminating calls over Feature Group D, the measurement of access minutes begins when the terminating Feature Group D switch receives answer supervision from the terminating end user's end office, indicating the terminating end user has answered.

The measurement of terminating call usage over Feature Group D ends when the terminating Feature Group D switch receives disconnect supervision from either the terminating end user's end office, indicating the terminating end user has disconnected, or the Customer's point of termination, whichever is recognized first by the switch.

Access minutes or fractions thereof are accumulated over the billing period for each end office and are then rounded up to the nearest access minute for each end office.

Issued:	, 2009	Effective Date:	, 2009

- 2.8 Application of Rates (Cont'd.)
  - 2.8.1 Charges Based On Duration of Use (Cont'd.)
    - 2.8.1.1 Rate Periods except Holidays

Applies to intraLATA intrastate traffic

		Times Applicable	
		To But Not	
Rate Period	From	Including	Days Applicable
			Monday - Friday
	8:00 A.M.	9:00 P.M.	Monday - Friday
Day	9:00 P.M.	11:00 P.M.	Monday - Friday
Evening	11:00 P.M.	8:00 A.M.	Weekend
Night	11:00 P.M.	8:00 A.M.	(Friday - Monday)

Issued: , 2009 Effective Date: , 2009

- 28 Application of Rates (Cont'd.)
  - 2.8.2 Rates Bas ed Upon Distance

Where the charges for service are specified based upon distance, the following rules apply:

(A) Distance between two points is measured as airline distance between the wire centers of the originating and terminating telephone lines. The wire center is a set of geographic coordinates, as referenced in the NATIONAL EXCHANGE CARRIER ASSOCIATION, INC. TARIFF FCC No. 4, associated with each NPA-NXX combination (where NPA is the area code and NXX is the first three digits of a seven-digit telephone number).

Issued:	, 2009	Effective Date:	, 2009

2.8	Application	of Rates	(Cont'd)
∠.0	Abblication	or raics	Com u. i

- 2.8.2 Rates Based Upon Distance (Cont'd.)
  - (B) The airline distance between any two wire centers is determined as follows:
    - (1) Obtain the "V" and "H" coordinates for each wire center from the above-referenced NECA tariff.
    - (2) Compute the difference between the "V" coordinates of the two wire centers; and the difference between the two "H" coordinates.
    - (3) Square each difference obtained in step (2) above.
    - (4) Add the square of the "V" difference and the square of the "H" difference obtained in step (3).
    - (5) Divide the sum of the squares by 10. Round to the next higher whole number if any fraction is obtained.
    - (6) Obtain the square root of the whole number result obtained above. Round to the next higher whole number if any fraction is obtained. This is the airline mileage.

<sup>(7</sup> )	Formula =		(In.	$V2)^{2} + (H1)^{2}$	H2) <sup>2</sup>
				10	
		v			

Issued:	, 2009	Effective Date:	, 2009
ibbaca.	, 200)	Effective Bate.	, 200

28 Application of Rates (Cont'd.)

### 2.8.3 Mileage

The mileage to be used to determine the Local Transport Facility monthly rates are calculated on the airline distance between the end office switch where the call carried by Local Transport originates or terminates and the customer's serving wire center. The V&H coordinates method is used to determine mileage. This method is set forth in Section 2.8.2.

The Local Transport Facility mileage rates are shown in Section 5.1.3 (B) in terms of per mile per access minute. To determine the rate to be billed, first compute the mileage. Should the calculation result in a fraction of a mile, the fraction will always be rounded up to the next whole mile before determining the mileage. Then multiply the mileage by the appropriate Local Transport Facility rate. The amount to be billed shall be the product of this calculation (i.e., the number of miles multiplied by the per mile rate) multiplied by the number of access minutes.

Issued:	. 2009	Effective Date:	. 2009

#### SECTION 3. SERVICE AND RATE DESCRIPTIONS

#### 3.1 Access Services

Switched Access Service, which is available to Customers for their use in furnishing their services to end users, provides a two-point communications path between a Customer's premises (or a collocated interconnection location) and an end user's premises. It provides for the use of common terminating, switching and trunking facilities. Switched Access Service provides for the ability to originate calls from an end user's premises to a Customer's premises (or a collocated interconnection location), and to terminate calls from a Customer's premises (or a collocated interconnection location) to an end user's premises in the LATA where it is provided. Switched Access Service must be ordered separately for each LATA in which the customer desires to originate or terminate calls.

Switched Access Service is provided in the following service categories, which are differentiated by their technical characteristics and the manner in which an end user or Customer accesses them when originating or terminating calls.

FGD Access, which is available to all Customers, provides trunk side access to Company end office switches with an associated uniform 10XXX or 101XXXX access code for the Customer's use in originating and terminating communications. End users may also originate calls to a selected FGD Access Customer by dialing 1+NPA-NXX-XXXX when using the Company's presubscription service.

Toll Free Data Base Access Service, which is available to all Customers, provides trunk side access to Company end office switches in the originating direction only, for the Customer's use in originating calls dialed by an end user to telephone numbers beginning with the prefix "800", "866", "888", or "877", for example. Toll Free Data Base Access Service is offered in conjunction with Feature Group D Access.

Transit Traffic Service is an access service in which the Company transits traffic originated by a third party who is not an End User or other user of the Company's local exchange or exchange access service through its wire centers to a Customer. Transit Traffic Service is comprised of various facilities, connections, features and functions. It provides for the use of common terminating, common switching and switched transport facilities of the Company but does not include local switching.

Issued:	, 2009	Effective Date:	. 2009

## 31 Access Services (Cont'd.)

#### 3.1.1 Access Service Order

An Access Service Order is used by the Company to provide a Service. When placing an order for Access Service, the Customer shall provide, at a minimum, the following information:

For Feature Group D Switched Access Service, the Customer shall specify the number of busy hour minutes of capacity (BHMC) from the Customer's premises to the end office by traffic type. This information is used to determine the number of transmission paths. The Customer shall also specify the Local Transport and Local Switching options. When FGD is ordered by specifying the number of trunks and direct routing to an end office is desired, the customer shall specify:

the end office and the Local Transport and Local Switching options desired.

When FGD is ordered by specifying the number of trunks and end office routing via an access tandem operated by another Exchange Telephone Company is desired, the customer shall specify:

the access tandem.

the Local Transport and Local Switching options desired, and an estimate of the amount of traffic to be generated to and/or from each Company end office subtending another Exchange Telephone Company's access tandem.

In addition, for Feature Group D with the SS7 signaling option, the Customer shall specify the switching point codes and trunk circuit identification codes for trunks with the SS7 signaling option, and the STP point codes, signaling link codes and link type for each Common Channel Signaling Access (CCSA) connection ordered. When a Customer orders FGD in trunks, the Customer is responsible to assure that sufficient access facilities have been ordered to handle its traffic.

Issued:	, 2009	Effective Date:	, 2009

#### 3.1 Access Services (Cont'd.)

#### 3.1.1 Access Service Order (Cont'd.)

For 800 Data Base Access Service, the Customer shall order the service in accordance with the preceding provisions set forth for Feature Group D. If the Customer desires any of the optional features available with 800 Data Base Service, the Customer shall so specify on the order for service.

#### 3.1.1.1 Access Order Service Date Intervals

Access Service is provided with one of the following Service Date Intervals:

- -Standard Interval
- -Negotiated Interval
- -Advance Order Interval

#### (A) Standard Interval

A schedule of Standard Intervals applicable for Switched Access Services and is as follows:

Trunk Groups Standard Interval

1 to 4 Trunks 28 Days
5 to 24 Trunks 30 Days

#### (B) Negotiated Interval

The Company will negotiate a service date interval with the Customer when:

- (1) There is no Standard Interval for the service, or;
- (2) The quantity of Access Services orders exceeds the quantities specified in the Standard Intervals, or;

т 1	2000	Ecc 4: D 4	2000
Issued:	2009	Effective Date:	. 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.1 Access Order Service Date Intervals (Cont'd.)
      - (B) Negotiated Interval (Cont'd.)
        - (3) The Customer requests a service date beyond the applicable Standard Interval service date except as set forth in (C) following.

The Company will offer a service date based on the type and quantity of Access Services the Customer has requested. The Negotiated Interval may not exceed by more than six months the Standard Interval service date, or, when there is no Standard Interval, the Company offered service date. All services for which rates are applied on an individual case basis are provided with a Negotiated Interval.

Common Channel Signaling Access (CCSA) links will be provided on a Negotiated Interval. New or existing FGD trunks ordered with the SS7 signaling option will be provided on a Negotiated Interval.

Issued:	, 2009	Effective Date:	, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.1 Access Order Service Date Intervals (Cont'd.)
      - (B) Negotiated Interval (Cont'd.)
        - (3) (Cont'd.)

The addition and/or deletion of a Toll Free Access Service six digit customer identification NXX is provided with a Negotiated Interval. The addition of a Toll Free Access Service ten digit customer identification record to the Toll Free Access Service data base or the deletion of a Toll Free Access Service ten digit customer identification record from the Toll Free Access Service data base is provided with a Negotiated Interval.

Maximum Interval	
Initial Establishment of service where Customer	
is:	
- Not yet provided with any Trunk Group service	
in the LATA	6 Months
- Provided Trunk Group service in the LATA	90 Days

Issued:	2009	Effective Date:	2009

- 31 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.1 Access Order Service Date Intervals (Cont'd.)
      - (C) Advance Order Interval

When placing an Access Order, a Customer may request an Advance Order Interval for a service date of 12 to 24 months from the Application Date for the following services:

- A minimum of 24 voice grade equivalent Switched Access Service lines or trunks or 720 BHMCs

Orders for less than the minimum quantities will be accommodated under Standard or Negotiated Interval provisions. Advance Order Interval Access Orders are subject to all ordering conditions of Standard and Negotiated Interval Access Orders except for the following:

(1) Advance Payment

A nonrefundable Advance Payment will be calculated as follows:

# Advance Payment (Nonrefundable)

The minimum monthly charge for the minimum period plus the applicable Nonrecurring Charges for the services ordered.

This Advance Payment is due ten (10) working days from the date the Company confirms acceptance of the order, or on the Application Date, whichever date is the later date. If the Advance Payment is not received by such payment date, the order will be canceled.

Issued:	, 2009	Effective Date:	, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.1 Access Order Service Date Intervals (Cont'd.)
      - (C) Advance Order Interval (Cont'd.)
        - (1) Advance Payment (Cont'd.)

When the Access Services are connected on the service date, the Advance Payment will be applied, as a credit, to the Customer's billed service charges. When there has been a decrease in the number of services originally ordered, as set forth in (2) following, only the portion of the Advance Payment for services actually installed will be credited.

(2) Cancellation or Partial Cancellation of an Advance Order Interval Access Order

When the Customer cancels an Access Order, the order will be withdrawn. The Advance Payment will not be credited or refunded.

Any decrease in the number of ordered Access Services will be treated as a partial cancellation, and the portion of the Advance Payment for the services canceled will not be credited or refunded.

Issued:	, 2009	Effective Date:	, 2009

## 31 Access Services (Cont'd.)

#### 3.1.1 Access Service Order (Cont'd.)

#### 3.1.1.2 Access Order Modifications

The Customer may request a modification of its Access Order at any time prior to notification by the Company that service is available for the Customer's use or prior to the service date, whichever is later.

Any increase in the number of Switched Access Service lines, trunks or busy hour minutes of capacity or CCSA signaling connections will be treated as a new Access Order (for the increased amount only).

#### (A) Service Date Change Charge

Access Order service dates for the installation of new services or rearrangements of existing services may be changed, but the new service date may not exceed the original service date by more than thirty (30) calendar days. When, for any reason, the Customer indicates that service cannot be accepted for a period not to exceed thirty (30) calendar days, and the Company accordingly delays the start of service, a Service Date Change Charge will apply. If the Customer requested service date is more than thirty (30) calendar days after the original service date, the order will be canceled by the Company and reissued with the appropriate cancellation charges applied.

Issued:	, 2009	Effective Date:	2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.2 Access Order Modifications (Cont'd.)
      - (A) Service Date Change Charge (Cont'd.)

A Service Date Change Charge will apply, on a per order per occurrence basis, for each service date changed. The applicable charge is found in Section 5.1.1.

(B) Partial Cancellation Charge

Any decrease in the number of ordered Switched Access Service lines, trunks or busy hour minutes of capacity ordered with a Standard or Negotiated Interval Access Order will be treated as a partial cancellation and the charges as set forth in Section 3.1.1.3 following will apply. Partial cancellation charges do not apply to Advance Order Interval Access Orders.

Issued:	, 2009	Effective Date:	, 2009

- 31 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.2 Access Order Modifications (Cont'd.)
      - (C) Design Change Charge

The Customer may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. Design changes do not include a change of customer premises, end user premises, end office switch, Feature Group type except for changes to Feature Group D. Changes of this nature will require the issuance of a new order and the cancellation of the original order with appropriate cancellation charges applied.

The Design Change Charge will apply on a per order per occurrence basis, for each order requiring a design change. The applicable charge is found in Section 5.1.1.

If a change of service date is required, the Service Date Change Charge will also apply.

#### DSO Order Expedite Charge

The Company will impose an expedite charge on any order for circuits requested by the customer to be installed earlier than the standard interval for DSO orders. A charge of twenty-five dollars per DSO expedited, regardless of service address, will be assessed to the customer.

Issued:	, 2009	Effective Date:	, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.2 Access Order Modifications (Cont'd.)
      - (D) Expedited Order Charge

When placing an Access Order for service(s) for which Standard Intervals exist, a Customer may request a service date that is prior to the Standard Interval service date. A Customer may also request an earlier service date on a pending Standard, Negotiated or Advance Order Interval Access Order. If the Company agrees to provide service on an expedited basis, an Expedited Order Charge will apply.

If the Company receives a request for an expedited service date at the time a Standard Interval Access Order is placed, the Expedited Order Charge is calculated by summing all the nonrecurring charges associated with the order and then dividing this total by the number of days in the Standard Interval. The charge is then applied on a per day of improvement basis, per order, but in no event shall the charge exceed fifty percent of the total nonrecurring charges associated with the Access Order.

Issued:	, 2009	Effective Date:	, 2009

- 31 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.3 Cancellation of an Access Order
      - (A) A Customer may cancel an Access Order for the installation of service at any time prior to notification by the Company that services available for the Customer's use or prior to the service date, whichever is later. The cancellation date is the date the Company receives written or verbal notice from the Customer that the order is to be canceled. The verbal notice must be followed by written confirmation within ten (10) days. If a Customer or a Customer's end user is unable to accept Access Service within thirty (30) calendar days after the original service date, the Customer has the choice of the following options:
        - The Access Order shall be canceled and charges set forth in (B) following will apply, or

Billing for the service will commence.

If no cancellation request is received within the specified thirty (30) calendar days, billing for the service will commence. In any event, the cancellation date or the date billing is to commence, as applicable, shall be the 31st day beyond the original service date of the Access Order.

Issued: , 2009 Effective Date: \_\_\_\_\_, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.3 Cancellation of an Access Order (Cont'd.)
      - (B) When a Customer cancels a Standard or Negotiated Interval Access Order for the installation of service, a Cancellation Charge will apply as follows:
        - (1) When the Customer cancels an Access Order, a charge equal to the estimated provisioning costs incurred at a particular date for the service ordered by the Company shall apply.
        - (2) If the Company misses a service date for a Standard or Negotiated Interval Access Order by more than thirty (30) days, due to circumstances such as acts of God, governmental requirements, work stoppages and civil commotions, the Customer may cancel the Access Order without incurring cancellation charges.

Issued.	2009	Effective Date:	2000

## 31 Access Services (Cont'd.)

#### 3.1.1 Access Service Order (Cont'd.)

#### 3.1.1.4 Minimum Period

- (A) The minimum period for which Access Service is provided and for which charges are applicable, is one month.
- (B) The following changes will be treated as a discontinuance of the existing service and an installation of a new service. All associated nonrecurring charges will apply for the new service.

The changes listed below are those which will be treated as a discontinuance and installation of service and for which a new minimum period will be established.

- (1) A move to a different building.
- (2) A change in type of service.
- (3) A change in Switched Access Service Interface Group.
- (4) A change in Switched Access Service traffic type.
- (5) A change in STP Access link.
- (6) A change in STP Port.
- (7) A change in Company-provided Switched Access Service to a Collocated Interconnection arrangement or vice versa.
- (8) A change to an existing Feature Group D Service to include the provision of 64 kbps Clear Channel Capability.

Issued:	, 2009	Effective Date:	. 2009

#### 3.1 Access Services (Cont'd.)

#### 3.1.1 Access Service Order (Cont'd.)

#### 3.1.1.5 Minimum Period Charges

When Access Service is disconnected prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period.

The Minimum Period Charge for monthly billed services will be determined as follows:

For Switched Access Service, the charge for a month or fraction thereof is equal to the applicable minimum monthly charge for the capacity.

All applicable nonrecurring charges for the service will be billed in addition to the Minimum Period Charge.

## 3.1.1.6 Nonrecurring Charges

Nonrecurring charges are one-time charges that apply for a specific work activity (i.e., installation or change to an existing service). Types of nonrecurring charges that apply for Switched Access Service are: installation of service and service rearrangements.

#### (1) Installation of Service

Nonrecurring charges apply to each Switched Access Service installed. For Switched Services ordered on a per trunk basis, the charge is applied per trunk or out of band signaling connection. For Switched Services ordered on a busy hour

Issued:	. 2009	Effective Date:	. 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.6 Nonrecurring Charges (Cont'd.)
      - (1) Installation of Service (Cont'd.)

minutes of capacity basis, the charge is also applied on a per trunk basis but the charge applies only when the capacity ordered requires the installation of an additional trunk(s). In addition, nonrecurring charges apply when an out of band signaling connection is installed for use with FGD.

(2) Service Rearrangements

All changes to existing services other than changes involving administrative activities only will be treated as a discontinuance of the existing service and an installation of a new service. The nonrecurring charge described in (1) preceding will apply for this work activity. Moves that change the physical location of the point of termination are described below.

(a) Moves Within the Same Building

When the move is to a new location within the same building, the charge for the move will be an amount equal to one half of the nonrecurring charge for the capacity affected. There will be no change in the minimum period requirements.

Issued: , 2009 Effective Date: , 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.1 Access Service Order (Cont'd.)
    - 3.1.1.6 Nonrecurring Charges (Cont'd.)
      - (2) Service Rearrangements (Cont'd.)
        - (b) Moves to a Different Building

Moves to a different building will be treated as a discontinuance and start of service and all associated nonrecurring charges will apply. New minimum period requirements will be established for the new service. The Customer will also remain responsible for satisfying all outstanding minimum period charges for the discontinued service.

## 3.1.1.7 Network Blocking Charge

The Customer will be notified by the Company to increase its capability (busy hour minutes of capacity or quantities of trunks) when excessive trunk group blocking occurs on groups carrying Feature Group D traffic and the measured access minutes for that hour exceed the capacity purchased.

If the order for additional capacity has not been received by the Company within fifteen (15) days of the notification, the Company will bill the Customer, at the rate set forth in 5.1.3(C) following, for each overflow in excess of ordered capacity.

Issued: ,2009 Effective Date: ,2009

3.1	Access Services	(Cont'd)	۱
J. 1		Com a.	,

### 3.1.2 Standard Rate Categories

The following rate categories apply to all forms of Switched except as stated in 3.1.3:

- Carrier Common Line
- Local Transport
- End Office

#### (A) Carrier Common Line

The Carrier Common Line rate category provides for the use of Company common lines by Customers for access to end users to furnish Customer intrastate communications. Carrier Common Line is provided where the Customer obtains Company provided Switched Access Service.

- (1) Limitations
  - (a) A telephone number is not provided with Carrier Common Line.
  - (b) Detail billing is not provided for Carrier Common Line.
  - (c) Directory listings are not included in the rates and charges for Carrier Common Line.

Issued:	, 2009	Effective Date:	, 2009

- 31 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (A) Carrier Common Line (Cont'd.)
      - (1) Limitations (Cont'd.)
        - (d) Intercept arrangements are not included in the rates and charges for Carrier Common Line.
        - (e) All trunk side connections provided in the same combined access group will be limited to the same features and operating characteristics.
      - (2) Undertaking of the Telephone Company

Where the Customer is provided with Switched Access Service under this tariff, the Company will provide the use of Company common lines by a Customer for access to end users at rates and charges as set forth in Section 5.1.2 following.

- (3) Obligations of the Customer
  - (a) The Customer facilities at the premises of the ordering Customer shall provide the necessary on-hook and off-hook supervision.
  - (b) All Switched Access Service provided to the Customer will be subject to Carrier Common Line charges, excluding the Common Channel Signaling Access exemption.

Issued:	, 2009	Effective Date:,	2009

- 3.1 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (A) Carrier Common Line (Cont'd.)
      - (4) Common Channel Signaling Access Exemption

The Common Channel Signaling Access Signal Transfer Point (STP) Port Termination charge, as set forth in Section 5.1.3 following, is not subject to a Carrier Common Line charge.

- (5) Rate Regulations
  - (a) The Carrier Common Line charges will be billed per access minute to each Switched Access Service Customer.
  - (b) When the Customer reports interstate and intrastate use of Switched Access Service, the Carrier Common Line charges will be billed only to intrastate interLATA and/or intraLATA Switched Access Service access minutes based on the data reported by the Customer set forth in Section 2.3.3 preceding.

Issued:	, 2009	Effective Date:	, 2009

- 31 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (B) Local Transport

The Local Transport rate category provides for transmission facilities between the Customer's premises or collocated interconnection location and the Company's end office switch(es) where the Customer's traffic is switched to originate or terminate its communications.

Except as stated in the following paragraph, Local Transport service is provided in conjunction with Qwest Communications Company. Charges for Local Transport service are computed in accordance with Section 2.6.5 preceding (Ordering, Rating, and Billing of Access Services Where More Than One Exchange Telephone Company Is Involved). For purposes of determining Local Transport Mileage, distance will be measured from the wire center that normally serves the Customer's premises to the end office switch(es).

Issued:	. 2009	Effective Date:	. 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (B) Local Transport (Cont'd.)

The Company will provide Direct Trunked Transport between a Customer's premises and the Company's end office switch(es) upon request. At the Customer's option, Entrance Facility, Direct Trunked Transport, and STP Link Transport, may be provided by the Company, by the Customer, or by another carrier. If Direct Trunk Transport facilities are terminated at a Company end office switch location by a Customer, or by another carrier on behalf of a Customer, Entrance Facility Termination charges as described in 3.1.2(B)(1) will apply, but no other Local Transport elements will be charged. Where Common Channel Signaling Access is ordered by a Customer that uses Direct Trunk Transport facilities, the STP Port charge as described in 3.1.2(B)(7) will also apply.

The following paragraphs describe the Local Transport rate elements.

Local Transport - Entrance Facility;
Common Switched Transport;
Transport Multiplexing;
Direct Trunked Transport;
Interconnection Charge;

Issued: , 2009 Effective Date: , 2009

- 3.1 Access Service (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (B) Local Transport (Cont'd.)
      - (1) Local Transport-Entrance Facility

A Local Transport-Entrance Facility provides the communication path between a Customer's premises and the Company serving wire center of that premises for the sole use of the Customer. The Local Transport-Entrance Facility category is comprised of a DS1 rate. A Local Transport-Entrance Facility is required whether the Customer's premises and the serving wire center are located in the same or different buildings.

- (2) Local Transport-Common Switched Transport
  - (a) The Local Transport Termination rate provides for that portion of the voice frequency transmission path at the end office and at the Customer's premises.
  - (b) The Local Transport Facility rate provides for that portion of the voice frequency transmission path between the end office and at the Customer's premises.
- (<sup>3</sup>) Transport Multiplexing

The Local Transport multiplexing feature allows for a DS3 facility to be channelized into 28 DS1 services or for a DS1 facility to be channelized into 24 Voice Grade or Voice Grade equivalent services. Multiplexing is available at the serving wire center of the customer premises or a collocation location, or at the company's premises. The multiplexing rates and charges are set forth in Section 5.1.4.

Issued:	, 2009	Effective Date:	, 2009
	, —		_ ,

Julie Barghouthi
Sr. Vice President Product Development & Access Management
IntelePeer, Inc.
2855 Campus Drive, Suite 200
San Mateo, CA 94403

IntelePeer, Inc. A/73169185.1

- 31 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (B) Local Transport (Cont'd.)
      - (4) Local Transport- Direct Trunked Transport

The Local Transport-Direct Trunked Transport provides the transmission path from the serving wire center of the Customer's premises to an end office or as an option from the serving wire center to a tandem. This transmission path is dedicated to the use of a single Customer.

The Local Transport-Direct Trunked Transport rate category is comprised of a monthly fixed rate and a monthly per mile rate based on the facility provided. The fixed rate provides the circuit equipment at the ends of the transmission links. The per mile rate provides the transmission facilities, including intermediate transmission circuit equipment, between the end points of the circuit. The Local Transport-Direct Trunked Transport rate is the sum of the fixed rate and the per mile rate. For purposes of determining the per mile rate, mileage shall be measured as airline mileage between the serving wire center of the Customer's premises and the end office or directly to the access tandem using the V&H coordinates method.

(5) Local Transport-Interconnection Charge

The Local Transport-Interconnection Charge provides for interconnection with the Company's Switched Access network. This rate element will be applied to all Switched Access minutes of use (except Local Exchange Access Service) that originate or terminate at a Company end office.

Issued:	. 2009	Effective Date:	. 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (B) Local Transport (Cont'd.)
      - (6) Local-Transport Facilities

DS 1 facilities are available for Local Transport-Entrance Facilities and for Local Transport-Direct Trunked Transport facilities. A DS1 facility is capable of transmitting electrical signals at a nominal 1.544 Mbps, with the capability to channelize up to 24 voice-frequency transmission paths.

(7) Common Channel Signaling Access

Common Channel Signaling Access (CCSA) is comprised of a STP Port Termination rate and a STP Link Transport rate.

The STP Port Termination rate provides for the point of termination to the signal switching capability of the STP.

The STP Link Transport rate provides for the transmission facilities between the serving wire center of the customer designated premises and the Company STP.

Issued:	. 2009	Effective Date:	, 2009

Access Services (Cont'd.)

- 3.1.2 Standard Rate Categories (Cont'd.)
  - (B) Local Transport (Cont'd.)
    - (8) Interface Groups

The Interface Group is provided for terminating the Local Transport at the Customer's premises. The Interface Group provides a specified premises Interface. Where transmission facilities permit, the individual transmission path between the Customer's premises and the first point of switching may at the option of the Customer be provided with optional features.

Interface Group 1 provides DS 1 level digital transmission at the point of termination at the Customer's premises. The interface is capable of transmitting electrical signals at a nominal 1.544 Mbps, with the capability to channelize up to 24 voice frequency transmission paths.

Issued: ,2009 Effective Date: ,2009

- 3.1 Access Services (Cont'd.)
  - 3.1.2 Standard Rate Categories (Cont'd.)
    - (C) End Office

The End Office rate category provides the local end office switching and end user termination functions necessary to complete the transmission of Switched Access communications to and from the end users served by the Company's end office. The End Office rate category consists of the Local Switching rate element.

The Local Switching rate element provides for: a) the use of end office switching equipment; b) the terminations for the end user common lines terminating in the local end office; and c) the termination of a call at a Company Intercept operator or recording. The operator or recording tells a caller why a call, as dialed, could not be completed, and if possible, provides the correct number. Intercept rates are assessed to a Customer based on the total number of access minutes.

Automatic Number Identification (ANI) provides the automatic transmission of a seven or ten digit number and information digits to the Customer's premises for calls originating in the LATA, to identify the calling telephone number. The ANI feature is an end office software function which is associated on a call-by-call basis with (1) all individual transmission path in a trunk group routed directly between an end office and a Customer's premises or, where technically feasible, with (2) all individual transmission paths in a trunk group between an end office and an access tandem, and a trunk group between an access tandem and a Customer's premises. The Shared End Office Trunk Port provides for the termination of Tandem Switched Transport to an end office.

Access minutes for all Switched Access Service subject to the Shared End Office Trunk Port will be multiplied by the per minute rate set forth in Section 5.1.4.

Issued: , 2009 Effective Date:	, 2009

#### 3.1 Access Services (Cont'd.)

#### 3.1.3 Other Rate Categories

#### (A) Toll Free Data Base Access Service

Toll Free Data Base Access Service is a service offering utilizing originating trunk side Switched Access Service. The service provides for the forwarding of end user dialed toll free calls to a Company Service Switching Point which will initiate a query to the database to perform the Customer identification and delivery function. The call is forwarded to the appropriate Customer based on the dialed toll free number. Any dial around compensation relating to pay telephones will be billed in accordance to procedures and rates proscribed by the Federal Communications Commission. The Company reserves the right to bill end users of its toll free service for any dial around compensation costs the Company may incur.

## (1) Customer Identification Charge

The Toll Fee Data Base Access Service Customer Identification applies for the identification of the appropriate Customer. The charge is assessed to the Customer on a per query basis and may include an area of service which may range from a single NPA/NXX to an area consisting of all LATAs and NPAs in the State of Utah.

#### (2) Toll Free Number Reservation

The Toll Free Number Reservation service applies to the request of the Customer to have the Company attempt to reserve a specific toll free number for the Customer. The Company will not guarantee that a specific toll free number in any of the toll free prefixes (800, 877, 888, 866 or future prefixes as designated by NANPA) will be available at the time a Customer requests the specific number.

Issued:	. 2009	Effective Date:	. 2009

3.1 Access Services (Cont'd.)

#### 3.1.3 Other Rate Categories

#### (B) Local Exchange Service

#### 1) Definition

As used in this section, "local exchange call" means a telephonic communication (a) that is originated by a company that is authorized by the appropriate regulatory agency to provide local exchange telephone service, (b) that originates and terminates within a single "exchange area" or "local calling area" as defined in the approved tariffs of the originating company, and (c) that is dialed to an NPA-NXX code directly assigned to the Company in the State of Utah.

### 2) Description

Local Exchange Service is a service offering providing trunk side access to the Company's end office switches in the terminating direction only, for use by authorized providers of local exchange telephone service for the completion of local exchange calls that originate in the State of Utah and terminate to the Company's end users in the State of Utah. Local Exchange Service must be provided to a Point of Interface (POI) which will be established jointly by the Company and the customer at a location within the State of Utah and in the same LATA as the Company end office at which the local exchange call will terminate. Local Exchange Service provides a trans- mission path between the POI and the Company's end user.

Issued:	. 2009	Effective Date:	2009

- 3.1 Access Services (Cont'd.)
  - 3.1.3 Other Rate Categories (Cont'd.)
    - (B) Local Exchange Service (Cont'd.)
      - 3) Obligations of the Local Exchange Provider
        - (a) The Company may request an annual audit of the authorized local exchange provider billings for Local Exchange Services (calls originating from the Company's end-user to the authorized local exchange provider's end-user). The audit requirement is needed to ensure accurate billing between local exchange calls and non-local exchange calls.
        - (b) The authorized local exchange provider will be requested to provide a forecast of total usage by each trunk group or facility ordered from the Company for each POI used in a Local Exchange Service arrangement.
      - 4) Rating of Local Exchange Service

For billing purposes, Local Exchange Service calls originating from an authorized local exchange provider and terminating on the Company's network (for completion to a Company end user) will be rated at the Company's end office.

5) Rate Regulations

Local Exchange Service will consist of the following rate categories.

Issued: , 2009 Effective Date: \_\_\_\_\_\_, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.3 Other Rate Categories (Cont'd.)
    - (B) Local Exchange Service (Cont'd.)
      - 5) Rate Regulations (Cont'd.)
        - (a) Terminating Usage

The Terminating Usage rate will be applied on a per minuteof-use basis, as set forth in 5.1.3 following, for the completion of calls from a authorized local exchange provider end-user to a Company end-user.

(b) Direct Trunked Transport

Direct Trunked Transport provides the communication path between an authorized local exchange provider's POI and the Company's end office for the sole use of the authorized local exchange provider. The Direct Trunked Transport rates are billed on a monthly recurring and a per mile basis as specified in Section 5.1.3(B) of this tariff. A Customer may use a single Direct Trunk in conjunction with Local Exchange Service and other Switched Access Services. The Company shall require the Customer to file a Percentage of Local Usage (PLU) report which should, upon ordering Direct Trunked Transport from the Company, be revised on a quarterly basis.

Issued: , 2009 Effective Date: \_\_\_\_\_, 2009

- 3.1 Access Services (Cont'd.)
  - 3.1.3 Other Rate Categories (Cont'd.)
    - **(B)** Local Exchange Service (Cont'd.)
      - 5) Rate Regulations (Cont'd.)
        - (c) Service Orders

The Service Order charges as specified in Section 5.1.1 of this tariff will apply on a per order basis.

(d) Other Standard Charges

Standard charges set forth in 3.1.2 do not apply to Local Service, except for the specific rate elements identified in (b) and (c) above.

Issued:	, 2009	Effective Date:	, 2009

#### Julie Barghouthi

#### 32 Miscellaneous Services

#### 3.2.1 Presubscription

- (A) Presubscription is an arrangement whereby an end user designate to the Company an interexchange carrier (IC) to access, without an access code, for intrastate interLATA calls and interstate interLATA calls subject to the Company's FCC Access Tariff. This IC is referred to as the end user's Primary Interexchange Carrier (PIC). The end user may select as its PIC the Company, or any other IC that orders originating Feature Group D Switched Access Service at the end office that serves the end user. After the end user's initial selection of a pre-designated IC, for any additional change in selection, a non-recurring charge, as set forth in Section 5.3.1, applies.
- (B) At the request of a new or existing end user served by a Feature Group D end office, the Company will provide a list of ICs the end user may select as its PIC. At no additional charge for the initial selection, the customer may choose either of the following options.

Designate an IC as a PIC and dial 10XXX or 101XXXX to reach other ICs.

Designate that they do not want to be presubscribed to any IC and choose to dial 10XXX or 101XXXX for all calls to all ICs.

New end users subscribing to the Company's Local Exchange Service that do not specify a PIC will default to the Company as their initial PIC selection. Subsequent to the installation of Local Exchange Service, and after the end user's initial selection of a PIC, for any additional change in selection, a nonrecurring charge as set forth in Section 5.3.1, applies. This charge is billed to the end user that is the subscriber to the Local Exchange Service and applies only for selection of an IC which provides only intrastate service.

Issued: , 2009 Effective Date:\_\_\_\_\_\_, 2009

- 3.2 Miscellaneous Services (Cont'd.)
  - 3.2.2 Number Portability
    - (A) General

Number Portability is a service arrangement provided by to subscribers of the Company's Local Exchange Access Service, as set forth in 3.1.3(B) preceding. Number Portability allows an End User who switches from the Company's local exchange service to that of another certified local exchange carrier to retain the use of their existing Company assigned telephone number, provided the customer remains at the same location.

Number Portability utilizes a telephone number and electronic switching facilities to automatically forward all incoming calls to the Company assigned telephone number to the terminating telephone number assigned by another certified local exchange carrier.

Number Portability provides a single call path for the forwarding of no more than one simultaneous call to the forwarding call number. Additional call paths for the forwarding of multiple simultaneous calls are available on a per path basis at an additional charge.

All other access, local and toll rates and charges for all services ordered by the Customer, as set forth in other sections of this tariff and the Company's local exchange and federal access tariffs, continue to apply.

Issued:	. 2009	Effective Date:	, 2009

- 32 Miscellaneous Services (Cont'd.)
  - 3.2.2 Number Portability (Cont'd.)
    - (B) Regulations
      - 1) Number Portability is offered where facilities permit.
      - The Customer is solely responsible for obtaining authorization from the end user for the handling of the disconnection of the end user's service with the Company, the provision of service by the Customer, and the provision of Number Portability. The Company, at its discretion, may require the Customer to provide written evidence of its authority to act on behalf of the end-user.
      - 3) The Customer is required to provide sufficient terminating facilities and service at the terminating end of a forwarded call to adequately handle all traffic to that location and is solely responsible to ensure that its facilities, equipment and services do not interfere with or impair any facility, equipment or service of the Company or any of its end users.
      - 4) End to end transmissions characteristics may vary depending on the distance and routing necessary to complete calls over facilities and the fact that another carrier is involved in provisioning of service.

Issued:	, 2009	Effective Date:	, 2009

- 3.2 Miscellaneous Services (Cont'd.)
  - 3.2.2 Number Portability (Cont'd.)
    - (C) Rate Regulations

Number Portability will consist of the following rate categories.

1) Service Orders

Service Order charges as specified in Section 5.1.1 of this tariff may apply on a per order basis.

Issued: , 2009 Effective Date: \_\_\_\_\_\_, 2009

#### **SECTION 4. BILLING AND COLLECTION SERVICES**

#### 41 General

The Company will provide the following services:

Recording Service Automatic Number Identification (ANI) Billing Name and Address (BNA)

#### 4.2 Recording Service

Recording is the entering on magnetic tape or other acceptable media the details of Customer messages originated through Switched Access Service. Recording is provided 24 hours a day, 7 days a week.

The Company will provide Recording Service in association with the offering of Feature Groups B for 900 Access Service and D Switched Access Service for Customer messages that can be recorded by Company provided automatic message accounting equipment. At the request of the Customer, Recording Service will be provided for Feature Group D Switched Access Service on an end office and type of call basis. Type of call means message telecommunications service (MTS) including 700 and 900 Service, calls originating and/or terminating over a WATS access line, and station message detail recording for MTS and calls originating from a WATS access line.

The Company will provide Recording Service in its operating territory. The minimum territory for which the Company will provide Recording Service is all the appropriately equipped offices in a state operating territory for which the Customer has ordered Feature Group B for 900 service or D Switched Access Service. A state operating territory of particular telephone company includes all its LATAs or market areas which are located in the same state including the areas in contiguous states which are assigned to such LATAs or market areas and served by the same Company.

Issued:	, 2009	Effective Date:	, 2009

#### 4.2 Recording Service (Cont'd.)

For Feature Group B for 900 Access Service and FG D.Switched Access Service the term "used herein denotes an intrastate call originated by a Customer's end "customer message"

user. Station message detail recording is an optional feature which provides a record of customer messages originated by MTS and WATS access lines. Such detail will be provided as part of Feature Group D end office and type of call Recording Service when requested by the Customer.

### 4.2.1 Undertaking of the Company

- (A) The Company will record all customer messages carried over Feature Group B for 900 Access Service and FG D Switched Access Service that are available to Company provided recording equipment of operators. Unavailable customer service messages will not be recorded. The recording equipment will be provided at locations selected by the Company.
- (B) A standard format for the provision of the recorded customer message detail will be established by the Company and provided to the Customer. If, in the course of Company business, it is necessary to change the format, the Company will notify the involved Customers six months prior to the change. Assembly and Editing, Provision of Customer Detail, Data transmission to a Customer location, special orders for recording and program development will be provided to the Customer on a contractual basis.
- (C) Recorded customer message detail which is used at the request of the Customer to provide Message Processing and Message Bill Processing Service is not retained by the Company for longer than forty-five (45) days. The rated but unbilled message detail and the billed message detail will be retained for reference in place of the recorded customer message detail

Issued:	. 2009	Effective Date:	. 2009

- 4.2 Recording Service (Cont'd.)
  - 4.2.1 Undertaking of the Company (Cont'd.)
    - (C) (Cont'd.)

For recorded customer message detail not used by Message Processing Service at the Customer's request, the Company will make every reasonable effort to recover recorded Customer message detail previously made available to the Customer and make it available again for the Customer. The charges as set forth in 5.4.1 following will apply for all such detail provided. Such a request must be made within thirty (30) days from the date the details were initially made available to the Customer.

### 4.2.2 Liability of the Company

Notwithstanding 4.2.1 preceding, the Company liability for Recording Service is as follows:

Unless there is an expressed written agreement to the contrary, in the absence of gross negligence or willful misconduct, no liability for damages to the Customer or other person or entity other than as set for in (A) and (B) preceding shall attach to the Company for its action or the conduct of its employees in providing Recording Service.

### 4.2.3 Obligations of the Customer

(A) The Customer shall order Recording Service under a Special Order. The Customer shall order Recording Service at least one month prior to the date then the Customer message detail is to be recorded, unless Customer's request requires that Recording Service be provided by end office and type of call, then the ordering interval will be determined on an individual case basis.

Issued:	, 2009	Effective Date:	_, 2009

## 4.2 Recording Service (Cont'd.)

#### 4.2.4 Payment Arrangements and Audit Provision

#### (A) Notice and Scope

The Customer shall order Recording Service for Feature Group D Switched Access by end office and type of call in accordance with the terms and conditions established on an individual case basis Special Order.

- (1) Upon forty-five (45) days' prior written notice by the Customer to the Company (or such shorter period as the parties may mutually agree upon), the Customer or its authorized representative shall have the right to commence an audit during normal business hours and at intervals of no more that one audit in any six month period. The audit will be limited to all such records and accounts as may, under recognized accounting practices, contain information bearing upon amounts subject to being billed to the Customer's end users by the Company as part of its provision of Billing and Collection Services and the changes to the Customer for other services provided by the Company pursuant to this tariff.
- (2) The written notice of audit shall identify the date upon which it is to commence, the location, the Customer's representatives, the subject mater of the audit, and the materials to be reviewed.
- (3) The written notice of audit shall be directed to the Company's representative at the address stipulated by such representative.

Issued:	. 2009	Effective Date:	. 2009

- 4.2 Recording Service (Cont'd.)
  - 4.2.4 Payment Arrangements and Audit Provision (Cont'd.)
    - (A) Notice and Scope (Cont'd.)
      - (4) The Company may, within thirty (30) days of receipt of the Customer's notice of audit, postpone commencement by written notice for a period not to exceed fifteen (15) days, but only for good cause. The Company shall also indicate the new date for commencement of said audit.
      - (5) Upon completion of the audit, the Customer's auditors are to provide an oral report of their findings to the Company prior to their departure, followed by a letter within thirty (30) days confirming findings and postponed completion.
    - (B) Payment of Expense

Each party shall bear its own expenses in connection with the conduct of an audit. Special data extractions required by the Customer for its representative to conduct the audit will be paid for by the Customer. "Special data extraction" for auditing purposes shall mean programming, clerical and computer time required to create an output record (from existing data files) that cannot normally be created from current software programs in the production program library.

- (C) Requests for Examinations
  - (1) In addition to audits, the Customer, or its representatives, may request, from time to time, the opportunity to conduct an examination, as defined in (2) following. The Company will make reasonable efforts to accommodate requests for examination and to cooperate in the conduct of an examination.

Issued:	, 2009	Effective Date:	, 2009

- 4.2 Recording Service (Cont'd.)
  - 4.2.4 Payment Arrangements and Audit Provision (Cont'd.)
    - (C) Requests for Examinations (Cont'd.)
      - (2) An "Examination" shall, for purposes of this section, constitute a reasonable inquiry on a single issue or a specific topic related to Billing and Collection Service for a stated reason.

Upon concurrence by both parties that errors or omissions exist, adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit.

(D) Audit Provision

All information received or reviewed by the Customer or its authorized representative is to be considered confidential and is not to be distributed, provided or disclosed in any form to anyone not involved in the audit, nor is such information to be used for any other purposes.

(E) Minimum Period and Minimum Monthly Charge

The minimum period for which Recording Service without sorting is proved and for which charges apply is one month.

(F) Cancellation of a Special Order

A Customer may cancel a Special Order for Recording Service on any date prior to the service date. The cancellation date is the date the Company receives written or verbal notice from the Customer that the Special Order is to be canceled. Their verbal notice must be followed by written confirmation within ten (10) days. The service date for Recording Service is the date the Customer requests the recording to start. When a Customer cancels a Special

Issued: , 2009 Effective Date: , 2009				
	Issued:	, 2009	Effective Date:	, 2009

- 4.2 Recording Service (Cont'd.)
  - 4.2.4 Payment Arrangements and Audit Provision (Cont'd.)
    - (F) (Cont'd.)

Order for Recording Service after the order date but prior to the start of service, a Special Order charge and the minimum monthly charges will apply.

(G) Changes to Special Orders

When material changes to a pending Special Order for Recording Service are requested by a Customer, the pending Special Order will be canceled and the requested changes will be undertaken if they can be accommodated by the Telephone Company under a new Special Order. All cancellation charges as set forth in (C) preceding will apply for the canceled Special Order.

#### 4.2.5 Rate Regulations

The Special Order charge applies for each Special Order accepted by the Company for Recording Service or for a subsequently requested change.

Issued:	2009	Effective Date:	. 2009

#### 43 Automatic Number Identification

#### 4.3.1 Rate Regulations

When Automatic Number Identification (ANI) is delivered (with Feature Group D originating) and the Customer is charged the recording rate as set forth in Section 5.4.1, the ANI rate does not apply. If the Customer is not charged the recording rate, the ANI rate as set forth in Section 5.4.2 will apply for each ANI record delivered to the Customer.

## 4.4 Billing Name and Address Service

Billing Name and Address (BNA) Service is the provision of the complete billing name, street address, city or town, state and zip code for a telephone number assigned by the Company.

BNA Service is provided for the sole purpose of permitting the Customer to bill its telephone communications service to its end users and may not be resold or used for any other purpose, including marketing activity such as market surveys or direct marketing by mail or by telephone.

The Customer may not use BNA information to bill for merchandise, gift certificates, catalogs or other services or products.

BNA Service is provided on a manual basis. On a manual basis, the information will be provided by mail. Information may be provided by either voice telecommunications or through electronic mail if agreed to by both the Company and the requesting customer.

BNA information is furnished for sent-paid, collect, bill to third number, 700 and 900 service messages and messages charged to a calling card that is resident in the Company's data base.

Issued:	, 2009	Effective Date:	, 2009

- 4.4 Billing Name and Address Service (Cont'd.)
  - 441 Undertaking of The Company
    - (A) A request for information on over 100 and up to 500 telephone numbers should be mailed to the Company. The Company will provide the response by first class U.S. Mail within ten (10) business days.
    - (B) The BNA information will be provided for the calling number furnished to the extent a billing name and address exists in the Company' records, including non-published and non-listed numbers. If the billing name and address information for a specific calling number is confidential due to legal, national security, end user or regulatory imposed requirements, the Company will provide an indicator on the confidential records.
    - (C) The Company will provide the most current BNA information resident in its database. Due to normal end user account activity, there may be instances where the BNA information provided is not the BNA that was applicable at the time the message was originated.

Issued:	2009	Effective Date:	2009

- 4.4 Billing Name and Address Service (Cont'd.)
  - 4.4.2 Obligations of the Customer
    - (A) With each order for BNA Service, the Customer shall identify the authorized individual and address to receive the BNA information.
    - (B) The Customer shall institute adequate internal procedures to insure that BNA information, including that related to non-published and non-listed telephone numbers, is used only for the purpose set forth in this tariff and that BNA information is available only to those Customer personnel or agents with a need to know the information. The Customer must handle all billing name and address information designated as confidential by the Company in accordance with the Company's procedures concerning confidential information. The Company will provide to the Customer a statement of its procedures concerning confidential information.
    - (C) The Customer shall not publicize or represent to others that the Company jointly participates with the Customer in the development of the Customer's end user records, accounts, data bases or market date, records, files and data bases or other systems it assembles through the use of BNA Service.
    - (D) When the Customer orders BNA Service for both interstate and intrastate messages, the projected percentage of interstate use must be provided in a whole number to the Company. The Company will designate the number obtained by subtracting the projected interstate percentage from 100 (100-projected interstate percentage = intrastate percentage) as the projected intrastate percentage.

Issued:	, 2009	Effective Date:	, 2009

- 4.4 Billing Name and Address Service (Cont'd.)
  - 4.4.2 Obligations of the Customer (Cont'd.)
    - (D) (Cont'd.)

This whole number percentage will be used by the Company to apportion the rates and nonrecurring charges between interstate and intrastate in those circumstances where the recorded message detail is not sufficient to permit the Company to determine the appropriate jurisdiction. This percentage will remain in effect until a revised report is received as set forth following.

Effective on the first of January, April, July and October of each year the Customer may update the jurisdictional report. The Customer shall forward to the Company, to be received no later than twenty (20) calendar days after the first of each such month, a revised report showing the interstate percentage of use for the past three months ending the last day of December, March, June and September, respectively. Except where jurisdiction can be determined from the recorded message detail, the revised report will serve as the basis for the next three months billing and will be effective on the bill date in the following month (e.g., February, May, August and November). No prorating or back billing will be done based on the report. If the Customer does not supply the report, the Company will assume the percentages to be the same as those provided in the last quarterly report. For those cases in which a quarterly report has never been received from the Customer, the Company will assume the percentages to be the same as those provided in the order for service.

(E) The Company shall use reasonable efforts to provide accurate and complete lists. The company makes no warranties, expressed or implied, as to the accuracy or completeness of these lists.

Issued:	2009	Effective Date:	. 2009

4.4 Billing Name and Address Service (Cont'd.)

#### 4.4.3 Rate Regulations

- (A) Service Establishment Charges apply for the initial establishment of BNA Service on a manual basis.
- (B) A charge applies for each request for BNA information for a telephone number provided on a manual basis.

The Company will keep a count of the requests and of the messages processed. The Company will bill the Customer in accordance with these counts whether or not the Company was able to provide BNA information for all requests and messages.

(C) Where the recorded message detail is sufficient to determine a message is an intrastate message, the rates set forth in 5.4.3 following apply to each such message.

Usage for which the recorded message detail is insufficient to determine jurisdiction will be prorated by the Company between interstate and intrastate.

The percentages provided in the reports as set forth in 4.4.2(D) preceding will serve as the basis for prorating the charges. The intrastate charges are determined as follows: For usage sensitive (e.g., requests or messages processed) chargeable rate elements, multiply the intrastate percent times actual use times the stated tariff rate.

(D) When a Customer cancels an order for BNA Service after the order date, the Service Establishment Charge applies.

Issued:	, 2009	Effective Date:	, 2009

#### SECTION 5. RATES

51 Access Service

5.1.1 Service Orders

(A) Service Implementation

\$ 15.00

(1) Installation Charge

- Per trunk

\$125.00

(2) Access Order Charge

- Per Access Request

\$ 25.00

(B) Service Date Change

- Per Access Order

\$ 25.00

(C) Design Change

- Per Access Order

(D) DSO Expedite Charge

- Per DSO Order

\$ 25.00

5.1.2 Switched Access Service

Per Access Minute

**Tandem Service** 

\$0.003250

**End Office Service** 

\$0.020100

Issued: , 2009

Effective Date: \_\_\_\_\_\_, 2009

5.1 Access Service (Cont'd.)

5.1.3 Local Transport

(A) Entrance Facility

Nonrecurring Monthly
(1) DS1 \$200.00

-Per Point of Termination

Installation Charge \$500.00

Nonrecurring Monthly

(2) DS3 \$2,200.00

-Per Point of Termination

Installation Charge \$700.00

(B) Direct Trunked Transport

Facility Mileage	Monthly Rate	Per Mile
DS1	\$70.00	\$20.000
DS3	\$650.00	\$115.00

(C) Network Blocking Charge' \$0.01

Applies to FG D onl	<u>Y</u>		
Issued:	, 2009	Effective Date:	, 2009

- 5.1 Access Service (Cont'd.)
  - 5.1.3 Local Transport (Cont'd.)
    - (D) Chargeable Optional Features

Nonrecurring

- (1) SS7 Signaling Option Conversion

  -Per First Trunk Converted \$175.00

  -Per Additional Trunk Converted \$40.00
- (2) Change in Point Code -First \$170.00 -Per additional change \$34.00

Issued: , 2009 Effective Date: \_\_\_\_\_, 2009

- 5.1 Access Service (Cont'd.)
  - 5.1.3 Local Transport (Cont'd.)
    - (E) Non-chargeable Optional Features
      - (1) Supervisory Signaling

DX Supervisory Signaling arrangement

- Per Transmission Path

SF Supervisory Signaling arrangement

- Per Transmission Path

E&M Type I Supervisory Signaling arrangement

- Per Transmission Path

E&M Type II Supervisory Signaling arrangement

- Per Transmission Path

E&M Type III Supervisory Signaling arrangement (available with FGD)

- Per Transmission Path
- (2) Customer specification of the receive transmission level at the first point of switching within a range acceptable to the Company (available with FGB)
  - Per Transmission Path

Issued:	. 2009	Effective Date:	, 2009

- 5.1 Access Service (Cont'd.)
  - 5.1.3 Local Transport (Cont'd.)
    - (E) Non-chargeable Optional Features (Cont'd.)
      - (3) Customer specification of Local Transport Termination Four-wire termination in lieu of two-wire termination (available with FGB)

        —Per Transmission Path
      - (4) Signaling System 7 Per signaling connection arranged
      - (5) 64 kbps Clear Channel Capability
        -Per Transmission Path

Issued:	, 2009	Effective Date:,	2009

5.1 Access Service (Cont'd.)

5.1.4 End Office

Local Switching Per Access Minute

Common Switching Chargeable Optional Features

			Rate
Automatic	Number	Identification/	
SS7	Charge	Number	
-Per Attempt			\$0.00050

Issued: , 2009 Effective Date: \_\_\_\_\_\_, 2009

- 5.1 Access Service (Cont'd.)
  - 5.1.4 End Office (Cont'd.)

Local Switching (Cont'd.)

Common Switching Non-Chargeable Optional Features

Up to seven Digit Outpulsing of Access Digits to Customer (available with FGB)

- Per Transmission Path Group

Service Class Routing (available with FGD)

- Per Transmission Path Group

Alternate Traffic Routing (available with FGD)

- Per Transmission Path Group

International Carrier Option (available with FGD)

- Per End Office and Access Tandem

SS7 Signaling Option

- Calling Party Number (available with FGD)
- Carrier Selection Parameter (available with FGD)

Issued:	, 2009	Effective Date:	, 2009

51 Access Service (Cont'd.)

5.1.4 End Office (Cont'd.)

Local Switching (Cont'd.)

Trunk Side Transport Termination Non-Chargeable Options

Standard Trunk for Originating, Terminating or Two-Way Operation (available with FGB and FGD)

Rotary Dial Station Signaling Trunk (available with FGB)

Operator Trunk, Full Feature Arrangement (available with FGD)

Operator Trunk, Assist Feature (available with FGD)

Non-Chargeable SS7 Signaling Option

Calling Party Number (available with FGD)

Charge Number (available with FGD)

Carrier Selection Parameter (available with FGD)

Access Transport Parameter (available with FGD)

Ingrand.	2000	Effective Date:	, 2009
Issued:	, 2009	Effective Date:	. 2009

Access Service (Cont'd.)

# 5.1.4 End Office (Cont'd.)

Local Switching (Cont'd.)

	Monthly Recurring Charge
Multiplexing DS3 to DS1	\$300.00
Dedicated Switch Port	
Per DS1 Port	\$ 55.00
Cross Connect	
Per DS1 connected	\$ 25.00

Issued: 2000 Effective Date: 2000				
	Issued.	7009	Effective Date:	2009

- 5.1 Access Services (Cont'd.)
  - 5.1.5 800 Data Base Access Service

				Rate
(A)	Customer		Identification	\$0.0050
(B)	Toll Free	Number	Reservation	\$1.00

Issued:	2009	Effective Date:	2009

SECTION 5. <u>RATES</u> (Cont'd.	(Cont'd.)
----------------------------------	-----------

5.2 Reserved for Future Use

Issued:	. 2009	Effective Date:	. 2009

# 53 Miscellaneous Services

## 5.3.1 Presubscription

	Non-Recurring Charge
Presubscription,	
- Per Telephone Exchange Service	
Line or Trunk, automatic	\$1.25
Manual	\$5.00

Issued: , 2009 Effective Date: \_\_\_\_\_, 2009

5.4 Billing and Collection Services

Recurring Charge

5.4.1 Recording

-Per Customer Message

\$0.010

5.4.2 Automatic Number Identification

-Per Attempt

\$0.0120

5.4.3 Billing Name and Address

- Service Establishment Charge

\$1,600.00\*

- Query Charge Per Telephone Number

\$0.02

<sup>\*</sup> The service establishment charge applies for each separate mailing address that the information being provided by the Company is being sent to. This charge will also apply for each electronic mailing address.

5.5	Transit Traffic Service	Per MOU \$0.0065
5.6	LNP Query Service	Per Query \$0.002
5.7	Pay Phone Dial Around Recovery	Per Call \$0.65

Issued: 2000 Effective Date:	
	2009
Issued:	. 2005

## VERIFICATION

This application shall be verified under oath.

## OATH

State of California	99
County of San Mateo	SS
Todd Smith makes oath and says (Insert here the name of affiant)	that he is <u>Vice President, Corporate Controller</u> (Insert the official title of the affiant)
of <u>I n t e l e P e e r</u> (Insert here the exact legal title or name	e of the Applicant)
and belief, all statements of fact contained	ation and that to the best of his knowledge, information, in the said application are true, and the said application fairs of the above-named applicant in respect to each and
	(Signature of affiant)
Subscribed and sworn to before me, a Notary	Public/ <u>ricta Irla S</u> (Title of person authorized to admini to oaths)
in the State and County above named, this	s <u>11Nay</u> of <u>Nove wwex</u> = 2009.



BANDANA SINGH Commission \* 1747146 Notary public - California Santa Clara Count y O/Comm FeiesMeN25,2011